# LESSONS LEARNED FROM EVALUATION OF DISABILITY POLICY AND PROGRAMS

Prepared by Burt Perrin Associates for Evaluation Services Evaluation and Data Development Human Resources Development Canada

**January 26, 1999** 

### TABLE OF CONTENTS

Introduction	l
A. Purpose and Organization of This Report	
B. Context	
C. Rational for Federal Involvement in Disability	
D. Scope, Methodology and Focus on theis Paper	
Major Characteristics of the Disabled Population and Definition	ns of Disability in
Canada	
A. Demographic Highlights and Implications	6
Discussion	8
B. Definitions of Disability	10
Discussion	
C. Diversity of Disability Programs and Services	13
D. Summary: Major Findings and Lessons Learned	14
<b>Employment: What Approaches Enable People with Disabilities</b>	to Work?16
A. The Potential of People with Disabilities to Work	
Discussion	18
B. Disability Management and Return to Work	19
Discussion	22
C. Job Accommodations	23
Discussion	26
D. Wage Subsidies and Employer Incentives	27
Discussion	29
E. Supported Employment and Sheltered Work	29
i. Supported Employment	29
ii. Impact of Supported Employment	
iii. Sheltered Work	
iv. Discussion	34
F. Vocational Rehabilitation	34
i. What works	36
ii. Problems and Limitations	38
iii. Discussion.	40
G. Impediments to Greater Employment of People with Disabilities	
i. Barriers to Employment	
ii. Lack of Coordination	
iii. Limited Availability and Variable Quality	
iv. Discussion.	
H Summary: Major Findings and Lessons Learned	

	4. Barrier Removal Legislation 47	
	A. History	47
	B. Standards	
	C. Enforcement Mechanisms	49
	D. Access to Enforcement	49
	E. Evaluating the Impact	50
	F. Employment: A Special Case	
	G. Summary: Major Findings and Lessons Learned	
5.	Disability Income Programs: Balancing Opportunity and Security	57
	A. Disability Income Programs In Canada	
	i. Social Assistance for Persons with Disabilities (SA)	58
	ii. Canada Pension Plan/Quebec Pension Plan Disability	
	Benefits (CPP/QPP)	62
	iii. Workers' Compensation (WC)	64
	iv. Long-Term Disability Insurance (LTD)	
	v. Motor Vehicle No-Fault Accident Benefits (MVAB)	
	vi. Personal Injury Awards and Settlements (PI)	
	vii. Income Tax Credits (IT)	
	B. Lessons Learned From Evaluations And Reviews Of Disability Income Systems	75
	C. The (Realistic) Prospects for Comprehensive Disability Income System Reform	
	D. Five International Comparative Studies	
	i. "International Comparison of Public Disability Insurance"	80
	ii. "Return-to-Work Strategies From Other Systems May Improve Federal	
	Programs"	
	iii. Annual Report 1994-95	
	iv. "Social Insurance, Disability and Personal Injury: A Retrospective View".	
	v. Disability, Work and Cash Benefit	
	E. Summary: Major Findings and Lessons Learned	88
6.	Independent Living and Community Support Services	
	A. Introduction	
	i. Types of Services	
	ii. Other Characteristics of Community Support Services	
	B. Policy Direction for Services and Implications for Evaluation	
	Implications for the Focus of Evaluation of Community-Based Services	94
	C. Impact and Effectiveness of Alternative Approaches to Independent Living	
	Some Overall Findings	
	D. Direct Funding.	
	E. Impact of Independent Living Centres	100

	F. The Role of the Family	101
	i. Family Caregivers	101
	ii. Family Support of Independence	
	G. Impact of Other Factors	
	i. Housing and Living Arrangements	103
	ii. Effectiveness of Assessment Services	
	iii. Services for Different Populations	104
	iv. Cost Effectiveness and Decreased Institutionalization	105
	H. Incorporation of the Principles of Independent Living into Agencies	106
	I. Summary: Major Findings and Lessons Learned	109
7.	Implications for How and What to Evaluate in the Future	111
	A. Status of Current Information Base	
	B. Evaluation Paradigms and Approaches	
	C. Success Indicators for Disability Policies and Programs	
	D. Areas for Future Evaluation	
	E. Summary: Major Findings and Lessons Learned	
8.	Conclusion: Overall Lessons Learned	121
Bibli	iography	125

### Acknowledgements

This technical report on lessons learned from the disability policies and programs was prepared by Burt Perrin of Burt Perrin Associates under the direction of the Evaluation and Data Development Branch of HRDC. The study benefited from the contributions of David Baker, who was primarily responsible for drafting the chapter on barrier removal legislation, Harry Beatty for the chapter on disability income systems, and Patti Bregman for the chapter on independent living and community support systems.

The project team also included a working group from various HRDC branches. This group was composed of staff from the Office for Disability Issues (ODI), the Income Security Programs (ISP), and Strategic Policies.

This technical paper on disability is one of the lessons learned series on various social-economic subjects.

### 1. Introduction

### A. Purpose and Organization of This Report

This report summarizes lessons learned about what works or does not work in various disability policy and program areas. It is based on a review of evaluation information from a variety of sources in Canada and several other jurisdictions.

Chapter 2 presents demographic highlights of the disabled population in Canada, identifies the varying ways in which disability has been defined, and discusses the resulting implications for policy and the conduct and interpretation of evaluation.

Chapters 3 through 6 identify what is known about the effectiveness of disability policies and programs as follows:

- Chapter 3 identifies what is known about the potential of people with disabilities to work and the effectiveness of various employment strategies and approaches.
- Chapter 4 discusses the impact of barrier removal legislation in several countries, compares Canadian approaches with those of other countries, such as the *Americans with Disabilities Act*, and identifies the most effective characteristics of different legislative approaches.
- Chapter 5 reviews and discusses approaches to income security, and identifies the dilemma of balancing opportunity and security.
- Chapter 6 discusses the role of personal supports and community-based services in permitting independent living and employment, and identifies the impact and effectiveness of alternative approaches to independent living.

Chapter 7 reviews the current status of evaluation in the area of disability programs and services, and identifies implications for future evaluation approaches.

The scope of this study — essentially evaluations of any policies or programs related to any aspect of disability intervention — is extremely broad. Going beyond consideration of a single type of initiative for a discrete target group, it considers the effectiveness of many different types of initiatives for people with widely ranging needs and characteristics. It takes into account interventions at different levels, ranging from specific programs and services, such as supported employment or attendant care, to broad policy and legislative directions. This breadth of study, the range of initiatives and types of evidence considered, makes a succinct summary a challenging task.

One of the major "lessons learned" arising from the study is the importance of interrelationships and linkages across seemingly distinct policy and program areas. Decisions in any one area can influence effectiveness in others. For example, effectiveness in ensuring employment for the disabled is determined not only by employment and training initiatives, but by the availability of personal supports, transportation and housing, the manner in which income security is provided, taxation and other factors. What is most apparent is the need for a holistic, coordinated approach to disability policy and programs.

### **B.** Context

Taking stock of what is known about what works or does not work in the area of disability programs is particularly timely in the light of recent developments. For example, the federal government has been reconsidering its role in the "new union". Implementation of the new *Employment Insurance Act* provides for a new approach to the manner in which labour market integration will be supported. The federal government has indicated its willingness to "devolve" many employment services to the provinces, and has already entered into some cooperative agreements. Funding for the Vocational Rehabilitation of Disabled Persons (VRDP) Agreement has been extended for another year. During this time, the federal government will be working with the provinces to redesign the program.

In 1996, the Ministers of Human Resources Development, Finance, Justice and National Revenue appointed a Federal Task Force on Disability Issues to address the question: "What is the federal role in the area of disability?" The Task Force report: *Equal Citizenship for Canadians with Disabilities: The Will to Act* (Federal Task Force, 1996) proposed a cross-government disability policy framework and provided 52 recommendations on the themes of citizenship, labour market integration, taxation, income support and legislative reform.

The federal government, in response to the report of the Federal Task Force on Disability Issues, appointed in 1996 by the Ministers of Human Resources Development, Finance, Justice and National Revenue, has indicated its intention to take steps to address inequities, to increase access of people with disabilities to government services and to provide for increased opportunities for participation of disabled people in society. Recently, some action has been taken. The 1997-98 federal budget contained a number of measures taken from the Task Force recommendations. They include the Opportunities Fund of \$30 Million per year for three years; broadening the list of expenses eligible for the medical expense tax credit and increasing the limit on part-time attendant care to \$10,000; higher limits on the deduction for attendant care expenses; and a new refundable tax credit to cover high medical expenses for low-income working Canadians with disabilities. Minister of Human Resources Development, the Honourable Pierre Pettigrew, has acknowledged his responsibility for disability issues and has indicated that he would be placing priority on improving access by people with disabilities to the services and programs of Human Resources Development Canada (HRDC).

### C. Rationale for Federal Involvement in Disability

The starting point of the recent disability debate in Canada is often cited as the 1981 report: *Obstacles: Report of the Special Committee on the Disabled and the Handicapped.* This

report is credited with placing disability issues within a broader framework of human rights and citizenship, rather than as a health, medical or welfare issue. This perspective was officially recognized and formalized the following year, when Section 15 of the *Canadian Charter of Rights and Freedoms* granted persons with mental or physical disabilities the right to equality under the constitution. The *Canadian Human Rights Act* and provincial Human Rights legislation also reflect the principle of equality of access.

In the 1980s, a number of significant events served to focus attention on disability-related issues. In 1983, a Minister Responsible for the Status of Disabled Persons was designated and what is now the Office for Disability Issues was created. That year also marked the beginning of the United Nations Decade of Disabled Persons. The Standing Committee on Human Rights and the Status of Disabled Persons was established in 1987. The committee was instrumental, in 1991, in initiating the National Strategy for the Integration of Persons with Disabilities. The interdepartmental National Strategy was a five-year initiative with the objective of bringing people with disabilities into the social and economic mainstream of Canadian society. Its objectives centred on equal access, economic integration and effective participation of people with disabilities.

According to Statistics Canada, there were 4.2 million Canadians with disabilities in 1991, up from 3.3 million in 1986, an increase associated with an aging population. This represents 15.5 per cent of Canada's population. Disabilities cut across gender, age groups and social groups, although the rate of disability among Aboriginal people is twice that of non-Native Canadians. Some 60 per cent of people with disabilities live below the poverty level in Canada. While the full costs of disability to society are not known, the estimated cost to the federal government of income support programs and provision of goods and services is more than \$6 billion annually. This does not take into account the costs of provincial programs and private sector insurance plans, or the lost income and foregone taxes from people who are capable of work but who face barriers which prevent them from employment.

Disability-related issues in Canada are a shared responsibility. While many programs within the federal government specifically aimed at people with disabilities have now been consolidated under HRDC, activities and programs across all federal departments can have a bearing on people with disabilities. Because of the inter-relationships across diverse program areas, a number of reviews, ranging from *Obstacles* to the most recent Task Force report, have identified the need for a coordinated government vision and action. Disability-related issues are also a shared responsibility between the federal and provincial/territorial governments. Still, the principles of inclusiveness implied in Canadian citizenship and obligations under the constitution to ensure equity in the use of funds give the federal government a base to promote its commitments to equality contained in international and national instruments that underpin full citizenship.

On the international front, there has been comparable recognition of the right of people with disabilities to inclusion and equality, and the responsibilities of governments to remove barriers which prevent people with disabilities from being able to exercise their right to participate fully in the activities of their societies. The UN *Standard Rules on the* 

Equalization of Opportunities for Persons with Disabilities (United Nations, 1994) provide principles for policies and action by governments to enable people with disabilities to be able to exercise the same rights and obligations as others. Canada is a signatory to the Standard Rules and played a lead role in their development. The stated objectives of most western nations have shifted away from a paternalistic approach to providing services to people deemed not capable of participating in the mainstream of society, towards policies that encourage and facilitate independence and responsibility (Lunt and Thornton, 1993).

In summary, this report suggests there are two basic goals for government policy regarding disability. These can also be viewed as broad "success" criteria to be used in assessing the effectiveness and impact of policies and programs. They thus set the context and basis for evaluation.

- Equality and Full Participation. As discussed above, various government reviews, consistent with international developments, identify the right of people with disabilities to inclusion and full participation in society. They state that the objective of government policies and programs should be to remove barriers which make this impossible. The provision of special programs and supports is consistent with this objective, indeed is a central corollary.
- **Cost Effectiveness.** There is general recognition that all public programs and services must be provided as cost-effectively as possible. Given the limited direct government funding available for social programs, it would be helpful to government, to provider agencies, and to people with disabilities to be able to identify the types of programs and services that provide the greatest value.

Services for people with disabilities are frequently looked at as a cost with no compensating financial benefits. Another perspective recognizes the significant cost, to society and the taxpayer as well as to the individual, of maintaining the status quo. Complicating the issue is the fact that cost savings resulting from disability initiatives may appear in different budget or program areas from those where funds are expended, and may occur over a prolonged period of time. Better information about the financial benefits of programs and services for people with disabilities can aid in the development and implementation of future policies and program directions.

### D. Scope, Methodology and Focus of This Paper

This study provides an overview, indicating the general state-of-the-art regarding the broad area of disability policy. It is based on a review of existing documentation from a number of different jurisdictions and information bases around the world. It attempts to identify policy-level evaluations to the extent possible. Some of the information identified consists of formal evaluation studies and needs assessments. Other sources of information may be more descriptive in nature. Also examined are a number of reviews in various areas.

Contacts in Canada and in a number of other countries were used to help identify and obtain documentation, and also in some cases to help assess if there were any gaps in what had been identified. A variety of literature searches were also carried out and the Internet was used to identify and obtain a number of documents.

There has been an increasing amount of evaluation and research in several jurisdictions exploring the effectiveness of various approaches and strategies and on different topic areas. As this study shows, there *is* considerable evaluation and research information about what makes for effective programs and policies in the disability area. Indeed, there *are* many good examples of "success" — from Canada and from other jurisdictions. But this information needs to be identified and consolidated in a way that is useful to the policy development process.

The Bibliography lists the sources of information identified. The study does not cite each of these sources in the text.

The scope of this study is very broad, identifying major themes and giving a sense of the evaluation activity that has been done or is under way. The time available for information gathering was quite limited; consequently the search for information, while extensive, was not exhaustive. It was not possible to examine every identified study in detail. In general, a cross-disability perspective was adopted, with limited attention to literature about specific disability groups.

The documentation reviewed considered a wide variety of different types of individuals, programs and policies. These ranged from broad policy overviews to evaluation of program approaches in very specific contexts. For these reasons, a formal evaluation synthesis was not appropriate.

Despite these limitations, the study presents a fair overview of the major themes and "lessons learned" about the effectiveness of interventions related to disability. This information has many important implications for the development and implementation of policies regarding people with disabilities.

# 2. Major Characteristics of the Disabled Population and Definitions of Disability in Canada

### A. Demographic Highlights and Implications

Statistics Canada carried out two post-censal disability surveys, in 1986 and 1991, entitled Health and Activity Limitations Surveys (HALS). Findings from the most recent HALS have been reported in several sources, including Statistics Canada publications, the Social Security Reform Supplementary Paper on Persons with Disabilities (HRDC, 1994), statistics compiled by the Office for Disability Issues, and most recently, Fawcett's (1996) book commissioned by HRDC which looks specifically at the economic characteristics of Canadians with disabilities. The master file of the Labour Market Activity Survey (LMAS) also contains data on the relationship between disability and labour market activity.

Some highlights from the most recent HALS have particular implications for this study.

- In 1991, 4.2 million Canadians, 15.5 per cent of the population, have a disability an increase from 1986, when the number of people with a disability was 3.3 million, or 13.2 per cent of the population.
- Prevalence of disability is closely related to age, with the disability rate ranging from 7 per cent of children under age 14 to 50 per cent of those over age 65. The rate of disability for working age Canadians, aged 15 to 64, is 13 per cent.
- Working age Canadians (age 15 to 64) number 2.3 million, or 59 per cent.
- People with disabilities are much more likely to live in poverty than other Canadians. Of adults with disabilities, 43 per cent had an individual income of less than \$10,000 per year and 26 per cent had income of less than \$5,000. Adults with severe disabilities are much more likely to be poor than those with mild disabilities. These figures do not take into account the extraordinary costs associated with disability, which can be substantial in many situations (see, for example, Federal Task Force, 1996; Perrin, 1991; Roeher Institute, 1994a).
- In 1991, 48 per cent of working-age people with disabilities were employed, 8 per cent unemployed, and 44 per cent "not in the labour force." People with severe disabilities are least likely to be in the labour force. Nonetheless, 26 per cent of this group do participate, despite severe disabilities.

Adults with disabilities have lower educational levels than adults without disabilities. For example, HALS found that for 65 per cent of disabled people, the highest level of education

was high school or less (compared to 50 per cent for others); 6 per cent of adults with disabilities had a university degree (14 per cent for others).

For adults with disabilities, education appears to be at least as important for employment as it is for others. Data from HALS (HRDC, 1994) indicate that the employment rate of adults with disabilities with university education (67 per cent) is more than double that for those with elementary school education (30 per cent). The comparable figures for adults without disabilities are 87 per cent and 53 per cent. As Fawcett (1996) indicates, in spite of the effectiveness of education in increasing the labour force participation (and employment) of persons with disabilities, rates of participation and employment are considerably less than for people without disabilities at comparable levels of education.

There are, however, some encouraging signs. Levels of education of people with disabilities has increased from 1986 to 1991. The increase appears to be among young people. Nevertheless, the 1991 HALS survey (HRDC, 1994) still found that the highest level of schooling completed by youth (aged 15 to 24) with disabilities is lower than for youth without disabilities. The education levels of young women, both with and without disabilities, are somewhat higher than for young men.

Hum and Simpson (1996) used the master file of the LMAS to investigate the effect of disability on labour market activity. As they point out, HALS gives little data on labour market activities, while the LAMS is a rich source of data. They used multivariate regression analyses in order to identify the impact of various factors that influence the labour market activity of Canadians with and without disabilities. In their analyses, they found "evidence of a higher return to education for persons with disabilities who remain employed than for comparable workers without disabilities." They interpret these results cautiously, however, as they did not incorporate men and women who are not employed in the sample. They suggest that this is an area for future research using the LMAS master file.

HALS, and other data sources indicate that the disabled population in Canada is heterogeneous. People with disabilities do not form a unified group with common characteristics. They are just as diverse as the rest of the population, for example, in terms of

- age variation
- gender
- living arrangements
- geographic location and community size and type
- occupation
- income
- cultural/racial background
- attitudes and capabilities
- interests.

Disability-related characteristics also vary widely. There are a number of very different types of disability. As noted earlier, the severity of disability varies from person to person.

Disability can affect people in different ways. Even people with the "same" disability may be affected differently by it, and have differing needs and require differing forms of supports and accommodations.

Some disabilities are present at birth, while others develop later in life — even for the "same" disability (visual impairment, for example). The stage at which a disability develops can sometimes have significant implications for the types of supports an individual needs. Some disabilities are permanent, with their impact continuous. But others may have periodic impact. This can be true not only for certain types of disabilities - manic depressiveness, for example - but in situations where the severity of the disability varies over time, or even day to day. People with muscular weakness, for example, may need to use a wheelchair some but not all of the time.

Some disabilities, such as multiple sclerosis, can be progressive, so that the nature and severity of the disability, and how it affects the individual, can vary over the course of time. While the term disability usually refers to a long-term condition (HALS requires the disability to endure for at least six months), some disabilities are temporary in nature, particularly many of the mild disabilities.

Indeed, a major finding of Fawcett (1996) is that disability is not a static state, that a surprising number of people with disabilities experience significant changes in their disabled status from year to year. She points out that despite the fairly constant overall rate of disability in the total population from year to year, use of longitudinal data from the LMAS indicates that there is significant turnover in disability status among individuals. In addition, environmental changes - such as changes in family responsibilities and life circumstances, attitudes of employers, availability of special programs or accessibility aids - can make greater or lesser labour force involvement possible.

### **Discussion**

A large portion of the Canadian population — 4.2 million people or 15.5 per cent in 1991 — have a disability. That is not an insignificant number (in Europe, the disabled have been described as the largest minority group).

Moreover, with an aging population, the number of people with disabilities in Canada is expected to increase. The "baby boomers" represent the largest demographic cohort of the overall population. This cohort is starting to enter middle age. Over the medium term, the increase in the *numbers* of people who can be expected to develop a disability will be greatest among working-age Canadians. This will have significant financial and social implications — unless ways are developed to assist people with disabilities to continue to participate in the labour market and in society.

As noted above, education is at least as important to employment success for people with disabilities as it is for others. The changing nature of work, along with advances in

technology, *may* present opportunities for people with disabilities (see Samorodov, 1996). The availability of affordable technology can make it easier to accommodate the special needs of many people with disabilities. Increasing flexibility at the workplace, such as increasing use of flexible work hours and work-at-home practices, may also make it easier for some people with disabilities to be employed.

But it is not at all clear that a labour market in constant change will be to the advantage of people with disabilities. In particular, the increasing demands for higher-level skills as well as other factors may increase the employment disadvantage faced by people with disabilities with limited levels of education. This is a likely area for useful research in identifying the potential impact of the "new economy" on the employment prospects of people with disabilities.

Education is not a direct federal responsibility. Nevertheless, there may be opportunities to make better use of existing vehicles, such as federal support for post-secondary education, literacy, and support for youth, to assist people with disabilities in improving their education.

The heterogeneity of the disabled population also has important implications for both policy and evaluation. As Hum and Simpson (1996) state, based on their analysis of Canadians with disabilities and the labour market:

"Simply identifying those with disabilities is inadequate, since this is too heterogeneous a group. An attempt to aid all Canadians with disabilities, regardless of the degree of severity, may give rise to misleading or disappointing results."

A limitation of the data presented in this chapter is that they are now more than six years old. They precede the current economic downturn, as well as the recent reductions in social services at all levels. It is therefore probable that the current economic and employment situation of people with disabilities is less positive than the data indicate.

Data such as those provided by HALS seem to be a basic prerequisite in order to be able to monitor the impact of disability policy. Unfortunately, this post-censal disability survey has been discontinued. Hum and Simpson (1996), as well as Fawcett (1996) to some extent, were able to use the master file of the LMAS, despite some limitations, to examine the impact of disability on labour market activity, because a supplement on disability in 1989 and 1990 was attached to this longitudinal survey. The LMAS itself was discontinued in 1990.

It would seem desirable to reinstate HALS, or a similar vehicle for obtaining basic information about people with disabilities, even if on a more modest scale. Indeed, obtaining more timely information, no more than one to two years old, would be much more useful in assessing the impact of disability policies. This might be a subject for discussion between HRDC and Statistics Canada.

### **B.** Definitions of Disability

Defining disability, and determining if someone has a disability, may seem straightforward. The evidence, however, indicates that it is not. The literature shows that there are various ways of defining and identifying disability. How it is done can result in differing characterizations of the disabled population and differing conceptions of disability. This in turn can have important implications for both policy and evaluation.

For example, Lunt and Thornton (1993) point out that definitions of disability are used for a variety of different purposes. They indicate that disability is often defined to be as inclusive as possible, such as in overarching legislation, while in other cases, definitions are used to control access and to ration services. They point out that there is a tension between these non-restrictive and inclusive definitions.

A number of commentators and reviews (European Day of Disabled Persons Secretariat, 1996; Mainstream 1992; Oliver, 1990; Pfeiffer, 1995; Rioux and Bach, 1994) have indicated how the manner in which disability is determined reflects philosophical orientations as to how disability is viewed — and dealt with. They show how the traditional, medical approach to defining disability reflects the medical model approach to the "treatment" of disability. Medical definitions, which view people with disabilities as a special class of "ill" people, do not take into account the consequences of an impairment for an individual in terms of being able to function. This has led to functional definitions, such as those used by HALS, based on an individual's ability to perform activities of daily living. Disability is also sometimes defined partly in vocational terms ("permanently unemployable", for example).

Some commentators (Lunt and Thornton, 1993; Pfeiffer, 1995) have indicated that medical and functional definitions of disability emphasize incapacity rather than the capacity of people. The European Commission (1996b) argues that these definitions lead to disability being viewed simply as a problem that an individual has, failing to recognize that:

"The term 'disability' represents a complex system of social restrictions. For example, wheelchair users are denied opportunities because the places they wish or need to travel to and from are inaccessible. Visually impaired people are denied opportunities simply because ways to interact fully with the world by non-visual means are not made available to them. Persons with learning difficulties may fail to learn a task which they have the potential to perform because the appropriate training techniques required to assist them are not available."

The Mainstream 1992 report shows how the way in which disability is viewed and defined has been evolving from a "warehouse" to "greenhouse" to an "open house" perspective. While the exact terminology varies, the concepts are the same as others use in describing the evolution from a medical to a functional to a socio-political model of disability (Oliver, 1990; Rioux and Bach, 1994).

The definition of disability developed by the World Health Organization in 1980 is gaining increasing acceptance. The WHO definition recognizes multiple levels:

- Impairment a physiological, anatomical or psychological loss or abnormality of function (e.g., deafness, paraplegia).
- Disability a restriction of functional ability and activity caused by an impairment (e.g., loss of hearing, reduced mobility).
- Handicap A social or environmental disadvantage resulting from impairment or disability (e.g., communication, access to buildings).

Implicit in the WHO definition is the concept that an impairment need not lead to a disability, and a disability need not result in a handicap. A handicap results from the interaction between the characteristics of the individual and the environment. As Hollander (1993) stated:

"A disability is not a characteristic of a person but a relationship between the person and the environment. A disability is relative in this sense. This is important because it places the responsibility on the environment rather than on the person."

The WHO conception of disability is being recognized in legislation (Sweden, the United States) and policy (province of Québec, European Union) in an increasing number of jurisdictions. Indeed, it is reflected in Canada's constitution, given the interpretations of Section 15 by the Supreme Court citing the failure of society to make reasonable accommodations, to fine-tune its structures so that they do not result in discrimination against people with disabilities.

The WHO model of disability has been subject to considerably further refinement and specification, such as the International Classification of Impairments, Disabilities, and Handicaps. While there is agreement with the general approach, some of this work has been subject to criticism (Pfeiffer, 1995) for being overly medically oriented, for focusing on characteristics of the individual rather than those of society which produces the handicap, and for devaluing the quality of life of people with disabilities.

The report of the Social Security Review regarding Persons with Disabilities (HRDC, 1994) argues that determining eligibility for programs or benefits should be a different matter from clearly defining the target population. However, the report points out that:

"A common failing is to confuse the definition of disability with the eligibility criteria. It is important to remember that it is acceptable for different programs to have different objectives. However, it is not the definition of disability that should change from one program to another, but rather the eligibility criteria."

This report further asserts that: "In Canada, the tendency has been to define disability on the basis of the intentions of programs." Thus eligibility criteria often act as *de facto* operational definitions. Despite support in principle for the WHO definition of disability, programs in Canada tend to define eligibility in terms of a combination of medical, functional and vocational criteria, often also taking into account the cause of the disability and the circumstances under which it occurred - at the workplace, in an automobile accident, and so forth.

### **Discussion**

As noted above, the manner in which disability is defined has implications for how people with disabilities are viewed and treated. This in turn has important implications for the design and development of policy and program approaches.

For example, the WHO definition of disability, which emphasizes the interaction between the individual and the environment rather than the limitations or characteristics of the individual, suggests that program and policy focus should be on exploring ways of modifying societal barriers in order to be able to accommodate people with their differences. This is clearly the direction identified in Canada in major reviews such as the Task Force report and the Mainstream 1992 review, as well as by new directions in other countries. It is, however, is a different focus than that suggested by medical and functional definitions of disability, which suggest a focus on modifying or adapting individuals.

The "environmental" (as opposed to "functional") model of disability definition also has significant implications for the focus of evaluation studies and the interpretation of their findings. It suggests that evaluation studies should consider the extent to which programs and policies attempt to accommodate the needs of people with disabilities and the degree of their success in doing so. For example, evaluation of employment programs, rather than considering the capacity or incapacity of individuals with disabilities to work, should identify barriers to employment, what circumstances or supports could overcome them, and the extent to which these were applied and were effective.

It is, of course, quite appropriate for different programs to have different objectives and eligibility requirements. But the confusion of eligibility requirements with definitions, and the lack of consistency in the way disability is defined can make comparisons across programs difficult or impossible. This has implications for the design and conduct of evaluation studies, and suggests caution in interpreting data obtained from across different programs.

To illustrate, the use of different definitions can result in major differences in factors such as the size, characteristics and needs of the disabled population or sub-populations. Definitions can affect eligibility for services and characteristics of clients. Program impact can be skewed by the definitions used. For example, as the Canadian Human Rights Commission has indicated, relaxed definitions of disability have been used by some organizations to make their figures appear more positive for Employment Equity reporting requirements.

Definitions are not neutral. They have major impacts on how people with disabilities are viewed and treated. They have implications for how programs and policies are provided. The manner in which disability is defined can open up opportunities — or can limit them. Definitions provide a context for asking appropriate evaluation questions about the effectiveness of programs and policies.

### C. Diversity of Disability Programs and Services

However one defines disability, it is clear from HALS and other sources that the disabled population in Canada is very diverse. People with disabilities, by and large, have more in common with the general population than with each other.

Given the size and diversity of the disabled population in Canada, it is neither surprising nor unexpected that there is a wide range of policies and program areas, as well as services and fields of study, related to disability. There are policies and services related to some aspect of disability at all levels, from the national to community levels, involving government, the community and private sectors. Some policies and programs may affect everyone, whereas others may be very specific depending upon the nature of the disability and a myriad of other factors.

There is a wide variety of specific disability-related services, dealing with a range of needs such as housing, personal supports, transportation, medical or vocational rehabilitation, family services, and so forth. Some services are provided on a cross-disability basis, but many are directed at people with specific types of disabilities or impairments. Services can be very specialized. For example, they may be directed as very specific types of disabilities, such as specialized services for children with Down's Syndrome, or separate programs for blind and low vision people. Supported employment programs may be developed for people with intellectual disabilities, with very specific needs or characteristics.

Disability services in Canada are most likely to be provided by community-based agencies (Except, of course, for income support programs, which are provided by various government agencies, or private insurers in certain circumstances). Some agencies may provide a variety of different services, some only a single type. Organizations providing disability-related services operate within a range of different philosophies about disability. For example, there is an increasing number of programs designed to enable individuals to make their own choices about their service needs. But there are still many programs that are professionally driven, as well as programs based on models of segregation rather than integration.

There are also many different fields of study and activity examining sometimes quite specific aspects of disability. Dozens of academic disciplines may relate to disability in some way. Each of these interests has its own body of information, its own experts, its own set of beliefs and dissemination channels.

Thus the disability "system", in Canada as elsewhere, can be categorized as a fragmented collection of parts. There is often a lack of coordination. This is a major theme in many different reviews and critiques of the disability area in Canada.

People with disabilities also use the same regular programs and services that serve the general population. These include, for example, regular policies, programs and services regarding employment, education and training, transportation, housing, leisure and recreation, community-based services, and many others. "Generic" programs and services can influence the quality of life of people with disabilities more than specialized services. How regular services are structured will influence the extent to which they can be used by people with disabilities on the same basis as others - for example the extent to which they take into account potential barriers to people with disabilities. Disability-specific services are often needed only when regular programs are unable or unwilling to accommodate people with disabilities.

As indicated, for example, by the Mainstream 1992 Review, the focus has been shifting from provision of specialized services for people with disabilities to the use of mainstream services, programs, and facilities. The role of specialized services is changing, away from the provision of parallel or segregated services, towards the provision of support, facilitation and advocacy to assist the disabled to use regular services, and towards assisting these services to accommodate people with disabilities. This movement has significant implications for the development and evaluation of policy.

Given the diversity of the policies and services and fields of study related to disability, it may be inappropriate to speak of *the* disability area. The diversity and heterogeneity has implications for policy and for how evaluation is carried out. At the very least, this diversity needs to be taken into account. Various parliamentary and other reviews have indicated how the complexity of the disability area can lead to gaps and overlaps and people falling between the cracks. This suggests that a systems approach in the design of policies, and in evaluating effectiveness, might be helpful.

### D. Summary: Major Findings and Lessons Learned

- People with disabilities constitute a sizable proportion of the Canadian population. Demographic changes will result in increasing numbers of people with disabilities in Canada, particularly among people of working age. Level of education is at least as important in enabling people with disabilities to obtain employment as it is for others.
- Needs of people with disabilities are not uniform. The disability area
  consists of a wide range of largely uncoordinated services and
  components. The heterogeneity, diversity and complexity of disability
  area needs to be taken into account in the development and evaluation of
  policies that can affect people with disabilities.
- Disability is defined in a variety of ways, which are often confused with eligibility criteria for programs. This can make comparability across different programs difficult or impossible, and needs to be taken into account in the design of evaluation studies and in the interpretation of findings.
- The most commonly accepted definition of disability, from the World Health Organization, recognizes disability as involving an interaction between the individual and the environment. This definition acknowledges that an individual's ability to function in society and to be employed, for example, depends as much on the openness of society to accommodate people with their differences, as it does on the specific functional limitations of individuals identified as "disabled". This principle has important implications for the development of policies and for the focus of evaluations.

## 3. Employment: What Approaches Enable People with Disabilities to Work?

### A. The Potential of People with Disabilities to Work

As Fawcett (1996) states: "There is a common misconception that persons with disabilities who are not in the labour force are either unable or unwilling to work. . . . There is ample evidence to suggest, however, that neither perception accurately characterizes this population."

True, not *all* people with disabilities are ready and able to work. Employment may be difficult, painful or impossible for some people part or all of the time. Income support, as discussed in the following chapter, is a necessity for these people. Both internal and external barriers, such as competencies, education, work and social skills, lack of experience and exposure to the working world, and others may make work difficult. There is evidence that many of these barriers, while challenging, are not insurmountable.

But there is considerable evidence that a substantial proportion of the disability population not currently employed *is* capable of at least some form of work under the right circumstances.

The HALS data contain evidence of the willingness and ability to work of people with disabilities who are not in the labour force. Fawcett (1996) developed a composite measure of work potential which showed that 31 per cent of people with disabilities who were out of the labour force displayed one or more indicators of work potential. Fawcett describes this as an instance of "hidden unemployed." These indicators of work potential include people who:

- are active in volunteer or unpaid work;
- are students, whose status indicates that they are not out of the labour force due to an inability or unwillingness to work for pay, but because they are still preparing for work;
- express the intention to look for work within six months;
- have not worked for pay in at least five years, but have looked for work within the last two years;
- looked for work within the past five years but have been turned down because of their disability.

Fawcett also combined HALS data, which showed that 44 per cent of people with disabilities who were not in the paid labour force identified at least one environmental barrier or personal circumstance which prevented them from working. Examples of barriers identified include: fear of losing income support or disability benefits (such as assistive devices, subsidized medication) if they looked for work, or fear of difficulty in reestablishing benefits if the job did not

work out; discouragement by family or others from seeking paid work; lack of access to job information; inadequate training; past discrimination in the labour force; worry about being isolated on the job due to disability; feeling that there were no jobs available; lack of accessible transportation.

### Fawcett (1996) concluded:

"Overall, 56 per cent of persons with disabilities who were out of the paid labour force showed signs of future work potential or indicated that at least one environmental barrier kept them from looking for work. . . . This means that if circumstances (other than the disability itself) had changed, more than half of the persons with disabilities who were out of the labour force would have been likely candidates for paid employment."

Fawcett also points out that a surprising number of people with disabilities experience changes in their disability status from one year to the next. Thus many people who may not be able to work at any given time due to disability may be able to work at a later date.

There is considerable evidence from many other sources documenting the potential for a substantial proportion of people with disabilities who are not in the labour force to be employed. For example, while most people with severe disabilities are not in the labour force, a significant proportion (26 per cent in 1991) were. Why are some, but not other, people with severe disabilities able to obtain work?

The Roeher Institute *On Target* (1992) study, drawing upon HALS data and other sources of information, found that *unavailability of accommodations and supports is a bigger barrier to employment than severity of disability*. The three types accommodations most often identified by people with disabilities not in the labour force as required to enable them to work are: modified or reduced hours (33 per cent), job redesign (27 per cent), and accessible transportation (14 per cent).

There are many examples in the literature illustrating the potential of individuals with severe disabilities, given appropriate supports, to work. For example, an evaluation sponsored by the United States Department of Health and Human services and the United States Department of Labor (Conley, Azzam and Mitchell, 1995) examined a number of projects that attempted to assist students with very severe disabilities (mainly "severe mental retardation", many with multiple disabilities) to obtain integrated employment using natural supports. The intention of the project, confirmed by the evaluation, was to select people with the most severe disabilities. In the absence of these projects, almost all of these students would have been placed in traditional sheltered employment or workshops.

#### This evaluation concluded:

"Severity of disability is usually not an insurmountable barrier. Although there are obvious instances of persons whose disability is so severe that there is little

possibility of vocational success, the projects have shown that integrated employment with natural supports is a realistic goal for most people who are currently placed in segregated employment sites [and similar alternatives]."

Nevertheless, this evaluation stated that "the fact that such employment is possible, however, does not mean that it is easily obtained." The evaluation identified some of these barriers and how projects were able to deal with them. For example, most employers indicated a strong need for substantial up-front assistance when employing people with severe disabilities. While increased costs were viewed by employers as relatively minor, most employers were heavily supported by publicly funded sources in dealing with inappropriate behaviours, training and other considerations. The evaluation identified this support as a key ingredient to the success of the project.

The evaluation acknowledged that this type of project requires additional up-front costs. It noted, however, that unlike more traditional approaches such as sheltered employment or workshops, costs would be offset in many (but not all) cases by subsequent savings.

As discussed in Section 3.F, supported employment programs similarly work with people with severe disabilities, enabling people to work with supports who otherwise would not be able to do so. Various supported employment programs have identified the potential of people with severe disabilities and limitations to do productive work.

The Neil Squire Foundation operates a Creative Employment Options program, which trains individuals with severe physical disabilities. A large proportion of program participants, most of whom were unemployed for a long time without prospects before entering the program, found competitive employment upon graduation. The Foundation (Birch, 1996) has carried out a preliminary calculation of economic benefits arising from participation in the program, based on savings in social assistance spending and taxes paid to government. These preliminary calculations suggest a return on investment to government of better than six to one.

#### **Discussion**

There is strong evidence that a substantial proportion of people with disabilities who are not currently in the labour force are capable of being employed in some way, given the proper supports and the removal of barriers that now make it impossible. The evidence provides support for the inter-relatedness of disability issues. It suggests that approaches that focus only on employment and do not address other barriers may be limited in effectiveness. The relationship between success in employment with the availability of other services is discussed further in Section 3.G.

Many people with disabilities, given the right work environment and removal of barriers, are capable of working on the same basis as anyone else. For others, particularly with severe disabilities and lack of skills, regular, full time employment may be challenging. But they may still be capable of doing *some* work. For example, most of the participants in the projects

evaluated by the Conley et al. (1995) study worked less than half time. This may not be sufficient to generate a subsistence income, but given the right income support policies, can more than cover the costs of the necessary supports and result in a number of other social and economic benefits as well.

Showing that it is possible for people with severe disabilities to work does not mean that it is easy. The evidence indicates that the severely disabled face many barriers. Successful interventions may require significant up-front costs, although there is potential for longer term savings. In addition, it should also be noted that labour market demand plays a significant role in determining their employment opportunities.

The evidence also suggests the need to get away from thinking in terms of the employability of people with disabilities as a yes/no dichotomy. Many people, in particular those with very severe disabilities, may be limited in the *amount* of work they are able to do. Some people may not be capable of fully productive work, but can still do something, given the proper supports. Many people currently excluded from the labour force may be capable of earning some income through employment, even though this may be less than what they are currently receiving through various income support schemes.

As Chapter 5 indicates, the all-or-nothing concept of employability incorporated into eligibility requirements of disability income systems such as the Canada Pension Plan acts as a powerful disincentive for people receiving social benefits to search for or to accept employment. The GAO (1996h) stated that:

"The either/or nature of the disability determination process encourages applicants to focus on their inabilities. The documentation involved in establishing one's disability can, many believe, create a 'disability mind set' that erodes motivation to work, and the length of time required to determine eligibility can weaken skills, abilities, and habits necessary to work."

The Mainstream 1992 review, Parliamentary Standing Committee reports, and other sources document that greater opportunities for employment as a major interest of the disabled community. They identify exclusion from the labour force as one of the leading sources of frustration among people with disabilities.

### B. Disability Management and Return to Work

Disability management refers to the active follow-up and management of employees who are off work due to work-related or non-work-related disabilities and injuries. It typically involves providing case management services and the availability of individualized, outcome-oriented rehabilitation services, taking into account the individual's interests and abilities. The type and intensity of services and supports vary depending upon the individuals' needs and what would enable them to return to work.

One aspect of disability management involves the individual in learning how to address functional limitations resulting from the disability. But another aspect focuses on the workplace, identifying what accommodations are needed to permit return to work. These may include, for example, redesign of the work site in some way, flexible or variable work hours, or assignment to a different job within the individual's capabilities.

Disability management strategies have been used in various countries, including Canada, the United States and Europe, in both the private and public sectors. There is strong evaluation evidence showing their potential to enable individuals to return to work and to create significant cost savings. Perhaps as a result, these strategies are becoming more widespread.

The Canadian federal government recently completed a disability management pilot project involving provision of case management services in three federal departments. An independent evaluation (Insight Canada, 1996) of the pilot project indicated that this approach represented "a major departure from traditional *laissez-faire* federal practices in the area of helping injured and disabled workers return to work." The evaluation identified the success of the project in enabling individuals to return to work sooner than they would have otherwise. It also identified major cost savings, and a return on investment for the project costs of 63 to 1 (cost savings resulted from actual and expected reductions in Disability Income, Long-Term Disability, and Workers' Compensation benefits). Significantly, there were even greater savings among employees whose disability was not work related than for those who were injured on the job.

The evaluation pointed out that it can sometimes be difficult to determine with any certainty how many people might return to work without intervention. But it notes that in at least one of the participating departments, virtually all the employees had been away from work for a long time.

The evaluation also identified a number of barriers. One major barrier is the lack of understanding among senior managers of the real costs of paying for workers who are absent due to injuries or disabilities. There is also lack of accountability for results, as the costs are incurred elsewhere in government. For these and other reasons - such as lack of interest in people with disabilities or lack of belief in their capability to be re-employed - the evaluation documented resistance to disability management among human resources departments as well as by senior management. This resistance impaired operation of the program in some instances, hampering early intervention in particular, because of lack of timely access to the names of people at an early stage. The evaluation recommended the development of an accountability framework in order to provide incentives.

The federal government experience reflects that of the private sector, where major cost savings have been found through proactive return-to-work strategies (Boxer, Young, and Kirchner, 1994; GAO, 1996i; Prost, 1996; Washington Business Group on Health, 1994). Private sector return-to-work efforts are becoming more commonplace. GAO (1996i) indicated that:

"Disability managers in the U.S. private sector spend money on return-to-work efforts because they believe such efforts are sound investments that reduce disability-related costs. Studies have estimated that the full cost of disability to employers ranges from about 6 to 12 per cent of payroll. Such costs include insurance premiums, cash benefits, rehabilitation benefits, and medical benefits paid through workers' compensation and employer-sponsored disability insurance programs. Companies may also incur additional expenses for training and using temporary workers and retraining employees with disabilities when they return to work. When businesses help workers with disabilities return to the workplace, they are able to reduce some of these costs."

Extensive research (Allingham and Hyatt, 1995; Butler, Johnson, and Baldwin, 1995; Johnson and Baldwin, 1993) using the Ontario WCB database has explored in detail the impact of vocational rehabilitation on the return to work of workers with permanent partial disabilities. This research has demonstrated the effectiveness of rehabilitation in fostering return to work. It points out, however, that the first return to work is not always an indication of permanency. It emphasizes the importance of the interaction between employer and employee and the need for accommodations to permit workers to be re-employed.

The need to view disability management as an ongoing versus a one-time process was reinforced in an evaluation of job accommodations sponsored by HRDC (Perrin, 1995). This study showed that accommodation needs can change over time for a variety of reasons including changes in the nature of the disability itself, changes in the requirements of the job as well as actual job positions, and aids and adaptations becoming out of date as a result of things like changes in technology.

Many WCB claimants have what might be considered relatively minor or temporal disabilities resulting from workplace injuries (although they are severe enough to keep people from being able to work, and as the research cited above indicates, they can make work re-entry difficult or impossible without workplace accommodations and other supports). But the same finding has been found for people with traumatic injuries. For example, Abrams, Barker, Haffey and Nelson (1993) carried out a cost-benefit analysis of a work re-entry program for persons with traumatic brain injury. This study found that three quarters of the participants obtained employment, that the ratio of total taxpayer benefit to total program operational costs was 2:1, and the ratio of total taxpayer benefit to the state cost was 4:1. Australian evaluation - (Athanasou, Brown, and Murphy, 1996; Murphy, Brown, Foreman, Young, and Athanasou, 1996; and Murphy and Athanasou, 1994) found comparable findings following spinal cord injuries. Thus there is strong evidence of the potential of back-to-work strategies for people with any type of disability.

While disability management refers specifically to disability in a workplace context, there is evidence that similar return-to-work strategies can be effective in enabling participants on disability benefits programs such as CPP to re-enter the workforce. As discussed in Chapter 5, the GAO (1996i) indicated that the same types of disability management practices used in the private sector could also be used for Social Security beneficiaries. Indeed, the study indicated that the social insurance programs in Germany and Sweden are applying such practices with

considerable success. The Social Security Administration found that successful rehabilitations in its return-to-work initiative save five dollars in benefit payments for every dollar spent.

What is the potential for active return-to-work strategies in enabling social insurance beneficiaries, such as CPP disability claimants, to become re-employed? On the one hand, claimants are by definition "unemployable." But also by definition, CPP claimants have previous work experience. Evidence such as that cited above strongly suggests that a substantial proportion of people *could* be able to be re-employed, given the appropriate types of support and encouragement and workplace accommodations. The evidence also strongly supports early intervention. The longer people are away from work, the less likely they are ever to return to work.

Indeed, one of the conclusions of the evaluation of the National Vocational Rehabilitation Pilot Project (HRDC, 1996a) was that: "There can be good potential for rehabilitation for some CPPD beneficiaries, even after several years on benefits, as the medical condition stabilizes and adjustments to accommodate disabilities have been made." The evaluation found that the pilot project achieved a 43 per cent success rate. This led to a recommendation that CPPD establish a permanent rehabilitation function.

But, as indicated in Chapter 5, it is important to bear in mind the conclusion of the GAO (1996i): even a little bit of success can result in large cost savings. It is *not* necessary to have immediate spectacular successes. Very small, marginal gains in this area can have a very significant long-term impact on the costs of a disability income program. As the GAO says, even "if an additional 1 per cent of the 6.3 million were to leave SSA's disability rolls by returning to work, lifetime cash benefits would be reduced by an estimated \$2.9 billion."

### **Discussion**

There is great potential for significant cost savings from disability management and return-to-work strategies. Indeed, return-to-work is probably the area with the greatest possible potential for cost savings, given the previous work experience of participants, along with their employment skills and abilities. Evidence on the return on investment of return-to-work strategies in both the private and public sectors suggests remarkable results.

But there are barriers to effective implementation. Disability management requires initial investment of staff resources and expenditures on case management and rehabilitation. The savings typically accrue to a different cost centre than that which incurs the cost. There can also be resistance from managers, including senior as well as line managers, and human resources departments who are not interested in or do not see the potential of investing in workers who develop a disability.

The evidence indicates that the following strategies are important to the success of return-towork strategies:

- Intervene as soon as possible: the longer away from work, the less likely a worker will return.
- Provide active case management.
- Recognize that workplace accommodations will frequently be necessary to permit reentry.

The full costs of long-term disability to an employer and to the taxpayer are not well understood. Tremendous cost savings are possible even with small successes with return-to-work strategies.

#### C. Job Accommodations

An accommodation is "any modification of the workplace, or in workplace procedures, that makes it possible for a person with special needs to do a job." (Canadian Mental Health Association, 1993). Accommodations include technical aids and devices, as well as physical adaptations to the workplace.

But accommodations also include adaptations to how the work is structured. They can include, for example, modifications to required work tasks and responsibilities or to how they are to be performed, use of flexible hours, provision of break opportunities for workers who tire easily, rearrangement of work materials so that they are more easily accessible (such as placing commonly accessed files within reach of someone who uses a wheelchair), and many other "soft" modifications. The accommodation needs of one senior manager with limited vision included receiving written materials and documents in alternative formats in advance of meetings (Perrin, 1993).

It is noteworthy that workplace accommodations include many adaptations that are frequently available to any worker to address special needs and to enhance productivity. There is no difference in kind between a tool, such as a dictaphone, which enhances the productivity of a senior manager, and a device, such as a computer sound card, which can enhance the productivity of a blind worker.

There is extensive documentation of the potential for accommodations to make employment possible for people with disabilities. For example, part of the scope of an evaluation of the operation of the Job Accommodation Network (JAN) in Canada, sponsored by HRDC (Perrin, 1995) involved consideration of the relative importance of accommodation regarding the employment of people with disabilities. This evaluation concluded that the biggest barriers to the employment of people with disabilities include lack of job opportunities, employer attitudes and limited training and work experience. It nevertheless indicated that accommodation is one important element that can make a difference to the employability of many people and make it possible for them to be productive.

JAN itself has compiled extensive documentation from its own evaluations, research and follow-up with employers, of how a wide range of both "hard" and "soft" accommodations have enabled workers with disabilities to carry out their job responsibilities and to address the needs of their employers. The ILO international report (Samorodov, 1996) indicated that job adaptations can be a very effective and cost-efficient vehicle for the promotion of employment for workers with disabilities - and can sometimes benefit non-disabled workers as well. As discussed earlier, research looking at the return to work of Ontario WCB claimants (Butler, Johnson, and Baldwin, 1995; Johnson and Baldwin, 1993) indicated the importance of job accommodations.

A study by the Roeher Institute (1994b) for the Canadian Labour Force Development Board found that:

"Over 70% of people with disabilities who require accommodations in mainstream jobs and who believe they have been adequately accommodated are on the job 35 hours or more in a given week. This figure approaches the proportion of non-disabled persons on the job 35 hours or more. Those who do not have adequate accommodations available to them are much more likely to be completely off the job in a given week. Those receiving adequate accommodations are more likely to be moderately and severely disabled than those not receiving or needing accommodations. In fact, individuals receiving adequate job accommodations are generally more severely disabled than the working age population with disabilities as a whole. Yet, they log more hours per week at work than other workers with disabilities."

A prevalent perception is that accommodations are costly. Indeed, the perceived cost of accommodations is a concern of many employers. The evidence, however, indicates that this perception is a myth. In reality, accommodations are typically not expensive. The ILO report (Samorodov, 1996) indicated that:

"Accommodation costs (job adaptation) for people with disabilities are usually very low, under US\$1,000 per person, notwithstanding fears of employers to the contrary. Accommodation costs are not, therefore, a major hindrance to the employment of workers with disabilities and are thus cost-effective."

JAN's data, based on ongoing surveys with employers who have provided accommodations (see Perrin, 1995 for examples of recent data) provide more specific information about the costs of accommodation provided by employers in the United States. Employers who have made accommodations say 18 per cent involved no costs to them, with an additional 50 per cent costing between \$1 and \$500. Only 5 per cent cost more than \$5,000. The mean cost of accommodations was \$992 while the median cost was \$200.

The same employers also estimated the money their companies saved as a result of being able to hire or retain a qualified employee, elimination of the cost of training a new employee, savings in workers' compensation or other insurance costs, increase in the workers's productivity, and

other factors. The savings identified averaged 27 times the expenditures they reported in providing accommodations.

Nevertheless, there are exceptional circumstances where an accommodation that is critical to an individual can be costly. Large organizations may be able to absorb these costs, but the JAN evaluation indicated that this would be difficult or impossible for smaller employers. Sometimes funding assistance might be available, such as through provincial vocational rehabilitation services or through HRDC, as well as favourable taxation. But few employers are aware of these, and the delays and bureaucratic procedures involved in getting government support are frequently incompatible with running an efficient business operation.

HRDC formerly had a separate fund which could be used for accommodation expenses. An evaluation of the effectiveness of Canada Employment Centres (CECs) for people with disabilities (Perrin, 1994) indicated that this fund was little known among CEC staff, bureaucratically complex and time consuming to access. As a result, it was rarely used. It was also designated for workplace training. Now Human Resource Centres Canada (HRCCs, the new term for CECs) managers have discretion to pay for selective accommodations. But these expenditures must come out of general HRCC budgets, providing little incentive to do so.

As the next chapter indicates (and see Baker, 1992), grant-levy schemes such as those used in Germany and France, in addition to providing incentives for employers to hire people with disabilities, including people with severe disabilities, also produces a fund which can be used to fund exceptional accommodations.

Accommodation costs are most likely to be high when they involve personal care or support. These may include, for example, attendant care or job coach support to a person with a developmental handicap. Because of their qualitatively different nature, high levels of personal support, such as for individuals with severe disabilities, are usually treated differently than other forms of accommodation. It is probably unrealistic to expect employers to cover these costs. But there may be other ways in which such services could be provided. For example, the study of transitional employment services for individuals with severe mental disabilities discussed earlier (Conley, Azzam and Mitchell, 1995) indicated the need for support from agencies to permit employment of severely handicapped people needing high levels of assistance. As discussed in Chapter 6, the availability of personal support, especially when directed by the individual, can assist in employability.

The JAN evaluation (Perrin, 1995) identified employer attitudes and lack of information or misinformation as the biggest barrier to providing accommodations in the workforce. There is a general lack of awareness of the potential of accommodations to enable many people with functional limitations to work. There is also a need for practical "how-to" information on how to make accommodations in specific situations.

A study exploring career development among managers and other employees with disabilities within the Ontario public service (Perrin, 1995) identified employer and supervisor attitudes as critical. This study identified factors associated with career success. It found that with the right

attitude, arranging for the actual accommodation usually turns out not to be a big deal, even for employees with severe disabilities. This evaluation found that the same basic factors that apply to the management and development of *all* employees also apply to the management and development of employees with disabilities. In particular, the role of the direct manager, and an openness to taking an individual approach and to good communications with the employee, were critical to the career success — and performance — of employees with disabilities.

HRDC's evaluation of JAN's operation in Canada<sup>1</sup> (Perrin, 1995) concluded that it offers a useful service to employers and others; but that its potential effectiveness and reach is limited by organizational factors. The evaluation recommended that JAN's services in Canada be refocused so that it could play a stronger role as an information resource to increase awareness of and interest in accommodation.

As discussed in the next chapter, the United States government, in support of implementation of the Americans with Disabilities Act (ADA), funds a variety of technical resource centres and other services to provide information and support to employers, rehabilitation workers and individuals about how to make accommodations.

It is worth noting that the principle of reasonable accommodation, which puts the onus on employers and others, is enshrined in employment equity and human rights legislation in Canada, as well as in the Charter of Rights of Freedoms, as the Supreme Court recently affirmed. It also is reflected in the legislation and policies of a number of other countries.

### **Discussion**

Lack of accommodations represents one of the major barriers to employment of people with disabilities. A common misunderstanding is that job accommodations are usually expensive. The reality, however, is that the vast majority of accommodations are of low cost, and many, such as modifications to work tasks or work site arrangements, require no expenditure at all.

The major barriers with respect to accommodations are attitudes and lack of information, both about the potential of accommodations to permit people with disability to be able to do productive work, and about the practical "how-tos" of implementing accommodations in specific situations. Addressing these barriers could result in increased employment opportunities for people with disabilities.

Employers could be primarily responsible for providing accommodations, other than for exceptional circumstances. Government could be mainly limited to making information and support available to counter the myths and lack of awareness concerning job

accommodations, as the United States government is doing in support of ADA. As noted earlier, the evaluation of JAN identified the potential of JAN to play a broader role in increasing awareness and interest in accommodation and in supporting government initiatives such as Employment Equity legislation. There may also be other ways of addressing these needs.

### D. Wage Subsidies and Employer Incentives

"Wage subsidies are put in place by governments as an incentive to the private sector to increase the training and employment opportunities for persons with disabilities." (Raskin, 1995) They are intended to compensate employers for lost productivity. They act in two ways: "First, they reduce effective wage costs to employers; second, they enhance the employability of people with disabilities who are seeking work (Samorodov, 1996)." Most wage subsidies are for limited duration, ranging from months to years, and sometimes a sliding scale of reduced subsidies is used.

The evidence indicates that the effectiveness of wage subsidies is mixed. Indeed, the ILO international report noted that, given the fact that wage subsidies are found in many countries, the lack of evaluative efforts is striking (Samorodov, 1996). The two ILO reports (Raskin and Samorodov cited above) provide useful summaries of the state of the art. As these ILO reports discuss, wage subsidies are intended to help with entry into the labour market, especially for people with little or no work experience. They can provide work experience for individuals who may have trouble obtaining it without some form of assistance.

On the other hand, subsidies can have a number of disadvantages. For example, being categorized as eligible for a wage subsidy can negatively reflect on the capabilities of individuals so categorized and lessen their chances of employment. Individuals who are not retained following a subsidized placement, in particular, may have difficulty obtaining future employment. Thus, it is uncertain that the use of wage subsidies will result in long-term employment following the end of the subsidy. Some employers may use subsidies to "churn" workers through their firm and as a substitute for paying the expenses themselves of training new employees. Subsidies could end up providing employers with a cheap pool of labour, taking advantage of rather than benefiting the subsidized workers. (Samorodov, 1996).

Samorodov (1996) cites an Australian study indicating that employers prefer subsidies to quota or other systems which provide a compulsion to hire. Yet the same study identified only a limited effect of the subsidies.

The experience in Quebec is indicative. Quebec has the most experience in providing employer subsidies for hiring individuals with disabilities through its *Contrat d'intégration au travail* program. A recent evaluation of this program (SOM, 1996) indicated that the positive effects were temporary. It did not appear to increase the income, employment or quality of life of program participants. This evaluation, as well as an earlier review by Le Vérificateur Général du Québec (1993), indicated that the program did not provide for specific individual objectives regarding improvements in productivity and behaviour or what would be required to maintain the person on the job. The Office des Personnes Handicapées du Québec, the sponsor of the

program, did not maintain sufficient records to effectively monitor the program, the nature of the clientele and the activities of employers. As a result, some employers hire persons who are not eligible for subsidies or receive higher subsidies than required.

Subsidies can also result in dead weight (jobs that would not have been created otherwise) and displacement (employment of a subsidized individual at the expense of someone else). A European Commission report (European Commission, 1996a), considering the impact of subsidies across all target groups, says that only 10 to 24 per cent of the expenditures result in net job gains.

On the other hand, despite these dangers, there can be some potential benefits of employer subsidies. The same European Commission report also notes: "Even if the [wage subsidy] scheme is associated with strong displacement effects, it might, nevertheless, be justified on equity grounds as a means of sharing out available work more justly."

The two ILO reports indicate that: "Wage subsidy measures that emphasize the individualized needs of the participant may be the most successful (Raskin, 1995; Samorodov, 1996)." These reports single out the Individualized Subsidized Job (ISJ) component of the Job Development Program (JDP), part of the Canada Jobs Strategy, formerly offered by the Employment and Immigration Canada (now HRDC) as an exemplary model (Raskin, 1995).

An evaluation of the Job Development Program identified the ISJ component as the most cost-effective approach of the various JDP components. It resulted in improved employability and earnings for participants with disabilities. Raskin (1995) observed that the "ISJ was not only cost effective for the quantitative element (cost benefit) but was also qualitatively beneficial," as it consisted of training options tailor-made to the needs of individuals and was closely linked to the private sector. Raskin noted that a major reason for the success of this program was the individualized nature of the placements for disabled participants, which included accommodation of the work site and attention to the individualized needs of the participant. He also indicated that those ancillary activities such as pre-training outreach that directly target groups of persons with disabilities to ensure participation levels that are representational was another reason for the success of the program.

Similar results were found in an evaluation of the Program for the Employment Disadvantaged (PED), formerly provided by EIC (Social Policy Research Associates, 1984). The purpose of this program was to aid in the employment of employment-disadvantaged people, including people with disabilities. It combined subsidies to employers, along with assistance for workplace modifications where needed, coordination, counselling and other forms of support by both Canada Employment Centre staff and through community-based agencies.

This evaluation found that the program was highly effective. For example, some 16 per cent of the participants with disabilities were still employed with the PED employer following the completion of the program and at the time of the follow-up evaluation survey; another 24 per cent went on to employment with other employers. In addition, the evaluation indicated that the

program led to improvement in productivity, job skills, employability, job satisfaction and related matters.

There can be other forms of incentives to employers besides subsidies. These can range from tax exemptions for work site modifications and accommodations to the provision of support from community-based agencies. A transitional employment program (Conley, Azzam and Mitchell, 1995) which provided intensive support to individuals with severe disabilities and to employers was discussed above. The supported employment model discussed later in this chapter is another example of an approach which uses an external agency to provide the support necessary to ensure that the employees are productive.

A model is Projects with Industry (Policy Studies Associates, 1986), operational in the United States for some time. This program involves a partnership between business, industry, labour and the rehabilitation community, ultimately leading to competitive employment for individuals with disabilities. The program combines training at realistic work settings, coupled with supportive services. Past evaluations of the program indicated its effectiveness in leading to subsequent employment for a significant proportion of the participants.

Also, as discussed in the next chapter, the grant levy or redistributive model provides incentives for employers to hire individuals with disabilities. The European Commission (1996a) report on employment in Europe indicated that subsidies are one way of reducing social costs to employers, but: "In most Member States . . . policy to reduce labour costs for particular workers has predominantly taken the form of reducing social contributions on employers rather than the payment of subsidies as such." Raskin (1995), in his ILO review, concludes that barrier removal is more effective than subsidies.

### Discussion

The effectiveness of wage subsidies to employers is mixed. Generally, only a minor proportion of the expenditures result in employment that endures in some form beyond the subsidy period. Other approaches, such as barrier removal, generally appear to be more successful. Subsidies appear to be most effective when they are part of a coordinated approach involving other forms of support to the individual and to the employer, when they are specifically tailored to address the identified needs of the individual, and when follow up and monitoring is provided.

### E. Supported Employment and Sheltered Work

### i. Supported Employment

Supported employment involves training and work at regular work settings. The unique feature of supported employment is the provision of a job coach who provides support and training to the individual, and who is responsible for helping to get the job done. As O'Brien (undated) says, supported employment:

"Expands the focus from the person alone to the person plus a skilled coach. Instead of simply assessing the job ability of the person alone, as practitioners within a typical continuum of day services do, supported employment practitioners consider what the person can do with the assistance of a job coach. This shift in focus allows many previously excluded people to work."

A key component to the success of supported employment is the flexibility to provide as much or as little support as the client needs, on an ongoing basis. Initially, support is often intensive, often on a one-on-one basis. This emphasis enables clients to be placed who do not initially possess all the necessary work or social skills required for immediate job success. As the worker demonstrates independence, the amount of support is gradually reduced (Perrin, 1986).

Supported employment involves a place-train model. This is the reverse of the traditional rehabilitation train-place approach. Indeed, with the traditional model of rehabilitation, as discussed later, the "placement" often never happens. Another critical difference from traditional rehabilitation is the role of the job coach. As indicated above, the task of the job coach is not just to help train the worker, but to work together with the participant as needed to get the job done and to satisfy the requirements of the employer.

Supported employment is a relatively recent innovation. It was first developed in the early 1980s by Paul Wehman and his associates at the Rehabilitation Research and Training Center at Virginia Commonwealth University (Wehman and Hill, 1986; Wehman and Kregel, 1984; Kregel and Wehman, 1996). In the mid 1980s, there were only a handful of largely experimental programs in Canada (Perrin, 1986, 1987). But over the last ten years, supported employment has mushroomed, particularly in North America. It is also attracting worldwide attention, with programs in many other countries as well (Lunt and Thornton, 1993).

Supported employment is now a field of study in its own right. There is a developing body of literature, both practically oriented and scientific in nature.

Supported employment has been used most extensively with individuals with intellectual disabilities. It has also been used, to some extent, for people with psychiatric disabilities. But the ILO report (Samorodov, 1996) indicated the potential for application of the model to other client groups, citing examples. It suggested the approach is especially good for younger workers.

Supported employment has been used primarily with people who are not considered able to work competitively, and who otherwise would probably be placed in a sheltered workshop. With many individuals, the support can be completely or largely withdrawn, so that they can eventually work independently. Others, however, require intensive ongoing support in some form.

There are many examples of individuals earning minimum or competitive wages. But probably in many situations, wages are at sub-minimal rates or none. Income from support usually serves as

a supplement to income security rather than as a replacement. For example, a survey of Canadian agencies providing supported employment services in 1988 (cited by Raskin, 1995) found that 21 per cent of participants no longer needed social assistance, and an additional 24 per cent were less dependent. But 55 per cent of participants were still dependent on social assistance to the same level as previously and almost 50 per cent of supported workers earned less than \$50 per week.

Despite impressive gains, most clients probably are still not working at full productivity. A majority of placements involve part-time work. Supported employment approaches have enabled people with even very severe disabilities to work productively in regular work settings. Indeed, the model was originally developed as a means to enable individuals with very severe disabilities to work in integrated settings. Nevertheless, concern has been voiced as to the extent to which supported employment programs are indeed serving the most severely disabled. Rankin (1995) cites findings from the 1988 survey of Canadian supported employment programs which indicate that a majority of clients are mildly to moderately disabled. He indicates that federal funding policies which limit the length of time for which a person can receive program support may serve as a disincentive to serving individuals with severe disabilities who may require ongoing support.

Supported employment placements offer varying degrees of integration (Kregel and Wehman, 1996; Rankin, 1985). Individuals often work side by side with regular workers. In other cases, they may work within an enclave or in other settings which provide limited contact with other workers. As Rankin indicates, the degree of integration provided by supported employment is a concern. Nevertheless, supported employment offers more potential for integration than other options for this client group.

A recent innovation or variant is to make use of natural supports, such as co-workers or neighbours, as an alternative or supplement to the job coach (O'Brien, undated; Conley, Azzam and Mitchell, 1995). This approach can provide for truer integration, with assistance and greater contact from "real" people rather than from paid staff. Kregel and Wehman (1996) report on a survey which indicates that a majority of supported employment provider agencies in the United States are using natural supports.

In theory, there is potential for greater cost effectiveness and for supporting a greater number of persons when making use of natural supports. With limited evaluation to go by, it is as yet unclear if this is indeed the case. The 1996 Kregel and Wehman study suggests that staff time on job sites has changed among about one third of agencies using natural supports. The role of paid staff shifts from providing support directly to an individual, to arranging and supporting others to do so. This can require significant coordination, along with initial efforts in arranging for the provision of assistance. Follow-up and ongoing support may also be required, with the paid staff person stepping in when problems may develop.

# ii. Impact of Supported Employment

There have been a number of research studies examining the impact and the benefits and costs of supported employment. For example, Conley, Rusch, McCaughrin, and Tines (1989) review the findings from a number of cost-benefit analyses as well an longitudinal studies which indicate that:

"Although costs of supported employment are greater than benefits during the first years of implementation, costs decrease and benefits increase over time because employees increase both their hours and average hourly wages, while the amount of support needed on the job decreases."

A review of the literature, with a good bibliography of studies looking at the impact of supported employment came to a similar conclusion:

"The research demonstrates that supported work programs are furnishing positive employment outcomes for persons with mental handicaps. These programs have provided a financial benefit to the supported workers through increased earnings and a benefit to taxpayers in decreased program costs. Although costs to taxpayers in the first years of program implementation may exceed financial benefits, the research suggests a substantial increase in benefits is found in subsequent years (Jamieson and Hawkins, 1992)."

Johnson, Lewis and Bruininks (1993) discuss the status of evaluation of supported employment. They describe a case study which provides a model of a benefit-cost analysis that found a benefit-cost ratio of 2.55 across 13 different program sites. But they caution that there is lack of consensus on the goals and outcomes that should be used in evaluation of supported employment. They state that: "Increasingly, researchers . . . have even argued that valuing supported employment strictly on the basis of cost-effectiveness is illegitimate, and that supported employment should be valued primarily from its capacity to assist individuals in achieving social and community integration."

Agencies appear to have had the most success in finding placements for supported employment in small business settings (Perrin, 1987; Rankin, 1995), but there are many notable exceptions. There have been difficulties in arranging for placements in the public sectors, due to a number of systemic barriers related to formal hiring policies (Drug, 1993).

Individuals with disabilities who have the opportunity to participate in supported employment are very positive, and prefer it to other alternatives (Kregel and Wehman, 1996). People want to work, for social as much as for economic reasons. They like the opportunity to be in a regular setting and to interact with other people who do not have a disability. They like having their work valued through pay, and they like getting a regular pay cheque.

Supported employment can be structured so that employers can hire people with even severe disabilities and expect the work to get done. Agencies have had most success marketing supported employment to employers based on a business case, rather than on "good cause" marketing.

The National Institute on Disability and Rehabilitation Research (1994) concluded that: "Supported employment has created unprecedented employment opportunities for people previously perceived as too severely disabled to engage in meaningful work." ILO, in its reviews of the cost effectiveness of the full range of employment services, found supported employment as the most promising model. For example, Raskin (1995) stated:

"Concerning the effectiveness of various types of wage subsidy and supported employment programs, evaluations indicate those that emphasize the individualized placements in the private sector that are designed to meet the personal requirements of each trainee, and which provide participants with an opportunity to interact with non-disabled co-workers and to form relationships offer the best prospects for success."

ILO's international review of the cost-effectiveness of policy options for workers with disabilities (Samorodov, 1996) concluded that:

"Supported employment — trying systematically to integrate workers with disabilities into open employment alongside able-bodied workers — can best be recommended as really efficient in terms of costs and benefits. Over time it may be expected to result in significant returns on investment. . . . Research shows that these programmes, even in the initial high-cost phase, are markedly cheaper than traditional day centres, and especially than sheltered employment schemes. . . Supported employment is thus a cost-effective alternative to other more traditional approaches; it also reduces people's usage of and dependency upon day services."

# iii. Sheltered Work

The traditional approach to "training" people with disabilities who could not find fully competitive employment has been to place them in sheltered workshops. Workshops offer simulated work, generally with the stated objective of training people in work skills. In practice, few people move on, and the "skills" and behaviours they develop are the contrary of what is appropriate for regular work. Token "wages" only are offered.

Recognition of the limitations of sheltered workshops is increasing, as is awareness of the superiority of alternatives such as supported employment. For example, Conley et al. (1989) points out that supported employment programs have resulted in the long-term maintenance of approximately 50 per cent of their workers for at least one year alongside non-handicapped workers. By contrast, traditional sheltered workshops have reported that only about 10 per cent of their workers were placed in integrated work settings over the same time period.

As a result, workshops are being phased out in favour of supported work and other alternatives across Canada, the United States and Australia (Lunt and Thornton, 1993).

Sheltered work is still widely used in Europe. But this is largely based on a different model from that used in North America, closer to what has been termed "protected work". Sheltered work in Europe tends to be based on an economic rather than the social work orientation in North America, with more emphasis on productivity. These centres often have able-bodied workers as well as workers with disabilities and tend to provide real pay.

Quebec, unlike other jurisdictions in Canada, has funded Centres de travail adapté (CTAs), which operate more like the European model of sheltered work. For example, CTAs are expected to operate as non-profits enterprises. The proportion of able-bodied workers may vary from 0 to 49 per cent.

But there are still severe limitations to this model. For example, two recent evaluations of Quebec's CTAs (Le Verificateur General du Québec, 1993; SOM, 1996), found that the economic interests of the CTA worked at cross purposes to their stated objectives of assisting workers with disabilities to enter the regular workforce. Because of a desire to improve their productivity, they frequently hired workers who did not meet the criteria for the program, and were reluctant to let their most productive workers go for regular employment. Similar findings were found in the CTAs in France (Velche, 1995).

#### iv. Discussion

The supported employment model has proved itself a better and more cost-effective alternative than sheltered work in almost all respects. It supports integration and is very popular with individuals, who welcome the opportunity to do real work in real settings. It provides opportunities for people previously perceived as too severely disabled to engage in meaningful work to work in regular employment settings. There is ample evidence of the cost effectiveness of supported employment, although costs can be high in the initial stages. The ILO has identified the superiority of the supported employment model to other alternatives in terms of benefits to individuals as well as cost savings, and has indicated the potential for widespread use of the approach.

#### F. Vocational Rehabilitation

A great deal falls under the rubric of "vocational rehabilitation". In Canada, vocational rehabilitation is provided according to source of disability, disability type, source of income, or attachment to the work force. The Workers' Compensation Boards (WCBs) provide vocational rehabilitation to those injured in the workplace; the Canada Pension Plan (CPP) has a pilot vocational rehabilitation project for recipients of CPP disability benefits; private

employer insurance companies provide or purchase vocational rehabilitation from private sources.

In essence, the various types of employment interventions discussed in this chapter can be considered as forms of vocational rehabilitation. Samorodov (1996) defines rehabilitation programs as those which "treat and retain workers with disabilities and return them to gainful activity." Traditional approaches to rehabilitation view training/rehabilitation as a discrete step separate from employment, using a "train then place" model. As discussed elsewhere in this chapter, this approach is changing, with supported employment using a "place and train on the job" model. There is increasing recognition that rehabilitation is most likely to be effective when it is as closely related as possible to employment.

Individual vocational rehabilitation programs generally are those which tailor the services provided or the goods and services purchased to fit individual needs. Vocational assessments, vocational planning, training, retraining or education, job placement and follow up constitute the core of individual vocational rehabilitation programs. Related to these services is the provision of living allowances and disability-related supports and services such as home or vehicle modifications, orthotics and prosthetics, wheelchairs, private tutoring, treatment programs, and mobility training. Goods and services are purchased from a wide range of suppliers, both public and private. Some vocational rehabilitation programs, WCBs for example, provide subsidies and other strategies to encourage employers to hire injured workers.

Vocational rehabilitation is sometimes thought of in terms of those services funded through the federal/provincial VRDP agreement. Each provincial and territorial government delivers an individual vocational rehabilitation program, cost shared with the federal government under the *Vocational Rehabilitation of Disabled Persons* (VRDP) *Act* and Agreement - except for Quebec, which now has a separate provincial delivery system for vocational counselling services.

In addition, there is a large network of volunteer community-based agencies which provide individual vocational rehabilitation services. These agencies can be provincially funded and cost shared under the VRDP agreement, or they can be funded as employment assistance services by HRDC, as well as through other sources. Some agencies providing vocational rehabilitation are disability specific, such as the Canadian Paraplegic Association, the Canadian National Institute for the Blind, and the Canadian Hearing Society. Others are more generic, such as the Neil Squire Foundation, the Canadian Association of Independent Living Centres, and Goodwill Industries. A key network is Associations for Community Living which delivers individual vocational rehabilitation services through such measures as supported employment programs and vocational assessment and training programs in sheltered workshops.

The *VRDP Act* is an enabling financial transfer mechanism which allows the federal government to enter into agreements with the provinces and territories to contribute to certain provincial and territorial costs of vocational rehabilitation programs. Currently there is a

standard agreement with all provinces and territories. A variety of programs are cost shared under the VRDP Agreement. These include individual vocational rehabilitation programs, vocationally related portions of alcohol and drug programs and mental health programs in the community and in psychiatric hospitals; an array of agency delivered programs; supported employment programs; and sheltered workshops. There *is* some similarity of programming across provinces is the individual vocational rehabilitation programs, which are in turn similar to the vocational rehabilitation programs of WCBs and CPP.

Support for post secondary education is a major feature of the individual vocational rehabilitation programs of the provinces. VRDP cost sharing is available for tuition, books and supplies, living allowances and support services such as attendants, interpreters, tutors and technical aids, provided to individuals. Some provinces coordinate or harmonize the assistance with the federal and provincial student loans program. Several provincial governments provide funding for special needs counsellors in community colleges and in some universities, often cost shared under VRDP.

#### i. What works

There have been no evaluations of the effectiveness of the VRDP Agreement. This would be challenging to do, given that VRDP is a mechanism to transfer funds to provincial and territorial treasuries according to rather general provisions and guidelines. It is up to each province and territory to establish eligibility and to determine the activities relating to the vocational rehabilitation of disabled persons. The effectiveness of the VRDP Agreement as a transfer mechanism would have to be evaluated, for example, in terms of federal objectives in transferring the dollars, and in terms of provincial objectives for obtaining the dollars. Basically there are 12 different provinces and territories that deliver programs through various departments and ministries. There are also many different criteria for eligibility, depending on the types of programming - mental health, alcohol and drug, as well as individual vocational rehabilitation, for example.

The requirements for provincial reporting of activities under VRDP are basically limited to the numbers of clients served. There is no requirement for provinces to report on those who have obtained employment or on other impact measures.

One provincial program where outcome statistics are available is the Vocational Rehabilitation Services (VRS) program delivered by the Ontario Ministry of Community and Social Services. Based on 1992-93 statistics for this program, Raskin (1995) found that: "The outlay for vocational rehabilitation services is clearly cost beneficial when compared to the wages the client earns, and is cost beneficial in all cases when compared to the option of maintaining persons with disabilities on social assistance." In 1992-93, \$20.7 million was expended on the VRS program in direct client-related expenditures, with an additional amount for staff and administration. Of those clients who received vocational training or restorative services, 56 per cent were employed when their cases were closed. On average, about 5,000 new cases are opened each year.

In the disability management section of this chapter, we refer to research that examined the impact of vocational rehabilitation on the probability of post-injury return to work among claimants of the Workers' Compensation Board (WCB) of Ontario (Allingham and Hyatt, 1993; Raskin, 1995). This research showed that vocational rehabilitation is having its intended effect — it is reducing the negative impact of residual disabilities on the return to work of injured workers, thereby reducing benefits and social costs. In 1992, the Board paid \$1,797 million in long-term disability benefits, and \$561 million for rehabilitation. A total of 28,704 persons or 7.6 per cent of claimants were referred for vocational rehabilitation and 46 per cent of these were rehabilitated, most returning to work with the enterprise that employed them at the time of the accident.

It is not always possible to compare vocational rehabilitation programs to each other. The needed data are not usually in place, and definitions of eligibility as well as program and service objectives differ. For example, in the United States, an attempt was made to compare the Veterans Affairs VR program with the State VR programs. In the Veterans' program, 8 per cent of eligible veterans were placed from 1991-1995. In the State program, 37 per cent were placed in suitable jobs. However, the Veterans' program tended to concentrate on higher education and defined a suitable job as a position consistent with the veteran's aptitudes, abilities, and interests. In contrast, the State VR program provided a wide range of services and defined a suitable job as including non-wage earning jobs. Most clients in the state program had severe disabilities. Veterans Affairs spent \$20,000 on each employed veteran and \$10,000 on each program dropout, with more than one-half of costs going towards veterans' subsistence allowances, while State VR programs spent on average about \$3,000 on each rehabilitated client and about \$2,000 on each dropout, none of which covered clients' living expenses (GAO, 1996j).

An American-based review of the literature (Weed and Lewis, 1994) surveying the past 25 years overwhelmingly identifies the benefits of rehabilitation for people with disabilities regardless of referral or payment source. These studies showed that the provision of rehabilitation services can be a cost-effective approach to assisting people with disabilities to work, and that earlier referral reduces the cost of rehabilitation and increases the likelihood of employment. As indicated earlier, the evaluation of the National Vocational Rehabilitation Pilot Project Evaluation for CPP (HRDC, 1996) identified the potential for rehabilitation even among many long-term beneficiaries, and recommended that CPP establish a permanent rehabilitation function.

In its research on transitional programming, The Roeher Institute (1994b) identified a number of factors contributing to the success of people with disabilities making transitions from school to work. These included access to needed disability-related aids and devices, adequate accommodations in the workplace, access to on-the-job training, participation in other forms of training, access to the personal support required to manage daily living activities, and access to rehabilitation services. This study also identified the importance of the following factors:

 Building on the individual choices of people and in so doing, promoting individual motivation and skill development;

- Involving family members and friends in the transition process;
- Early planning in the high school careers of students;
- Work experience through cooperative education and other arrangements;
- Keeping employers informed about the available pool of potential workers with disabilities.

Indeed, as noted earlier, one of the conclusions of the National Vocational Rehabilitation Pilot Project (HRDC, 1996) was that: "There can be good potential for rehabilitation for some CPPD beneficiaries, even after several years on benefits, as the medical condition stabilizes and adjustments to accommodate disabilities have been made." The evaluation found that the pilot project achieved a 43 per cent success rate (based on individuals' voluntary participation of the project). This led to a recommendation that CPPD establish a permanent rehabilitation function.

The ILO review (Samorodov, 1995) of the cost effectiveness of policy options for workers concluded that each dollar spent on vocational rehabilitation sets or maintains in motion a stream of output of several dollars. "In sum, funds that are expended on vocational rehabilitation programmes are repaid, on average, many times over. Rehabilitation can thus be considered a really cost-effective way of promoting the return to work and employment of people with disabilities."

#### ii. Problems and Limitations

Rehabilitation is not always effective. A GAO (1993) evaluation (*Vocational Rehabilitation: Evidence for Federal Program's Effectiveness is Mixed*) noted wide disparities in the types and amounts of services provided, with insufficient data to relate these to outcomes. That review indicated that short-term gains of rehabilitation often do not appear to endure. These findings led to a recommendation for a longitudinal study to explore broader issues such as who can be served, at what intensity, and with what results.

The study noted that just 5 to 7 per cent of those potentially eligible actually were served by state/federal vocational rehabilitation programs. The GAO recommended a longitudinal study to explore broader issues such as who can be served, at what intensity, and with what results. In a more recent study, GAO (1996g) stated that vocational rehabilitation has played a limited role because of restrictive state policies. "Beneficiaries are generally uninformed about the availability of VR services and are given little encouragement to seek them."

Individuals with disabilities and referring agencies have complained about difficulties in obtaining access to vocational rehabilitation services and programs. Waiting lists for some provincial vocational rehabilitation programs can be as long as a year or more. On the other hand, some Workers' Compensation Boards have made changes that include early intervention and a right to early vocational assessment and counselling (Roeher Institute, 1992).

Thus a major problem with vocational rehabilitation services is who actually receives services. Only a small proportion of those in need and who could benefit from these services actually obtain them. Due to lack of data, it is not always clear how priorities are set, and which types of clients are most likely to receive services. It is not clear whether those with severe disabilities and most in need are most likely to receive services. This is true in Canada as well as in the United States.

The provincial vocational rehabilitation programs exclude from eligibility those who are eligible for vocational rehabilitation from Workers' Compensation Boards and through the department of Veterans' affairs. Those in receipt of CPP are not formally excluded and can be referred to provincial programs for vocational rehabilitation. While this has not been expressed as a problem, experience in the United States may be instructive.

The Social Security Act states that as many individuals applying for disability benefits as possible should be rehabilitated into productive activity. To this end, people applying for disability benefits are to be promptly referred to state vocational rehabilitation (VR) agencies for services intended to prepare them for work opportunities. The federal government reimburses the state for successful outcomes. However, it was found that the success-based reimbursement system was ineffective in motivating VR agencies to accept beneficiaries as clients. Delayed VR intervention can cause a decline in receptiveness to participate in rehabilitation and job placement activities, as well as a decline in skills and abilities (GAO, 1996g).

The Roeher Institute's (1992) report on Canada's employment-related programs for persons with disabilities describes a range of problems regarding the way in which vocational rehabilitation programs in general are delivered and how they are organized, which make it difficult for individuals to link into training, education, and employment systems:

- Eligibility criteria make it difficult for people with disabilities to qualify for training, education and job placement opportunities;
- Wishes, interests or needs of the client may not be given adequate consideration;
- Vocational counsellors may have difficulties providing applicants with an overview of the full range of vocational options;

- There is a disturbing increase in assessment procedures without a corresponding investment in vocational counselling services, leading to delays in accessing training or other services;
- There is a shortage of counselling and planning support which is a key barrier to employment and training opportunities.

As the next chapter indicates, models of support which emphasize individual control and responsibility are usually more effective than models where professionals decide unilaterally what is best for someone else.

A study of financial assistance available to post secondary students with disabilities (National Education Association of Disabled Students, 1993) concluded that 44 per cent of the disabled post secondary students indicated that their income from all sources was not sufficient to cover their education-related services and equipment costs, considering their disability. This study also indicated that the provincial VRDP programs need to allow students more freedom to choose their program of study, and greater opportunities to pursue education beyond the undergraduate level.

Client involvement in the development of policy and evaluation strategies can also be a significant factor. For example, a participatory action research approach, involving both clients and professionals, was used to explore key issues and to prioritize goals for the Reauthorization of the Rehabilitation Act in the United States. Both groups identified employment as the most serious problem facing persons with disabilities. But clients and professionals had differing views regarding goals and priorities for rehabilitation policy (Cantrell and Walker, 1993). An evaluation of the consumer involvement process in the Mainstream 1992 Review (Perrin, 1993) identified how consumer involvement led to greater trust on all sides and to a better product. With upcoming federal/provincial/territorial discussions on the future of VRDP, it may be worth considering how clients should be involved in the process.

# iii. Discussion

Vocational rehabilitation *can* work in getting individuals into employment and back into jobs. It has a significant cost benefit to the taxpayer, and reduces costs to income support programs and to insurers. It plays an important role for individuals with disabilities who benefit both economically and in terms of quality of life.

The biggest limitation to the effectiveness of rehabilitation is lack of access. Only a small proportion of those people who could benefit from these services are able to access them.

Not all approaches are equally effective. There is a lack of evaluation or even the maintenance of basic data needed to determine who is being served and to what effect. In general, vocational rehabilitation is most effective when it is individualized and closely oriented to the labour market. Innovative approaches to rehabilitation recognize that rehabilitation/training and

employment should not be viewed as discrete steps. Without attention to the support needs after a person starts or returns to work, gains may be short-term only.

# G. Impediments to Greater Employment of People with Disabilities

As this chapter shows, there is overwhelming evidence of the *potential* of many more people with disabilities to work in regular settings. There is strong evidence of the *potential* of a variety of strategies such as disability management, vocational rehabilitation, supported employment, and workplace accommodations.

Why is it, then, that there has been no significant breakthrough in the employment of people with disabilities? The following discussion examines the systemic impediments to employment of persons with disability - discounting labour market effects, such as the fact that high unemployment rates have a "discouraged worker" effect on persons with disability, and that individuals may fear of losing eligibility for income benefits if they attempt to return to the workforce.

# i. Barriers to Employment

The greatest impediments to the employment of people with disabilities are non-employment barriers. This has been a theme of Canadian parliamentary Standing Committee reports going back to the original *Obstacles* report in 1981. It has been a theme in representations of clients and consumer organizations over the years, as reflected, for example, in the report of the Mainstream 1992 review. As cited earlier, the Roeher Institute's (1992) report on Canada's employment programs for persons with disabilities showed that lack of supports was a bigger barrier to employment than severity of disability.

A European report (European Day of Disabled Persons, 1996) stated that:

"It is essential to place disabled people's restricted access to employment within the wider framework of discrimination and to recognise the barriers that constrain disabled people *outside* the workplace. Indeed, increased employment opportunities will only be effective alongside reforms in other policy areas."

A *combination* of factors are necessary for a person with a disability (or for anyone else, for that matter) to be able to work. As the European Commission (1996b) put it:

"If a disabled person has a good personal assistance programme, has received a good education and has excellent employment opportunities, the investment made in personal assistance, education and employment may be wasted if there is no transport infrastructure to enable them to travel to and from work."

Following are some of the major barriers preventing many people with disabilities from working:

Accessible and Affordable Transportation. A job opportunity is of no use unless a person with a disability has some means of getting to it. Few people with disabilities are able to afford an automobile, and for many, public transportation is not accessible. Alternatives, such as taxis, can be non-existent or unaffordable. There are alternative transportation systems for people with disabilities in many (but by no means all) communities. It is beyond the scope of this paper to discuss the limitations of these systems, but it should be noted that they are now being reduced in many communities because of budgetary considerations.

**Education, Training and Work Experience.** Education and experience is becoming increasingly important for everyone in today's labour market. It was shown earlier in this study that educational levels of people with disabilities are lower than for others. The difficulties people with disabilities have had obtaining access to employment services at Canada Employment Centres (now HRCCs), including access to training programs were discussed above.

Education is not a federal responsibility. The federal government is devolving direct support of training to the provinces. Nevertheless, the federal government could still use other existing mechanisms, such as dedicating some support for post secondary education specifically to students with disabilities, through its support for literacy, and through its youth initiative. The Federal Task Force (1996) recommended that agreements with the provinces for the devolution of employment services should build in provisions to provide for equitable services for people with disabilities.

Despite the mixed effectiveness of employer subsidies, this chapter showed that a variety of programs designed to assist people with disabilities obtain workforce experience, including some previous EIC programs such as the ISJ, were effective. Providing training and experience on the job is also a basic premise of supported employment.

**Social isolation and segregation.** As shown in Chapter 6, it is difficult for individuals to learn the basic social skills necessary for interacting with others - including skills and habits necessary for interacting in the workplace and self confidence - when they are isolated.

**Personal supports.** The critical importance of supports to the employment of people who require this form of assistance has already been discussed. Eligibility requirements for personal assistance and other forms of supports frequently act as a barrier to employment. For example, in some situations, eligibility for supports is tied to specific programs or settings, such as sheltered workshops. These supports would be lost if a person leaves the segregated setting for employment. In other situations, a person needs to be declared "unemployable" to qualify. Eligibility for provincial vocational rehabilitation services requires a person to demonstrate the potential to be "competitive" and "independent." Depending upon how this is interpreted, it could rule out eligibility for people who can work, but only with supports, creating a catch-22 and discriminating against the people who may need the supports the most (Roeher Institute, 1992).

Supports, including aids and devices, are discussed in Chapter 6.

**Accessibility**. Accessibility is more than a ramp to the front door. It involves other factors, such as the job design, how an individual is treated by supervisors and coworkers, openness to reasonable accommodations, and so forth.

**Extraordinary Costs of disability.** People with disabilities bear many additional costs arising specifically from their disability (Perrin, 1991; Task Force, 1996). In many cases, these can be quite substantial. They may face considerable extra costs in accepting employment (such as extra transportation or personal assistance expenses) which could make employment uneconomical. The 1997 federal budget has widened the scope of eligible items that could qualify for tax deductions. But this still will not eliminate these extraordinary expenses.

**Disincentives Arising out of Income Support**. As discussed elsewhere, the manner in which income support is provided can act as a powerful disincentive to employment for many people. In some situations, individuals are allowed to earn only a very small amount before their benefits are reduced dollar for dollar. For poorly paid jobs, an individual may risk losing disability benefits as well as income and be worse off financially by working. The all-or-nothing eligibility requirement of many income support programs, which requires a person to be certified "unemployable" in order to qualify, means that a person taking a job that does not work out may jeopardize eligibility for reinstatement of benefits. Thus for many people, it would not be rational, from an economic perspective, for them to seek or to accept employment.

**Attitudes.** Perhaps *the* major barrier is attitudes of others, including employers, the public and administrators of government programs, including HRDC's employment programs. Much can be done with no additional expenditures. But negative attitudes and a lack of understanding result in discrimination, both overt and systemic, which serves to exclude people with disabilities from society in general as well as from the workplace. Constant rejection will take its toll on the self confidence of many people. This in turn can become a vicious circle, resulting in further lack of interaction and lack of opportunity.

Attitudes are often responsible for the lack of understanding of the costs of disability, and underlie a resistance to invest in strategies - despite their demonstrated return on investment - that can enable people with disabilities to work, such as disability management and other approaches.

#### ii. Lack of Coordination

As the above section states, success in employment is dependent upon other factors and the availability of a variety of other services. A number of reviews have identified the need for more coordination across program areas and across jurisdictions. For example, the 1990 Report of the Standing Committee on Human Rights and the Status of Disabled Persons (A Consensus for Action: The Economic Integration of Disabled Persons) pointed out the need for a comprehensive approach to "untangle the web of interconnected causes and effects."

The National Strategy for the Integration of Persons with Disabilities was developed, in part, in response to the above Standing Committee Report (Price Waterhouse, 1993). One of the primary objectives of NSIPD was to serves as a vehicle for a coordinated cross-departmental government approach. The Interdepartmental evaluation (HRDC, 1995) confirmed "the need for coordination and leadership to produce synergy and complementarity."

Nevertheless, the National Strategy was not successful in producing more coordination across departments. The Interdepartmental evaluation found a lack of sufficient consultation both across departments and with the disabled community and the provinces. It found little requirement for interdepartmental accountability, little incentive for departments to share information and inadequate communications. The evaluation concluded that; "The implementation of a coordinated interdepartmental approach requires a commitment from all participating departments to contribute to [a] shared, corporate vision."

The Canadian experience is not unique. For example, a U.S. GAO (1996a) report with the descriptive title: *People with Disabilities: Federal Programs Could Work Together More Efficiently to Promote Employment* stated:

"Because people with disabilities often face multiple barriers to employment, including insufficient job training, lack of transportation, and employer discrimination, they may require services from more than one program to make employment feasible. . . Because services are often not coordinated among programs, people with disabilities may receive duplicate services or face service gaps."

At the *program* level, a common theme of various evaluation studies is that programs that appear to have been most successful have tended to take a holistic and individualized approach. They look at the capabilities, needs and barriers facing an individual, and develop plans that can involve the coordination of a range of services, including action by employers, arranging for accommodations, and assistance from agencies with support and on-the-job training in both employment and social skills. The GAO (1996a) identified common strategies among highly successful employment training programs for economically disadvantaged adults. These strategies include: provision of transportation, child care, basic skills training and other supports needed to enable clients to complete training and to acquire employment, strong links with the local labour market, a combination of formal training and one-on-one support.

# iii. Limited Availability and Variable Quality

As this chapter has shown, rehabilitation *can* be very successful as well as cost effective in enabling people with disabilities to work. One of the biggest barriers to success with rehabilitation is its limited availability, with only a small portion of the eligible population actually receiving rehabilitation services. The amount of services people can receive often falls far short of what is needed to permit employment. It is not always clear what basic priorities are used to determine eligibility.

Again, this situation is not unique to Canada. As indicated earlier, the GAO (1993) noted that only a very small proportion of those potentially eligible actually were served by state/federal vocational rehabilitation programs. The GAO recommended a longitudinal study to explore broader issues such as who can be served, at what intensity, and with what results.

There is compelling evidence of the *potential* of various strategies, such as disability management and supported employment, to enable many people with disabilities to work. But these require effective implementation. There is limited evaluation of the effectiveness of individual programs. The GAO (1994e) indicated that few employment programs in the United States collect information on participant outcomes or study program effectiveness. The same is true in Canada.

And not all program approaches are appropriate or effective. In general, rehabilitation approaches are most likely to be effective when they are practically oriented and focused on the workplace, such as disability management and supported employment, when they take a coordinated approach, and when the individual is directly involved.

#### iv. Discussion

The greatest impediments to employment of people with disabilities are *non*-employment barriers, such as lack of transportation, limited education and training opportunities, various disincentives, lack of supports, and other factors. Failure to recognize and to address these barriers results in discrimination and the exclusion of people with disabilities from the workplace. The need for a coordinated approach has been well documented, but has proved elusive at the government policy level. The National Strategy was not successful in producing a shared vision or interdepartmental coordination.

# H. Summary: Major Findings and Lessons Learned

- There is ample evidence documenting the potential of many more people with disabilities (including severe disabilities) to work.
- A number of people may require supports in order to work. The all-or-nothing approach to employability, which is inherent in the criteria of many income support and employment programs, effectively excludes many people who are capable of at least some degree of employment.
- The greatest impediment to the employment of people with disabilities are *non*-employment barriers, such as lack of transportation, limited education and training opportunities, and lack of supports. Lack of supports can be a bigger barrier to employment than the degree of disability. The need for coordination has been well documented.
- Workplace accommodations can enable many people with disabilities to be employed in regular employment settings. Contrary to common perceptions, most accommodations are not costly, with half costing less than \$250 and many not involving any cost.

- The cost effectiveness of vocational rehabilitation has been well documented.

  Disability management, where the return on investment of return-to-work strategies can be many times the investment, is particularly cost effective.
- Not all rehabilitation approaches are equally effective. Integrated approaches, which involve employers, provide necessary supports and accommodations to individuals as well as to the employers through outside agencies, and which take an individualized approach, seem to be most effective.
- The effectiveness of wage subsidies has been mixed. Other approaches, such as barrier removal, generally appear to be more successful.
- Supported employment, involving training and work at the job site with the help of a job coach, provides opportunities for people previously perceived as too severely disabled to engage in meaningful work, and can permit those who traditionally have been relegated to sheltered workshops or to day centres to do real productive work in regular employment settings. It is very popular with individuals, can facilitate integration, and is less costly than alternatives.
- The National Strategy did not succeed in producing a shared inter-departmental vision or coordination. Lack of coordination across program areas, federal government departments, and jurisdictions, has been identified as responsible for inconsistent policies, overlaps and gaps.

# 4. Barrier Removal Legislation

by David Baker

Barriers prevent people with disabilities from achieving their full potential for financial and social independence. Those responsible for barriers may occasionally remove them out of goodwill or out of desire to expand market share, but there is no proof that awareness campaigns and public relations campaigns have had more than a marginal impact. As a result, there is a clear international trend in all industrial, and many developing countries, to enact barrier removal legislation.

#### A. History

Prior to 1973, the government's role was confined to providing social programs designed to alleviate in some measure the burden of disability. Other than employment programs for disabled veterans, the private sector was not engaged in the process of inclusion or barrier removal.

At about this time, four strands came together to create the circumstances in which private sector involvement became a priority, and legislation became the mechanism for ensuring its involvement. First, human policy called for "normalization" of services (that is, offering

47

persons with disabilities the opportunity for lives that are as close to those of non-disabled persons as possible). Second, people with disabilities and their families become politically active and demanded to be heard by policy makers. Third, governments began to be concerned about the growth of the welfare state, making independence and private sector involvement more desirable. Finally, the experience of legislation dealing with the social disadvantage experienced by other minorities, notably members of visible minorities, provided a model to be adapted and utilized.

The first prototype of using human rights legislation to enforce barrier removal was the *Rehabilitation Act of 1973* in the United States. At about the same time building codes, which had previously addressed safety and generic convenience issues, were expanded to include accessibility standards for newly constructed buildings. These American innovations quickly found their way into Canada, with access provisions being added to building codes in the mid-1970's and generic human rights legislation being amended to include persons with disabilities amongst the protected classes between 1977 and 1982. The trend has accelerated in recent years with Australia and France introducing legislation in 1992 and Germany adding a constitutional amendment in 1994. The United Kingdom enacted legislation that came into effect in December 1996.

Apart from Canada and the United States, experience with mandatory barrier removal legislation has been limited. Nevertheless, some reference will be made to issues that have arisen in these jurisdictions during implementation.

Other industrialized countries have relied on legislated labour market policies to achieve the economic integration of persons with disabilities. These policies will be examined in the employment section of this chapter.

Later in this chapter, Canadian and American results with barrier removal is compared. As will be apparent, the major divergences occur following the enactment of the *Americans with Disabilities Act* in 1990. The Canadian attempt at a second generation anti-discrimination equivalent to the *ADA*, referred to as "Omnibus Legislation", is generally acknowledged to have been inconsequential. Mandatory employment equity legislation in Canada will be examined in the employment section of this chapter.

#### **B.** Standards

Standards that clearly prescribe the means of compliance (*prescriptive standards*) are most easily enforced, since the amount of interpretation involved is minimal. This is the approach taken in building codes, and in regulations under the *National Transportation Act*.

Debates about prescriptive standards tend to concern their sufficiency (have they gone far enough in removing barriers) on the one hand, and their rigidity (new and better technologies are available but not prescribed) on the other.

In order to avoid rigidity but ensure access, the Americans have shifted to *descriptive standards* which describe the results to be achieved in order to be in compliance. An example would be the requirement that a technology be "accessible and useable" by people with disabilities, without prescribing the technical specifications of the technology. This approach accommodates changes in technologies over time. However, debates can arise about how the standard is to be interpreted in a particular set of circumstances. In Canada, there was a decision to move to a descriptive standard for over-the-road bus accessibility in 1993, but this federal-provincial agreement has not been implemented.

Another innovation introduced by the Americans was the *conditional standard*. A business is required to meet the standard provided it does not impose "undue hardship" or it is "readily achievable". This adds a further level of interpretation which is problematic in that it depends on the subjective circumstances of the business, a condition with which a person with a disability could not be expected to be familiar. The clarity of the standard itself still makes enforcement straightforward. The business must demonstrate exceptional (and presumably temporary) circumstances which would warrant excusing it from the standard of general application.

In Canada, the standards that apply in human rights and Charter cases are far from clear. For convenience, the "discrimination" standard will be described as a *conditional formula*. Examination of the litigation record shows that few cases hinge on the condition of a business' financial means. Virtually all depend on a determination of whether the barrier is discriminatory. Notwithstanding the fact that term has been in use since the mid 1960's, it remains a matter of dispute, with the formula constantly evolving. As a result, there are very few clear precedents from which it could be said a "standard" for barrier removal had emerged. Because of the stigma associated with the term discrimination, and the low rate of success achieved by complainants, there is a strong incentive to oppose these applications. Cost effectiveness is therefore very low, as is the resulting social integration.

#### C. Enforcement Mechanisms

Canada has relied on generalist enforcement mechanisms: human rights commissions. Such bodies have been process rather than goal orientated. Because the standards are unclear, the costs of enforcement have been very high relative to the barriers removed.

In the United States, and all the other countries that have followed its lead in enacting disability-specific barrier removal legislation, the enforcement mechanisms are more specialized. In the United Kingdom and Australia, for example, disability-specific sections have been established in issue-specific departments or agencies.

Specialization has contributed in these jurisdictions to the ability to consult credibly with both the disabled community and the business sector (transportation, for example) to which the legislation applies. Consultation has been essential to the development and periodic revision of

standards. It has also contributed to effective public education and credible technical assistance. As a consequence, the barrier removal process is much less dependent on litigation.

In Canada, experience with the Canadian Transportation Agency (CTA) and the Canadian Radio Television and Telecommunications Commission (CRTC) confirms the thesis that specialized agencies are more cost-effective at barrier removal and successful at achieving social integration (with resultant impact on labour market integration). The CTA particularly has combined a regulation initiating role with an adjudicative role, which has contributed significantly to barrier removal. However, current government policy favouring voluntary guidelines over enforceable standards has detracted from the steady progress that had been occurring in this area.

The CRTC has made significant strides in telephone-hearing aid compatibility, message relay systems, captioning of television and, most recently, accessibility of the information highway to persons with disabilities. Since most of these issues have been addressed through an adjudicative rather than standard creating mechanism, progress has been slower and less comprehensive than in the area of transportation. As well, the decisions have tended to follow and reflect legislative decisions made by the United States Congress.

#### D. Access to Enforcement

*Public enforcement* of standards tends to be the most comprehensive. This occurs most often where standards are specific and can be addressed prospectively through a licensing, permit or plan approval process. Reliance on a public enforcement mechanism is least likely to be successful where the standard tends to be a conditional formula, as is the case under the federal *Employment Equity Act*, or human rights legislation.

Canadian human rights commissions all have authority to initiate complaints, but driven as they are by backlog management, they tend not to exercise this jurisdiction. As a consequence, cases of systemic importance tend not to be litigated in Canada. In the United States, the Department of Justice has enunciated priority areas in which it is committed to achieving specific goals. It will only take on cases which are consistent with these goals, and only as many additional cases as can be handled within its allocated budget. Often these goals are systemic in nature. Notwithstanding public perception to the contrary, significant change has occurred despite the relatively small number of cases going to litigation (Dept. of Justice, 1996). A large number of cases are resolved without the need for litigation. These settlements receive publicity comparable to court judgments and are assumed to accurately reflect the law.

In the United States, the public enforcement bodies can permit private litigation to occur as an alternative to the public process. In the case of employment, individuals are free to pursue private remedies without even going to the Equal Employment Opportunity Commission (EEOC).

Exclusive reliance on *private litigation* can create access problems and preclude effective remedies unless the process is developed with care. The CTA has developed a user-friendly and credible dispute resolution process which resolves cases within 120 days, without the necessity of retaining counsel or conducting an oral hearing in most cases.

The CRTC on the other hand, conducts a very formal and resource intensive process. Only the most articulate and well-informed person with a disability would be able to pursue a systemic remedy without a significant expenditure.

*Incentives* can play an important independent or complimentary role in barrier removal. In Canada, the *Income Tax Act* provides in s. 20(1)(qq) a limited acceleration of the deduction of capital expenditures. In the United States, the *Internal Revenue Code* s. 44 provides a generous tax credit for a wider range of eligible accommodation expenses, in addition to the s. 190 accelerated deduction upon which the Canadian deduction was based. Uptake on these incentives appears to be directly related to the rate of legislated barrier removal, suggesting that legal enforcement is the primary motivator, although reduced cost may help "the medicine go down".

# E. Evaluating the Impact

The evidence indicates that barrier removal is slow to non-existent without legislative action. The inter-city bus industry is the only major transportation mode in Canada not subject to legislated standards, notwithstanding virtually zero accessibility. Voluntary measures have not worked. The industry has been under threat of legislative intervention for six years with negligible impact. Not surprisingly it is the subject of most criticism by persons with disabilities (Goss Gilroy, 1995c).

There is considerable American evidence that implementation of legislated standards that are:

- specific and clear;
- developed following consultation with service providers and persons with disabilities;
   and
- enforced following a phase-in period during which technical and legal advice is available;

has resulted in significant progress with barrier removal (GAO, 1993a, 1994a; Gallup, 1992; UCPA of A, 1993; West, 1991, 1994, 1995). Of particular interest are the Regional Information Centres, which are regarded as credible and useful by American business. No comparable expertise is available to Canadian business outside the employment area. Likewise, no comparable Canadian success in removing barriers can be demonstrated.

There is also considerable evidence that American businesses have shifted from opposing legislation to supporting it. For example, trade associations by and large have worked cooperatively to implement the *Americans with Disabilities Act* rather than lobbying to have it

weakened or repealed (Foley, 1994). A majority of senior human resource managers found it had a positive impact on hiring decisions (Global Strategy Group, 1966). Senior executives expressed overwhelming support for the ADA, with more supporting strengthening than weakening it (Louis Harris, 1995).

Furthermore, there is also evidence of support among the public for barrier removal, including a clear willingness to pay the costs of barrier removal in a wide range of service areas (Louis Harris, 1991). A comparable survey is about to be conducted by Louis Harris in Canada.

A survey of persons with disabilities indicates a remarkable 96 per cent not only knew what the ADA was, but agreed it had already made a difference in their lives, only four years after implementation (UCPA of A, 1996). Disability organizations share this belief (National Council on Disability, 1995). The perception that change is occurring amongst persons with disabilities is considered as significant as the changes themselves (Blanck, 1995). People who had given up hope in the face of barriers they considered insurmountable, are demonstrating renewed enthusiasm to enter the mainstream (National Council on Disability, 1995, 1996).

Studies in the United States have demonstrated the *ADA's* positive impact on barrier removal by private services available to the public (GAO, 1993a, 1994a), public school boards (GAO, 1995a), communications (Strauss, 1993; Blue Ribbon Panel, 1994), the federal government (Mendelsohn, 1995) and state and local governments (Pfeiffer and Finn, 1995).

# F. Employment: A Special Case

Employment is somewhat different from other barrier removal issues because:

- Employment decisions are particular to the individual employer, employee and job situation and therefore pose special problems when applying general standards;
- The issue has a close relationship with other social welfare and labour market issues (who bears the cost of disablement and do policy makers care about the lost productive capacities of people with disabilities?); and
- A person's ability to be employed depends on the removal of barriers in a number of other areas (e.g. training and education, transportation and housing) which are not necessarily within an employer's control.

For these reasons the issue merits special attention.

Virtually every industrialized country has a legislative employment program for persons with disabilities. They generally fall into two groups:

• Over-arching anti-discrimination legislation (such as the United States, Australia and Canada); and

• Labour market policies designed to ensure a pre-determined level of labour force participation (most other industrialized countries with Germany the most successful prototype).

The United States has the most highly developed and longest running anti-discrimination policy. There is also more evidence of success than in Canada. The *Rehabilitation Act of 1973* established the principle of non-discrimination and "the duty of accommodation". Prior to passage of the *Americans with Disabilities Act*, the law appears to have been ". . . of most value to people with disabilities who are otherwise well-prepared to compete in the labour market [that is, those with good educations] or purchase goods and services in the product market [the well to do]; it will do little good and may do harm to the doubly handicapped (Burkhauser et al, 1982)."

Overall, the labour force participation of persons with disabilities declined marginally from 1973 to 1990, demonstrating extreme sensitivity to fluctuations in generic employment rates. It is likely that much more dislocation would have occurred in the absence of legislation, since this was a period of restructuring. The *Rehabilitation Act* provided incentives to develop disability management programs intended to return people with disabilities to work following an industrial accident. Improved representation of some sub-groups appeared to be linked more to barrier removal in education than in employment. Qualified persons with disabilities are able to compete, with accommodation, on a more competitive basis.

While the employment rights of people with disabilities have not been substantially changed by the *Americans with Disabilities Act*, Department of Census data indicate that the employment rates for severely disabled people improved markedly between 1991 and 1994 (McNeil, 1996). This may be explained in part by the general improvement in employment rates for the population as a whole. Persons with disabilities would be expected to benefit disproportion-ately from such improvements because of their relatively marginal labour market attachment.

It must also be said that the ADA benefits identified above (clear standards, specialized enforcement, high public profile, available expert advice) are apparent in the employment field as well. Special factors would include the deterrent effect of several high profile multi-million dollar damage awards and unrestricted access to the courts (as opposed to discretionary screening by public enforcement agencies). The assurance of a hearing on the merits appears to encourage settlement rather than the delaying tactics typical of human rights litigation in Canada.

The Canadian experience, based on jurisprudence and employment data generated by Statistics Canada, offers no support for a conclusion that statutory human rights legislation has in any way improved the employment prospects of persons with disabilities. This is not to suggest that some already employed people did not enjoy the benefit of productivity-enhancing accommodations which improved their prospects of promotion. Likewise it is possible that more dislocation could have occurred without the protection of human rights legislation. The conclusions reached about anti-discrimination employment legislation in the pre-ADA period in the United States apply in Canada.

One group that appears to have clearly benefited from human rights legislation in Canada is unionized employees. Arbitration enforcing human rights legislation has proven far more effective than the public enforcement mechanism of human rights commissions. Non-unionized employees and applicants for work in unionized workplaces do not benefit.

Employment equity is a Canadian hybrid. Reference to "goals and timetables" suggests a labour market approach. However, it is actually a piece of systemic anti-discrimination legislation. Data suggesting that the federal legislation from 1986 to 1996 had improved the representation of persons with disabilities, as reported by employers, has been discredited. The small number of employers reporting improved performance were demonstrated to have adopted a broader definition of disability, producing inflated results (Abt, 1992b; various Canadian Human Rights Commission reports, Roeher Institute, 1992).

Government employment is the one area where some progress could be documented during the mid- to late 1980s. However, the downsizing of government during the 1990s appears to have disproportionately affected persons with disabilities, and reversed whatever progress had previously occurred (Ontario Public Service, 1992).

Within both public and private sectors, employment equity was found to have far more beneficial impact on women and members of visible minorities than on persons with disabilities and aboriginal persons (Ontario Public Service, 1992; various Canadian Human Rights Commission reports).

Mandatory employment equity legislation was repealed in Ontario before it was implemented. New federal mandatory employment equity legislation has not yet been implemented. It would be premature to speculate on what impact it will have. This is an area where careful data collection, based on rigid application of the statutory definition of disability, would be desirable.

The labour force participation model has shifted from a system of substantial state intervention involving quotas (enforced by criminal sanctions), reserved employment and sheltered employment, to a "redistributive" (or "grant-levy) model, typified by Germany. The state intervention model still used in the United Kingdom has been discredited as stigmatizing and ineffective (Baker, 1992; Lunt and Thornton, 1993). The redistributive model has now become the standard in Europe and much of Asia, with new countries adopting the model every year, Ireland, Belgium, Indonesia being recent examples.

France shifted to the redistributive model in 1988. Its difficulties illustrate the importance of setting appropriate incentive levels and the need for effective use of technical resources in support of integration (Lunt and Thornton, 1993; Velche 1995). Japan demonstrates that the model can lead to segregation of clusters of persons with disabilities, similar to some supported work placements within private businesses in North America (Ison, 1992). In both jurisdictions, targeting incentives or integrated job placements may be deemed desirable.

The redistributive model involves identifying a target group (usually those considered severely disabled) and offering long-term incentives to employers to include them in regular employment. The system focuses on finding and maintaining employment, rather than on removing barriers to advancement within the workplace, although it often provides substantial subsidies to cover the costs of job accommodations.

"The financial measures appear to have three aims: to compensate for reduced productivity or costs associated with employing a disabled person; to provide a reward or bonus for taking on a disabled person; and to cover all or some of the costs of adapting the workplace or working environment to meet the circumstances of disabled workers (Lunt and Thornton, 1993)."

The schemes are "redistributive" in the sense that they shift resources from employers who are unable or unwilling to include disabled employees to those who are. The financial measures can take the form of a long-term subsidy (more familiar in a North American context) or the application of a tax or levy (in Germany and Japan, for example) when targets are not reached.

The anti-discrimination and labour market approaches would appear to benefit differing groups within the disabled community. Some have questioned whether this would not suggest that the two approaches are both compatible and complimentary (Baker, 1992; Lunt and Thornton, 1993; Hum and Simpson, 1993), since the needs of differing groups will vary.

By targeting labour market incentives on those who are deemed otherwise "unemployable" for purposes of establishing eligibility for disability-based income replacement programs, it would be possible to identify persons who are least likely to be stigmatized, and most likely to generate overall cost savings. For those who would find linking disability and reduced productivity to be unacceptable, the anti-discrimination model, improved by drawing on the experience of other jurisdictions, would appear to be the preferred option. A careful examination of potential cost savings to be derived from introducing a labour market incentives model in Canada and the willingness of those on disability income programs to consider participation in such a program would be helpful so that policy development could proceed on a better-informed basis.

# G. Summary: Major Findings and Lessons Learned

- Barriers, not the disability *per se*, are the major stumbling block to full participation for most people with disabilities.
- Redistributive approaches, such as used in Germany and some other countries, provide a system of permanent incentives which can permit the stable participation of the significant minority of individuals with severe disabilities whose labour market attachment is marginal.
- Clear and precise standards, such as those provided under the *Americans with Disabilities Act*, have been demonstrated to be far more effective at barrier removal than the Canadian approach involving generalized criteria and relying on litigation for enforcement. The Canadian approach is adversarial in nature, and can develop resistance rather than support for barrier removal.
- The ADA has succeeded in generating support from the business sector and the public at large, as well as among people with disabilities. It has also been successful in removing many barriers in a number of different areas.
- The ADA experience indicates that support from the business community comes after implementation, not before. Attitudes follow behaviour.
- The availability of technical assistance, such as provided in the United States to support implementation of the ADA, results in a constructive, rather than adversarial approach, and is important in generating support and facilitating compliance.
- People with disabilities did not benefit from the former federal Employment Equity Act. The new Act relies upon a conditional formula rather than providing clear barrier removal standards. Thus one can anticipate opposition and failure comparable to the experience with Canadian human rights commissions, unless detailed regulations are enacted.

# 5. Disability Income Programs: Balancing Opportunity and Security

by Harry Beatty

#### A. Disability Income Programs In Canada

There are seven principal disability income programs (if we can extend the use of the term "program" to cover personal injury tort litigation) in Canada (Beatty, 1992). These are:

- i. Social Assistance for Persons with Disabilities (SA)
- ii. Canada Pension Plan/Quebec Pension Plan Disability Benefits (CPP/QPP)
- iii. Workers' Compensation (WC)
- iv. Long-Term Disability Insurance (LTD)
- v. Motor Vehicle No-Fault Accident Benefits (MVAB)
- vi. Personal Injury Awards and Settlements (PI)
- vii. Income Tax Credits (IT).

These programs can be described as the major ones because they are the largest in terms of dollars and impact on people's lives. Employment Insurance sickness benefits, Veterans' Benefits, Criminal Injuries Compensation and related programs are important to claimants and recipients, of course, but are much smaller in scope.

Very significant numbers of Canadians with disabilities (and their families) are totally or partially dependent on these programs. The 1991 Health and Activity Limitation Survey reported that 683,000 Canadians with disabilities between ages 15 and 64 residing in households received some form of disability-related income in the reference year. In addition, nearly 300,000 received a medical expense credit, and more than 150,000 received the Disability Tax Credit (Statistics Canada, 1994). As shall be seen, there may be reason to believe that HALS underestimated these totals (they did not include persons living in institutions to begin with), and in any event the numbers dependent on disability-related income have increased significantly since that time. Including the dependants of persons with disabilities, it seems likely that in 1997 more than a million Canadians depend wholly or partly on disability-related income.

A full consideration of disability income systems in a society would require looking at them in the context of all other disability-related programs and benefits, including:

- · assistive devices programs
- drug plans
- attendant care programs
- · employment programs
- · transportation systems
- housing programs

and many more. This is because essentially the same type of need may be met through an income program, a health needs program, or a service program, depending on how they are designed. For example, an individual may receive an allowance to pay for prescription drugs, or be eligible to participate in a drug plan open to the general public, or receive funding for drugs from a disability-specific program. He or she may receive a shelter subsidy through an income program, or pay reduced rent through rent-geared-to income, or live in a specialized residence for persons with disabilities. In each case the government subsidy and the support provided to the individual are essentially the same, although the mechanisms are very different and may or may not be part of, or linked to, a disability income system. (This is a major reason why meaningful comparisons of disability income programs in different jurisdictions are so difficult to make.)

Following is a discussion of each of the seven disability income programs listed above, with emphasis on their major strengths and weaknesses, as identified in evaluations and studies. (As evaluations and studies, whether by governments, business, unions, insurers, or disability advocates, usually arise because of perceived problems, it is much easier to find weaknesses than strengths from the literature.) A detailed description of the administration, funding and rules governing each system is beyond the scope of this paper. Sources of useful descriptive information, however, will be identified where appropriate.

# i. Social Assistance for Persons with Disabilities (SA)

#### **Description**

Social assistance falls under provincial/territorial jurisdiction, and there are considerable differences from province to province as to how social assistance for persons with disabilities is structured. But there are significant commonalities as well, largely as a result of the *Canada Assistance Plan* which was in operation from 1966 to 95, and under which guidelines were established as part of the federal/provincial/territorial cost-sharing arrangements.

Social assistance is basically a needs-tested program, under which the needs of an individual are compared with his or her financial resources (assets and income). It is a "last resort" program for those who do not have other financial resources to meet their essential needs. As such, it is a secondary payer to all other programs. Because of its "last resort" status, typically social assistance is "fine-tuned" to assess in some detail both the needs and the financial resources of the person. This leads to detailed and complex rule-making about income, liquid assets, earnings exemptions, special needs, shelter costs, and so on. The result is often a complex set of rules that is very difficult to understand and administer fairly - a sort of "income tax in reverse for the poor".

#### **Information Sources**

According to the HALS (Statistics Canada, 1994), there were 271,100 persons with disabilities in Canada aged 15 to 64, residing in households, and receiving "social assistance or welfare".

This may, however, have been an underestimate. The total listed for Ontario, for example, was just under 80,000. The Report of the Disability Determination Project Team, part of the Advisory Group on New Social Assistance Legislation process (Advisory Group on New Social Assistance Legislation, 1992) showed the Ontario Family Benefits GAINS-D caseload as passing 100,000 in 1990, with approximately 30,000 more in the "ill-health" category under general (municipal) welfare. Some of the difference may be explained by persons in institutions receiving GAINS-D, but as HALS gave a figure of just over 16,000 for the institutionalized group aged 15 to 64, there is still a discrepancy. We might hypothesize that many people with disabilities and their families do not necessarily identify Family Benefits GAINS-D as "social assistance" or "welfare" - they call it an "allowance" or a "pension". In any event, HALS appears to give a significant underestimate of social assistance disability recipients in Ontario.

The Report of the Disability Determination Task Force just referred to contains a significant overview of the Ontario system. For example, there were a number of findings regarding the caseload (as of 1992). Single persons comprised over 75 per cent of the cases: families under 25 per cent. Only about 2 per cent of the cases were related to accident or injury: the rest were related to congenital disability, illness or disease. (The low incidence of accident or injury cases was attributed to the wide workers' compensation and automobile no-fault coverage in Ontario.) At that time, 16 per cent of cases received CPP as well and got Family Benefits GAINS-D as a "top-up". Only 2 per cent, however, were getting GAINS-D as a "top-up" to Workers' Compensation.

Only 9.5 per cent of the cases, including both single persons and families, reported any earnings at all (including earnings by other family members in the family cases). The average duration on the program was 7 years.

Disabilities were almost equally divided between mental impairments (45 per cent) and physical impairments (55 per cent). Psychiatric disorders were 25 per cent and developmental disabilities 21 per cent, and the number of learning disability cases was negligible (unless the individuals qualified under some other diagnosis or label). Half of the physical impairments were due to musculoskeletal or nervous-system disorders.

Men were 55 per cent of the caseload, while women were 45 per cent. (This was not explained in the report - a possible reason is that women are more likely to be dependent on a spouse than are men.) Those over 45 outnumbered those under 45.

The Report showed an increase in the GAINS-D caseload at a rate between 5 per cent and 7.5 per cent between 1982 and 1992. The factors listed as contributing to this caseload increase were:

- · aging of the population
- · increase in the Medical Advisory Board's approval rate from 40 per cent in 1984 to 63 per cent in 1991 (the MAB was decentralized, implicitly recognizing conditions in the

local economy; new conditions were recognized, such as environmental hypersensitivities; the Social Assistance Review Board reversed a high per percentage of negative decisions on appeal)

- · increase in the success rate of appeals at the Social Assistance Review Board from 24 per cent in the early 80s to 71 per cent in 1991-92
- deinstitutionalisation, which helped to increase the caseload of persons with psychiatric disabilities
- an August 1991 Ministry guideline stating that socioeconomic factors were to be considered together with medical conditions
- pressure on GAINS-D to accept clients due to the low levels of general (municipal) welfare, and due to municipalities seeking to avoid costs by moving clients to the provincial system.

Caseloads have continued to grow to close to 160,000 on GAINS-D in 1996 and close to 40,000 on the "ill-health" component of General Welfare, according to an internal Ministry document provided to the author.

The Advisory Group identified principles which should govern the definition and determination of disability, related to fairness and efficiency. Some of the points emphasized were: that the social assistance system should contain a "disability top-up" which was not necessarily permanent; that persons with cyclical illnesses (like multiple sclerosis or manic-depressive illness) should not be excluded unfairly by duration requirements; that the disability definition should not be a barrier to self-sufficiency; that the determination process should be equitable to persons with all types of disabilities, and that there should be an interdisciplinary approach to determining eligibility.

The Advisory Group also set forth a brief discussion of comprehensive disability income reform proposals.

There is very little meaningful comparative information on social assistance for persons with disabilities in Canada. Comparisons of maximum rates, for example, tell us very little about the real effect of the programs. The Disability Determination Task Force provided a brief comparison. One study which goes beyond this from a critical consumer perspective is Sherri Torjman's *Income Insecurity* (Torjman, 1988). Aside from Ontario, Alberta and British Columbia have what are in effect separate social assistance programs or entitlements for persons with disabilities. Torjman includes New Brunswick in this group as well.

British Columbia has recently gone through a process of reviewing its program and adopting a new definition. This was largely in response to representations from organizations representing

the disability community: major disability organizations had developed a position advocating a functional definition of disability which would not serve as a barrier to independence (Ad Hoc '95).

# **Strengths**

The most important positive feature of social assistance for persons with disabilities is that it is available to all adults with disabilities resident in Canada as a last resort. It is the last line of defence against extreme poverty for this community. In many provinces and territories, the SA program for persons with disabilities is more adequate than SA for other disadvantaged groups.

The detailed rule-making available under social assistance arguably allows difficult questions of equity, need and social policy to be addressed, such as support for dependents, exemption of the principal residence as a liquid asset, earnings exemptions, and many more. As a needs-tested program SA is highly "targeted", so its resources can be seen as directed to those who need them most. For the same reason, SA eligibility can be used as a criterion of eligibility to other essential programs - drug and dental costs, rent-geared-to-income housing, and legal aid, to take three examples. (Of course, disability organizations typically argue for *universal* access to programs, without taking income or assets into account, for reasons discussed below under "Weaknesses". However, cost-conscious governments at all levels are moving in the opposite direction - for example, the federal Government with respect to both childrens' and seniors' benefits).

The costs of administering social assistance tend to be very modest - less than 10 per cent of the total costs of the program. However, there may be false economies here - in Ontario, workers may have caseloads of several hundred, which has led both to inadequate monitoring of cases (so that those who are really ineligible receive benefits) and to delays and lack of support which have adversely affected many clients.

# Weaknesses

The major weaknesses of SA as a disability income program are essentially inverses of its strengths. Because it is a "last resort" program, it is subject to an expanded caseload, particularly during a recession, as people have no where else to turn (as was found by the Disability Determination Task Force discussed above). This is especially true of "over 50" workers with health problems, who often have no realistic prospects of employment. The expanded caseloads lead, in turn, to cost pressures on the program, and to proposals to limit eligibility either through changes in definition or in the determination process and guidelines. The result may be that people who need SA may find themselves ineligible, and be forced onto general welfare programs as a last resort.

Other weaknesses have been identified by analysts such as Torjman (1988).

The complexity of SA makes it difficult to understand, both for clients and government officials, and extremely difficult to administer. The detailed rules aimed at fairness may wind

up creating the opposite result, where two applicants or recipients in essentially the same circumstances are treated very differently (where they have "outside" income from a different source, for example).

Most significantly, the "targeting" under SA too often means that individuals are penalized for efforts to become more independent and self-sufficient. The SA disability recipient who works will have his or her allowance reduced (usually by 75 per cent of earnings or more), and, even worse, risks losing his or her "categorical eligibility". If someone tries to work and then loses the job for whatever reason, he or she may be much worse off than someone who never tried to work at all. The work disincentive in SA is probably its most criticized feature. But other efforts to become more independent - from saving to buying insurance - may also be penalized by a partial or total loss of benefits under SA.

Torjman discusses the "Catch 22" situation created by a "permanently unemployable" test of eligibility. On the one hand, recipients are required to define themselves as "unemployable" in order to at least partially meet the costs of living with their disabilities. On the other hand, it keeps them "on the system".

# ii. Canada Pension Plan/Quebec Pension Plan Disability Benefits (CPP/QPP)

# **Description**

CPP and QPP disability benefits are nearly universal programs for employees and self-employed persons in Canada, with minimum requirement of contribution of 2 of the last 3 years or 5 of the last 10 under the current plan, and 4 of the last 6 years under the new CPP proposals. Under CPP/QPP, disability pensions are paid to former contributors who, because of a "severe and prolonged disability" are "unable to regularly pursue any substantially gainful employment". The eligibility of contributors turns on their history of contributions to the CPP/QPP, including actual contributions made by the individual, "deemed" contributions through credit-splitting, and contributions made to the national pension plans in other countries with which Canada has a social security agreement. The amount of the pension is based on a flat-rate component and a component based on contributions. There is also a benefit for dependent children of the disabled contributor. CPP is a primary payer from which other disability income payments are not deducted (QPP, however, is a secondary payer to that province's workers' compensation system). There is no financial eligibility test in CPP/QPP.

#### **Information Sources**

CPP/QPP disability pensions are described fully in pamphlet publications put out by the federal and Quebec governments. An intensive evaluation of CPP, including several background studies, has been undertaken in the past two years by the Government of Canada (HRDC, 1995c; HRDC, 1996a). The results of the evaluation are fully documented in the Final Report of the Evaluation and will not be repeated here.

### **Strengths**

The Final Report of the HRDC evaluation cited identified a number of strengths of the CPP disability pension program. These include:

- · Virtually universal coverage to employees and the self-employed.
- Minimum protection for those who might not be able to afford private coverage.
- · An important source of earnings for disabled persons.
- · Relatively constant adjudication of applications for benefits.

These are all important points. From the consumer perspective, the fact that CPP is available without an income or asset test (either individual or family based) and that it is not reduced by other payments is important. An example is someone who becomes ill but does not have LTD or WCB coverage, and who has a spouse with a modest income or assets which makes the family totally ineligible for SA. Many who become unable to work, especially those previously in lower-paying, contract or self-employed jobs without benefits, are eligible *only* for CPP. This group includes a disproportionate number of those in groups traditionally disadvantaged in the workforce, including women, persons already disabled, visible minorities, and First Nations people.

CPP/QPP disability is also relatively straightforward, compared to many others, to understand and administer (even though caseload increase has caused significant problems for the Government of Canada). There is no income or asset testing, no assessment of need, and no determination of family status involved. Pension levels are calculated using Revenue Canada data, as are contributions (although the additional rules relating to contributions - child-rearing drop-out, credit-splitting, contributions in other countries with which Canada has a Social Security Agreement - bring a considerable measure of complexity into the system).

The costs of administering CPP are, like other public systems, relatively low in relation to benefits. The CPP evaluation conducted by HRDC would seem to imply that some additional resources would facilitate a better administrative process. From the consumer's perspective, it would also address the delays that cause problems for individuals.

#### Weaknesses

The universality of CPP disability benefits coverage has been reduced by the recent (1997) introduction of legislation calling for a more restrictive eligibility rule (from 5 of the last 10 years, or 2 of the last 3 years, to 4 of the last 6 years). While this can be argued to be fairer, requiring a more substantive attachment to the work force, it also tends to exclude those who may be in most need of disability income protection - new workers, contract workers, those who have been unemployed, and those who have interrupted work histories due to cyclical or progressive illnesses.

The HRDC evaluation concluded that the "CPPD administration has been relatively constant in its adjudication of applications for benefits" and that "there is no evidence that more generous

CPPD adjudication resulted in inappropriate increases in grants". Nevertheless, the evaluation also identified as a potential problem CPPD's "scope for discretionary judgment". It is clear that categorical eligibility for CPP disability has already been narrowed by administrative action. This is positive from the cost-limitation viewpoint, but there is concern in the community of people with disabilities that those who realistically cannot work will nevertheless be found ineligible. The Ongoing Case/File Review, as recommended by the evaluation, would be an important tool for monitoring the consequences of the new administrative approach.

The HRDC evaluation found, based on a Statistics Canada survey of CPP disability beneficiaries who responded (although the response rate was not as good as could be hoped), that "a significant proportion (10 per cent) may have the potential for vocational rehabilitation and return to the work force". As discussed in Section 3.B (Disability Management), the review of the National Vocational Rehabilitation Program showed the same. There is little doubt that the rehabilitation initiatives under CPP could be strengthened, and that this would generate considerable cost savings. However, from the consumer perspective, there is great concern that a person who undertakes rehabilitation or return to work, but is unable to keep a job, will nevertheless be found categorically ineligible. From this all-or-nothing approach, the design and regulations of CPP disability benefits has a very significant work disincentive effect. This is an important contributing factor to its low return-to-work percentage among recipients (less than 1 per cent annually, although virtually every CPP disability recipient has been in the work force and many have very extensive employment histories).

Like SA, CPP/QPP is susceptible to caseload increase in times of recession, especially among older workers with health problems who are effectively excluded from the job market. This was clearly identified in the CPP evaluation.

# iii. Workers' Compensation (WC)

#### **Description**

Workers' Compensation is a broad, but by no means universal, system of disability income protection for employees covering accidental injuries or occupational diseases arising out of employment. It is within federal, provincial, or territorial jurisdiction, depending on the business or industry, although the federal government covers only about 10 per cent of employees and the administration of WC for federal employees is often delegated by agreement to the provincial WC administration. Depending on the jurisdiction, specific businesses or industries may be listed as included or excluded from coverage. The coverage is largely compulsory and on a no-fault basis, although there are exceptions (such as optional coverage for some businesses, limited or no coverage for employees who are injured because of their own serious misconduct). Benefits vary greatly among jurisdictions but typically include income benefits (temporary and long-term, total or partial), rehabilitation, medical care and other health needs, death and survivor benefits.

#### **Information Sources**

A text that covers Workers' Compensation in detail is *Workers' Compensation in Canada* (Ison, 1989). It covers the common features and differences among the various provincial schemes.

Statistics Canada publishes data on work injuries, based on WCB information. For 1994, a rise in work-related time-loss injuries was reported after four previous years of decline. The total number of claims in Canada for 1994 was 429,000. The increase was attributed basically to an increase in number of hours worked and an increase in the size of the work force. These data, however, do not reflect the number of long-term injured workers.

The range of issues involved in the Workers' Compensation system were reviewed in depth in Ontario in the early to mid-1980s by Professor Paul Weiler of Harvard University. His series of three reports prepared for the Ontario Government (Weiler, 1980, 1983, 1986) are an excellent starting point for an evaluation of the WC system. Weiler recommended the following:

- a higher level of earnings coverage so that the program would be more nearly universal;
- a dual award system, with one component based on wage loss and the other on noneconomic loss (pain and suffering), to reflect the real losses due to disability;
- a strong emphasis on rehabilitation and return to work, including a right to return to work (subsequently enacted in Ontario as in Quebec, but in few other jurisdictions);
- an independent appeals process with support for independent advocacy; and
- a strong on-going program of prevention of workplace injury and disease.

The broad outlines of Weiler's proposals, while controversial, were accepted in Ontario. In his second report (Weiler, 1983), he advanced a thoughtful analysis of the prospects for a comprehensive disability compensation plan. He also dealt in depth with occupational illness and with prevention. In his third report (Weiler, 1986), he examined the issues related to the compensation of *partial* disability in great depth.

In 1995, the Queen's University Industrial Relations Centre released a volume of research studies on WC (Thomason and Chaykowski, 1995). This contained rigorous studies based on data from WC Board in Canada, and compared the results to United States studies in particular.

One interesting study (Gunderson, Hyatt and Law, 1995) looked at re-employment and accommodation requirements under WC. They found that re-employment requirements, such as those enacted in Ontario, are "quite rare". These requirements are not found in the United States and are not enforced in the European countries which have them. They are really a "subset" of human rights accommodation guidelines.

The authors review several studies (mainly United States studies which show there are "potential adverse work-incentive effects of high income replacement" as exist in WC, unlike most other programs). However, these studies are inconclusive as to the *actual* impact on individuals' decisions to return to work if appropriate rights and accommodations are provided. A second study in the same volume (Thomason and Pozzebon, 1995) shows from Canadian data (like several United States studies they cite) that increasing WC levels increases the likelihood of a claim (based on national WC data).

The authors then look at the cost of accommodations, as an alternative to income support, but are unable to draw firm conclusions based on available evidence. They state that "unfortunately, there is little systematic evidence, especially in Canada, of the cost of accommodation". It is very difficult to study this issue accurately, according to the authors, as some accommodations may be "no cost" (such as flexibility of other employees in sharing or trading job responsibilities) and other accommodations may simply not be documented because they are seen as too "high cost" and are never carried out.

An important question is whether the costs of accommodation are born by employers (or their customers), or shifted to employees with disabilities (usually in the form of lower wages and benefits). The relevant Canadian study, according to the authors, is a PhD dissertation (Hyatt, 1992) based on Ontario WCB statistics. The authors say that Hyatt's results were somewhat as expected. Where the injured worker returns to the accident employer, there is no wage reduction (in fact, an average 1 per cent increase!), presumably reflecting both rules and perceptions which would make it unacceptable to impose a pay cut. On the other hand, where the individual with a disability returns to *another* employer with no prior responsibilities, the wage reduction is around 30 per cent, three-quarters of this reflecting reduced physical demands (that is, the person can now only do a less demanding job), and one-quarter reflecting the cost of workplace accommodations (the employer has to accommodate the person so pays less).

Another study in the same volume (Allingham and Hyatt, 1995) shows a link between vocational rehabilitation services and return to work for injured workers. However, the authors state that real effectiveness is difficult to measure because of the "estimation strategy" (basically, that the workers who pursue vocational rehabilitation may have been better prospects to return to work anyway and have self-selected).

An Ontario Task Force (Workers' Compensation Board, 1987) provided detailed recommendations on vocational rehabilitation within the WC system.

# **Strengths**

Workers' Compensation provides a moderate to relatively high level of income replacement (a maximum of \$17,000 to \$35,000 per year, depending on the jurisdiction) to injured workers considered to be *totally* disabled either temporarily or permanently. The income benefits are tax-free. There is also a range of additional rehabilitation and health benefits.

Of the dollars collected in WC premiums, approximately 90 per cent are spent on benefits, rather than administration.

WC premium-rating systems, which charge employers in higher-risk industries or with poor safety records more for coverage, can play a role in encouraging prevention.

In some jurisdictions, such as Ontario and Quebec, some success has been achieved through return-to-work legislation supporting return to work to the employer where the accident occurred.

### Weaknesses

From the employers' and administrators' perspective, the rate of earnings replacement for total disability is too high to encourage return to work. The research reports referred to above provide some support for this position, although it is somewhat inconclusive. WC clearly provides a higher level of income replacement than do most other disability income programs. Ontario has recently reduced the level of income replacement, and this might provide an opportunity to study the impact on injured workers and return to work.

WC has been funded more on a pay-as-you-go than a full actuarial basis in some provinces, which has led to a significant unfunded liability. From the employers' perspective, this is an unfair transfer of costs from past claims to future years, while from the injured workers' viewpoint, this leads to significant pressures to reduce benefits, which has occurred in recent years in several provinces.

Both employers and workers have been critical of the rehabilitation programs of the WC Boards in a number of Provinces. An underlying problem is that the WC Board delivers both income replacement and rehabilitation, which may lead to a confusion of objectives. The 1987 Ontario Task Force looking at vocational rehabilitation within the WC system emphasized that "the structure of the WCB must be changed to separate the function of claims and rehabilitation and rehabilitation must be made a full partner in the process" (Workers' Compensation Board, 1987, p. 62). Otherwise there is at least a *perception* on the part of the injured worker that rehabilitation is being offered as part of a claims-control process - hardly a sound basis on which to build an effective vocational rehabilitation plan. WC (and other disability income program deliverers) must of course be heavily involved with rehabilitation, having a strong interest in its success. But the rehabilitation program must be seen as relatively independent of claims adjudication.

From the injured workers' perspective, a major problem is that those with long-term partial disabilities often receive very low pensions which do not really reflect their disadvantage in the labour market. This is especially true of older workers who are unable to continue at strenuous labour, but cannot adapt easily to other employment because of factors such as limited education and language barriers.

The difficulty of proving a causal relationship between workplace environmental factors and occupational illnesses is also a significant problem (as discussed in depth by Professor Weiler in the study referred to above). For example, the link between substances in the workplace and cancer is very difficult to establish in practice, even though there may be a theoretical basis for it, because decades may intervene between the exposure and the disease it may have caused. Of course, this creates a funding problem for WC Boards as well, since significant recognition of occupational diseases would add to the unfunded liability problem.

### iv. Long-Term Disability Insurance (LTD)

### **Description**

Long-term disability insurance is most commonly available in Canada as a benefit provided by employers on a group basis. The coverage is usually provided by an insurance company, although some large employers self-fund their plans and pay an insurer to administer the plan. Coverage can also be purchased on an individual basis or through professional groups.

LTD provides coverage for disability arising from any cause during the contract period, most commonly on a "permanently unemployable" basis. A typical clause provides coverage for someone who is unable to do his or her "own employment" during the first two years of disability, but thereafter only for someone who is unable to do "any employment for which he or she is reasonably qualified by age, training and experience". Earnings replacement is in the range of 60 to 70 per cent of pre-disability earnings, commonly without indexation (except in the government sector). A common additional provision is a "cap" in the 85 per cent range on disability income from *all* sources. LTD is almost invariably a secondary payer to CPP disability and WC. Some policies, but by no means all, have rehabilitation and partial disability provisions. LTD payments are taxable where employers pay the premiums but non-taxable where employees or individuals pay the premiums.

LTD falls under provincial and territorial jurisdiction. Unlike government programs and automobile insurance no-fault benefits, however, the provisions of LTD coverage are not regulated closely by government — employers and insurers have considerable latitude in determining the coverage provided. The Canadian Life and Health Insurance Association (CHLIA) is active in developing LTD guidelines, but policies vary considerably.

### **Information Sources**

Because of its variability, there are few detailed resources available with statistical information about LTD. Some basic facts are available from the Canadian Life and Health Insurance Association - for example, it publishes a basic pamphlet on LTD (Canadian Life and Health Insurance Association, 1992).

Some overview information is contained in articles in periodicals such as *Benefits Canada*, which conducts an Annual Review of Group Insurers. The Benefits Canada Annual Review for

1995 (Bak and Dempsey, 1996) surveyed 44 insurers, and found an 8.1 per cent annual increase in "group health" premiums (the term "group health" appears to cover extended health benefit plans as well as LTD). The article, however, goes on to describe the major concerns to insurance providers as being "skyrocketing" claims, especially in late 30s and early 40s workers with non-physical disorders, according to one company executive. Stress, chronic fatigue syndrome and depression are identified as particularly increasing claims. As well as cost containment, cost shifting from the public to the private sector and taxation of benefits are identified as important issues for insurers (although the latter presumably relates to extended health benefits rather than LTD).

The 1991 HALS survey reported 32,615 Canadians with disabilities receiving income from a private disability insurance plan, and 71,715 from an employer disability insurance plan.

Professor Marvin Baer of the Faculty of Law, Queen's University, recently completed a (draft) study of the legal issues involved in long-term disability insurance for the Ontario Law Reform Commission (Baer, 1996), but as the OLRC is now defunct the status of this study is uncertain. Professor Baer made a number of recommendations directed at better protection of clients in the system. One important recommendation was that an Alternate Dispute Resolution system be introduced so that claimants involved in disputes with insurers would not necessarily be involved in the expense of litigation.

A good overview description of LTD and issues related to its use and effectiveness is found in *Illness and Disability in the Workplace* (D'Andrea, Corry and Forrester, 1996). Some of the case law up to 1992 has been covered by Michael Kelly in his article "Taking on the Terminator: Issues to Consider When Suing A Disability Insurer" (Kelly, 1992).

### **Strengths**

LTD is flexible — employers can negotiate coverages that are suitable for their employees.

LTD is the primary system available to provide an adequate level of income replacement to higher-income employees for disability regardless of cause. This is its overriding importance to employers, employees, and the self-employed.

Some LTD insurers have been innovative in their use of rehabilitation.

LTD is essentially on a "full-funding" basis, so under funding is less of a problem than in government plans (although insurer insolvency can result in recipients losing some of the value of their payments, as happened in the Confederation Life case).

### Weaknesses

LTD is unavailable or practically unaffordable to many in the workplace. Those in lower paying or contract positions typically do not have LTD. Small businesses cannot afford it. In our changing economy, LTD coverage is becoming available to fewer and fewer employees.

Unlike government programs such as CPP/QPP, LTD excludes or limits coverage for those with "pre-existing" disabilities, or makes coverage available only at a higher cost. This limits the employment and re-employment prospects of persons with disabilities. For example, someone who acquires an ongoing progressive disability, such as multiple sclerosis, while with one employer cannot risk changing employers, since disability due to multiple sclerosis would be a "pre-existing condition" exclusion under the new employer's LTD plan.

The lack of inflation protection in most LTD plans is a major problem for younger employees who become disabled, especially when combined with a modest level of total earnings replacement and taxable payments (which is most common for employees).

From the claimant's perspective, if a dispute arises with the LTD insurer it is very difficult to pursue a claim. It is necessary to sue the insurer civilly, which costs significantly in legal fees and delay, just when the person has already incurred a loss of income and, in many cases, increased personal expenses due to disability.

Some insurers take a very restrictive approach to rehabilitation - although, as already noted, others are very progressive.

## v. Motor Vehicle No-Fault Accident Benefits (MVAB)

### **Description**

It is difficult to describe MVAB generally in Canada because there is so much variation amongst provinces and territories. Compensation for motor vehicle accident injuries is balanced in different ways between MVAB and personal injury lawsuits (discussed next), depending on the province. Quebec, alone of all the provinces, has replaced personal injury tort litigation entirely with a public MVAB. Alberta and the Maritime provinces still have primarily a personal injury litigation system, with limited MVAB for those who can't sue a negligent party. Ontario, and to a lesser extent, Manitoba and Saskatchewan, have produced "hybrid" systems which combine MVAB and tort litigation in various "balances".

Automobile insurance is public in British Columbia, Saskatchewan, Manitoba and Quebec, and private in the other provinces and territories. As may be seen from the above, there is no necessary link between public insurance and a "no-fault" approach, although in general they may be seen to be more compatible.

MVAB coverages are typically defined in detail by the provincial governments responsible, although there may be optional coverages. As well as income replacement, rehabilitation, health and care costs are typically covered, although much more under MVAB in provinces like Quebec and Ontario than in the provinces which still have primary reliance on personal injury litigation.

### **Information Sources**

The Insurance Bureau of Canada (IBC) is the major source of information regarding automobile insurance. Its publication "Facts of the General Insurance Industry in Canada" summarizes the coverages available in the provinces and territories (Insurance Bureau of Canada, 1996). This publication shows an overall picture of rapidly increasing costs for injuries due to motor vehicle accidents. Total claims throughout Canada for private insurers increased from \$2.657 billion in 1983 to \$7.173 billion in 1995, on a more-or-less steady basis. This includes property damage as well as personal injury, but personal injury is the largest component and is increasing the fastest. Private insurers in 1995 had close to 110,000 no-fault accident benefits claims. No separate statistics are provided for third-party bodily injury claims, but Appendix "B" of the publication shows a sharp increase in third-party liability claims during 1991-1995, of which bodily injury is a major component. The IBC summary does not include the government insurers in British Columbia, Saskatchewan, Manitoba and Quebec.

The 1991 HALS reported only 12,205 Canadians with disabilities receiving payments from Motor Vehicle Insurance, but it is difficult to accept this as accurate given the figures just cited. As noted in the next section, HALS did not cover third party liability settlements and awards, whether arising from motor vehicle accidents or otherwise (Statistics Canada, 1994).

An excellent summary of the policy issues involved in different models of automobile insurance is found in *The Crisis in Insurance Regulation* by Banks McDowell (1994). Although this is an American study, the author covers the Ontario and British Columbia systems. McDowell advocates a minimum or "first" level of coverage provided by government insurance and excess coverage provided by private insurers on a "second" level (optional to clients). He reviews briefly and clearly the issues involved in balancing tort/no-fault and government/private, the two "great debates" in automobile insurance (often confused by the public).

Two major Canadian automobile insurance texts are Brown (1988) and O'Donnell (1991).

There have been numerous detailed studies of the automobile insurance system in Ontario and of the various options. Of particular importance are the "Report of the Inquiry Into Motor Vehicle Accident Compensation in Ontario" (Osborne, 1988) and the Ontario Automobile Insurance Board no-fault reference hearings (Ontario Automobile Insurance Board, 1989), each of which examined the issues in great depth. (The Ontario Government, however, moved more in a no-fault direction for automobile insurance than either of these studies recommended, primarily because each study had underestimated the rapid cost increases that were occurring in third-party litigation.)

### **Strengths**

MVAB is generally assessed in relation to personal injury litigation, with the perspective of the assessor on tort influencing the result greatly.

Supporters of a "no-fault" approach argue that benefits are provided much more quickly, and with less administrative costs, than under personal injury litigation. Typically there is a tribunal or alternative dispute resolution (ADR) system for MVAB, which is more accessible than the courts. (This position was very strongly advanced, for example, by the Consumers' Association of Canada in the Osborne inquiry and the Ontario Automobile Insurance Board reference hearings.)

MVAB, which is on a no-fault basis, compensates many persons injured in automobile accidents who are not compensated in the tort system. This includes drivers who were themselves at fault in their accidents (except that there are typically limitations or exclusions in coverage for particularly serious offenders, such as drunk drivers), but also includes those whose conduct was relatively blameless but have no other driver to sue, as would be the case with a single-vehicle accident caused by bad weather conditions.

### Weaknesses

Critics of an MVAB "no fault" approach to automobile accident compensation generally approach it, as already noted, from a "pro-tort" perspective.

MVAB, with limited exceptions, compensates regardless of "fault". In comparison to tort, it tends to compensate "innocent victims" less generously and "at fault" drivers more generously, which may be less just than the personal injury tort system.

As well, MVAB may have less of a deterrent effect on bad driving than tort - although other factors such as law enforcement and driver education and licensing clearly are also closely linked to accident prevention.

### vi. Personal Injury Awards and Settlements (PI)

### **Description**

As noted in the discussion of MVAB, the largest number of personal injury cases still arises from motor vehicle accidents (except, of course, in Quebec where such cases ave been eliminated). But many other causes of injury or illness lead to personal injury tort litigation - "slip/fall" accidents, product liability, sexual assault cases, and so on. Medical malpractice actions are particularly prominent in public attention, and may lead to the largest awards or settlements of any personal injury cases, although only a relatively small number succeed in Canada.

The rules governing civil litigation, including personal injury actions, are in provincial/territorial jurisdiction.

### **Information Sources**

Little comprehensive information is available to provide an overview of personal injury litigation in Canada. Most "evaluation", as already indicated, is on a partisan basis supporting either the "tort" or the "no-fault" side of an on-going debate - and the debate is conducted almost exclusively by law professors and lawyers.

An important study related to medical malpractice in Canada was conducted on behalf of the Conference of Deputy Ministers of Health by Robert Prichard. Entitled "Liability and Compensation in Health Care" (Prichard, 1990), this report recommended implementation of a no-fault compensation scheme for health care injuries, based on a Swedish model. The right to sue for medical malpractice would not be abolished: at some point, however, the individual claimant would have to choose between the two forms of compensation. While medical malpractice litigation is only a small component in the overall picture of disability compensation, there is nevertheless a considerable amount of valuable policy analysis in the Prichard study.

### **Strengths**

The perceived strengths and weaknesses of personal injury tort litigation are really the inverse of those for no-fault disability compensation like MVAB. Briefly, the important strengths of tort are:

- Justice is seen to be done through a full hearing of the case in a court of law.
- In theory, compensation for the "innocent victim" is on a full restitution basis, in which the plaintiff is compensated fully for all disability-related costs caused by the wrongdoer (in practice, however, the level of compensation is limited by factors such as the insurance or other funding available).
- The threat of civil litigation is argued to be a deterrent to intentional wrongdoing, unsafe practices, and negligence for example, companies are forced to make their products safe by the threat of litigation.

### Weaknesses

The major perceived weaknesses of the personal injury tort system are:

- The administrative costs of the system lawyers' fees, expert witnesses, court administration and so forth are high in relation to the compensation actually provided to persons with disabilities.
- There are very significant obstacles to many persons with disabilities in accessing the personal injury tort system, especially cost (for example, the risk of paying adverse costs if the plaintiff loses) and delay.

A person with a very significant disability which is theoretically compensable in tort may
be unable to get full or adequate compensation because of limited available insurance or
other funding, or because of difficulties in proving the case, as in an accident without
witnesses.

## vii. Income Tax Credits (IT)

### **Description**

There are several provisions available in Canada's income tax system that can benefit persons with disabilities. The two principal ones are the Disability Tax Credit (DTC) and the Medical Expense Credit (METC).

### **Information Sources**

These claims are described in detail in several materials published by Revenue Canada, most notably Interpretation Bulletin IT-519R (Revenue Canada, 1995). Tax expert David Sherman reviews all of the claims in great detail in *Taxes, Health and Disabilities* (Sherman, 1995). The Disability Tax Credit was evaluated in detail by the Department of Finance (Department of Finance, 1992). Most recently, the entire range of disability-related tax claims was reviewed by the Federal Task Force on Disability Issues (1996).

Income tax is practically in federal jurisdiction - only Quebec has its own tax return - but the provinces basically share the costs of these credits through the provincial income tax that is foregone.

### **Strengths**

The DTC and METC (and the other IT claims) are administered on a cost-effective basis through the income tax system. They do not require a separate administration. Delivery through the income tax system permits widespread eligibility and accessibility of the claims.

Together, the DTC and METC provide the beginning for a comprehensive approach to recognizing the costs of disability, with DTC covering non-specific, incidental, and "hidden" costs, while METC covers identifiable and specific expenses. They are supplemented as well by the other IT claims.

#### Weaknesses

There are major issues around eligibility for the DTC. The Government is concerned to limit eligibility to those most in need (see especially Department of Finance, 1992), while disability organizations believe that eligibility is being denied those with significant disability-related disadvantages and costs.

The DTC provides the same level of payment to everyone who qualifies, assuming they can make full use of a non-refundable credit. It could be criticized from this perspective as not being effectively targeted to those most in need. However, "fine-tuning" would require more categories or some other type of individualized assessment, and the administrative costs necessarily entailed by this would probably outweigh the gains in equity, which are extremely difficult to define and measure in any event.

The METC excludes a number of items that are legitimate disability-related costs: measures were announced in the 1997-98 Federal Budget to begin to address these.

There is also a major issue around non-refundability of the DTC and METC. Disability organizations criticize this feature as denying financial support to those members of the community most in need, while Government tends to view refundability as a form of delivery of social programs - which is provincial jurisdiction - and thus not appropriately placed within the tax system.

### B. Lessons Learned From Evaluations And Reviews Of Disability Income Systems

All the disability income systems discussed in the previous section have been the subject of major policy reviews in Canada during the past decade. In most cases, significant restructuring has followed the policy reviews. Despite the scrutiny these systems have been under, however, great caution must be exercised in identifying "lessons learned" from policy reviews and evaluations.

One major reason for this caution is that detailed reviews have tended to focus on disability income systems "one at a time", without considering their interrelationships, which can give a very misleading picture of what is happening. For example, if the eligibility criteria under one disability income system are made more restrictive, people with disabilities who otherwise would have qualified for it typically wind up claiming from another system.

Two examples may illustrate this. Automobile insurance, and in particular premium increases, are a political "hot topic". When the Ontario government reviewed automobile insurance and brought in a new system in 1990, automobile insurance was made a secondary payer to LTD, a system which attracts much less public attention. But there is no real logic to or cost-saving created through reversing the priority of two programs in this way. Why should LTD group insurance plans be first payer for a motor vehicle accident?

Another example (discussed in the HRDC CPP evaluation) is the Ontario government's implementation of an initiative to require all social assistance (Family Benefits) recipients to apply for CPP. While some recipients might benefit from CPP eligibility, others were "losers" because CPP is taxable while SA is not, a fact not taken into account by the provincial government for a person in both SA and CPP. Thousands of recipients received essentially the same total amount from government as a result of this process, and the only real result was a

"shifting" of costs from the provincial to the federal level. The interrelationships among the different disability income systems are dealt with in the next section.

Nevertheless, some major themes and lessons emerge from the various reviews and evaluations. As pointed out in Chapter 2, the definition of "disability" for eligibility purposes has become a major contentious issue in virtually every system. From the perspective of those funding the systems, primarily governments and insurers, it is essential to ensure that eligibility standards are strictly enforced so that costs are controlled and benefits delivered to those who truly qualify. This direction may be reflected in the use of qualifiers like "severe" in defining disabilities, or in requiring that qualifying disabilities be "prolonged".

From the perspective of persons with disabilities, on the other hand, the primary concern is that those who are in need, looking realistically at their disability in the context of their life situation, receive or continue to receive income support. Disability advocates and organizations are opposed to definitions which use standards such as "severe" and "prolonged" to limit eligibility, especially when this is combined with a policy of disregarding or de-emphasizing the social and economic factors that may interact with and add to the consequences of disablement, such as age, lack of education, and language barriers.

Closely related to the "disability definition" issue is the fundamental dilemma usually known as the "welfare trap" or "work disincentive" problem. In the disability income systems reviewed in the previous section (except for Income Tax credits), an important component of the existing definition of disability is that the individual be unable to work, at least on a long-term, if not a permanent, basis. This criterion of eligibility makes sense, of course, in that the very purpose of disability income systems is to provide an alternative source of income to those who are "unemployable" because of disability.

But, as a consequence, people with disabilities who may have significant potential for employment face the prospect of losing their disability income eligibility *entirely* by undertaking rehabilitation, training, education or returning to work. If they are found to be no longer "unemployable", they may lose their "categorical eligibility" for assistance from the disability income systems, perhaps forever. Reviews and evaluations inevitably identify this "welfare trap" as an issue of central importance, but are equally inevitably *unable* to identify solutions. So the critical "lessons learned" are (a) this is the most critical problem to address and (b) it is extraordinarily difficult to address this problem efficiently and equitably — to "balance opportunity and security", as the title of this chapter states.

The potential for people with disabilities to be employed is discussed in Chapter 3. When a person with a disability is employed, and becomes a contributing, taxpaying member of society rather than dependent on a disability income system, this is a "win-win" result. Everyone benefits. The possibility of cost-savings to the system funders are enormous. And, of course, the benefits to the individual, both financial and personal, can be tremendous. But the "welfare trap" is an enormous obstacle to this happening. Report after report in Canada, at both the federal and provincial levels, has shown a poor record of return to work by disability income system recipients. It is important to look more closely at why this is so.

A starting point is that "unemployability", and its inverse, "employability", are "all-or-nothing" concepts. A system based on "unemployability" as a key definitional concept is, in effect, assuming that persons with disabilities can be neatly divided into those who can work and those who cannot. In reality, however, a very significant percentage of persons with disabilities falls into a "gray area" between the two. Some can work, but only part-time. Some could work if barrier removal and individual accommodation were in place. Some could work if only they had transportation. There is a whole range of individual circumstances. But if any prospect of long-term security is to be maintained for the individual and his or her family, our "system" usually pressures the person (and his or her health advisors and advocates) to maintain status as an "unemployable" person. Otherwise he or she risks, not only the loss of long-term income security, but also other consequential benefits, such as drug coverage and rent-geared-to-income housing.

What appears to be needed is a more flexible approach to eligibility for disability income protection, which will take into account significant disadvantage in employment, even where that disadvantage has been successfully overcome for an extended period. This more flexible approach should also allow for continued eligibility for those able to work only part time or on a periodic basis. Continued eligibility for drug and other essential health benefits should, in particular, be ensured for those people with disabilities who enter the work force, especially those at lower income levels. Disability income systems in Canada have begun to move in this direction, but too tentatively to achieve significant results as yet. Of course, barrier removal and individual accommodation must be effectively in place for this to succeed.

Another "lesson learned" is that rehabilitation, training, education and return-to-work must be addressed *early* for those who are newly disabled. Reviews and evaluations of all disability income systems tend to focus on the significant problems created by long-term cases, which are expensive for the funder (government or insurer), and which almost invariably reflect a very difficult living situation for the individual. This point, also elaborated on elsewhere in this study, is closely linked to the previously-discussed issue of "unemployability". Too often, the newly-disabled person has limited resources, and little practical option except to apply for disability income as "unemployable". Then, two or three years later, rehabilitation or re-employment may be addressed, but in many cases, too late to be effective (although the CPP Vocational Rehabilitation Project and many other studies have shown that even those out of the workforce for several years *can* benefit from effective rehabilitation). The person has already long identified himself or herself as unable to work, and many others have made the same identification. Together with a more flexible approach to defining eligibility, disability income systems need to emphasize a much more timely approach to rehabilitation, including training, education and return to work.

Unlike other discussions of the "welfare trap" or "work disincentives", this study has *not* emphasized the details of how much employment income reduces disability income (the "taxback"). There is little evidence to suggest that individuals with disabilities do a detailed month-by-month calculation to compare working with not working. As already indicated, their primary concern relates to the potential long-term or permanent loss of categorical eligibility. In any event, most experts have recognized that, in a disability income system, combining a fair

guaranteed level of security for all with a low "tax-back" for those who are working is unaffordable - unless rules are added which make persons who are working categorically ineligible over a certain income level. (Richard Shillington, Centre for International Statistics, 1994, costed a Guaranteed Annual Income with reference to the impact on persons with disabilities.) This may be an argument for a long-term policy direction towards separating out programs for those with disabilities who are working from programs designed for those who are not.

In summary, while there are many reports and evaluations concluding that opportunity and security should be balanced fairly in disability income programs, much work remains to be done on how exactly to achieve this.

## C. The (Realistic) Prospects for Comprehensive Disability Income System Reform

A number of reviews and studies of various aspects of the disability income system in Canada have proposed that a comprehensive disability income system replace the current configuration of systems and programs (Weiler, 1983; Social Assistance Review Committee, 1988). But these proposals have typically been based on a few broad conceptual points rather than any detailed consideration of what would be involved in such a reform.

Advocates of "one big system" typically argue that it would be administratively more efficient, as many programs would be combined into one. They also argue that it would be fairer and more equitable, as the income and other supports received would no longer depend on factors such as the cause of an individual's disability.

There is some basis for both of these major points, but they are by no means as self-evident as their proponents might believe. With respect to administrative savings, it is common wisdom that larger bureaucracies are not necessarily more efficient. Nor do they necessarily provide better service. The equity issue is more complex. Different kinds of equities are involved.

Of the existing plans, some are for workers and contributors - workers' compensation, CPP/QPP, long-term disability. Other systems, such as no-fault motor vehicle accident benefits and court awards for personal injury, compensate earners on a higher level than non-earners. It is difficult to envisage a reform with no place for earners to insure their earnings, at least partially. Indeed, leading advocates of "universal" disability insurance schemes, such as Terence Ison (Ison, 1994), actually advocate a "universal" plan for *earners* and *contributors*, with a social assistance type of system for other persons with disabilities.

There are also equities related to justice claims. These are raised most forcefully by defenders of the personal injury tort system, who argue (as noted above) that those whose disabilities are caused by others have a right to full compensation by the party at fault. Justice claims are also at the basis of criminal injuries compensation. A somewhat different type of entitlement based on social justice underlies the veterans' benefits system. Injured workers and their advocates

advance a justice claim based on the historic "tradeoff" whereby the WC payment system was created at the cost of giving up the right to sue one's employer.

These brief indications are enough to show that there is by no means unanimous support for a "universal" disability compensation system, nor even a consensus about what such a system might involve or how it might work.

There are huge administrative and practical barriers to "putting together the systems" as well. The seven disability income systems we have discussed - and more would have to be included in a truly comprehensive reform - differ greatly. Some are in federal jurisdiction, some in provincial, and others, such as CPP/QPP, are effectively "mixed". Some are funded out of general revenues, some are funded through payroll deductions, and others are private. They all differ greatly as to benefit structure, administration, eligibility definitions, rehabilitation, and so on. It would be a monumental task to combine all of these in "one big system", even if a strong consensus that it should be done could be achieved.

For this reason, those who have sought to look at comprehensive disability income system reform in Canada have quickly moved from "one big system" to a consideration of intermediate steps that might be taken towards a more coordinated system.

The Joint Federal-Provincial Task Force established by federal and provincial Social Service Ministers in 1982 is the most significant attempt to realistically address the difficult issues involved in this area in Canada. The Task Force concluded (Health and Welfare Canada, 1983) that it would not be a realistic objective to combine all existing programs. Rather it put forward an analysis of six disability income program models, three for earners and three for non-earners. These models would build on existing systems but, in some cases, were quite different from them.

The models put forward by the Task Force for earners were:

- · Mandatory Long-Term Disability Insurance;
- · High Benefit Level CPP/QPP-Type Disability Protection Program;
- · Expanded Workers' Compensation;

### and for non-earners:

- · Improved social assistance by a "special benefit" or "disability amount";
- · Added flat-rate benefits for spouses and dependants;
- · Old Age Security-based model.

No definite conclusions were reached as to the best approach, but these models serve to illustrate some directions that might be taken.

Other ideas regarding comprehensive reform in Canada are found in Beatty, 1991, 1992; Bolduc, 1992; Ison, 1994; Muszynski, 1992; Neil Squire, 1994. The Roeher Institute (1994a)

has developed a model for a comprehensive plan to meet the *costs* of disability, separately from income replacement.

## **D. Five International Comparative Studies**

## i. "International Comparison of Public Disability Insurance"

A Working Paper for the CPP Disability Program Evaluation, Evaluation and Data Development, Strategic Policy, Human Resources Development Canada. March 1996. (SPR Associates, 1996)

This study compares the CPP disability program with QPP and with public disability insurance programs in seven other countries: Australia, Germany, Great Britain, the Netherlands, New Zealand, Sweden and the United States. Extensive comparative information was developed by the evaluation team.

Rapidly increasing caseloads are described as a "worldwide phenomenon". Although CPP had a period of exceptionally high increase, especially during 1990-95, the CPP still wound up with a *lower* percentage of people between ages 20 and 64 receiving disability benefits than any of the other countries. But is this a meaningful comparison? Probably not, as the entire *range* of disability income programs in a country would have to be looked at to decide whose program was "bigger".

It is an open question whether a period of significant increase in the caseload shows a problem with the program or not. As the population ages, as more disabilities become recognized and as jobs become scarce, one would expect a significant caseload increase. It may not be inappropriate at all.

Contribution rates are also very low in Canada, although again it is difficult to make true comparisons because the basic models from country to country are so different. Some (like Canada) have separate programs for contributors (workers) and others — basically a pension for workers and means-tested social assistance for others. Other countries have more of a "unified" model — essentially a universal program in Sweden and a generous means-tested one in Australia and New Zealand. Then there are very different approaches to the interaction between income programs, disability-related costs programs and disability-related services. It is difficult even to compare *descriptions* of the programs, let alone effectiveness.

Definitions of disability for eligibility purposes may be classed according to their dependence on *medical*, *functional*, or *vocational* criteria or factors. These may be used alone or in combination. The Quebec Pension Plan uses strictly medical criteria, which leads to a narrow definition. Great Britain has recently adopted a system based solely on functional criteria. The Netherlands has recently eliminated vocational criteria as well. This represents a significant narrowing of their previously very wide definition which brought in socio-economic as well as vocational factors.

Canada, like the United States, uses a combination of medical and vocational criteria. While there are a number of approaches, it is difficult to compare and determine which is the most appropriate. There is little doubt that a vocational test, even a strict one like "permanently unemployable" - basically the CPP test - is most capable of interpretation to include people whose disability is "enhanced" by the interaction with factors such as age, lack of education, and lack of experience at anything except demanding physical work. Jurisdictions seeking to narrow their criteria are, by and large, eliminating vocational factors. An important question is whether it is "right" or fair to do so.

Some of the countries in the study include partial disability in their national pension or income plan (Sweden, the Netherlands) or in a separate program (Australia, Germany), while Canada (like Quebec, Great Britain, New Zealand and the United States) does not. Partial disability is basically covered in Canada only in workers' compensation and motor vehicle no-fault schemes. But the implications of covering partial disability are unclear — does it mean benefits for those who otherwise would get nothing, or simply reduced benefits for those who otherwise would get full payments? Everything depends on how the program is interpreted — there is no consistent basis on which to make a comparison.

Benefit structures vary widely — flat-rate, earnings-related, needs-tested, dependent on family unit, and so forth — so again, comparisons are difficult. It appears clear, however, that CPP benefits are significantly lower than those in Sweden and the Netherlands, and the CPP maximums are well below Germany and the United States (but these countries have no minimums). CPP payments are more comparable to the regimes in Australia and New Zealand. The CPP maximum pension is less than \$10,000/year and is only 31 per cent of Canada's average industrial wage - hardly the "gravy train" sometimes portrayed in the media. But there is no clear way of defining what level of benefit is appropriate.

CPP is not linked administratively, and poorly linked in practice, to short-term sickness benefits under Employment Insurance. This is largely because the EI sickness benefit program is so short - 15 weeks compared to what appears to be the international norm of one year. This study identified the transition from EI sickness benefits to CPP as one where improvements should be made so that persons with disabilities don't have an unfair interruption in financial support. (There are other reasons for looking at "the first year" as a key area for improvement. This point will be returned to in discussing the United States General Accounting Office report "Return-to-Work Strategies From Other Systems May Improve Federal Programs".)

In comparison with other jurisdictions, including Quebec, the study criticizes CPP for its lack of disability assessment guidelines, for having too centralized a process, and for failing to make effective use of the "team" (that is, "multidisciplinary") concept in assessment. As well, the onus is placed on the applicant to obtain and submit the necessary information, which may be an unfair burden for some, such as those with limited literacy skills. These appear to be legitimate criticisms. Clearly, a disability income program is failing to meet a key objective if there is no guarantee in practice that "like cases will be treated alike".

All of the jurisdictions under study have some sort of appeal system, but some (Australia, Germany, the Netherlands, and the U.S.) have an administrative, internal review process, while others (Sweden, Great Britain, New Zealand) have an immediate appeal to an independent tribunal. Canada's is something of a hybrid system. But the differences between the systems may be more apparent than real, as those countries that do not have an administrative review have or are developing more stringent standards for initial case development, which amounts to the same thing in practice. The evaluators favour devoting resources to the initial decision-making level, which is probably fairer to more applicants. But there is no real way of comparing systems against an "absolute" standard of effective and fair decision-making.

Finally, some countries - especially Germany, Sweden and the Netherlands - clearly place much more emphasis on rehabilitation and work incentives than does the CPP. The evaluation indicated, however, that there are fundamental features of their programs which make it easier to incorporate rehabilitation and work incentive initiatives, particularly the higher levels of disability benefits and the recognition of partial disabilities. There are, however, opportunities to improve CPP's record in this area, which have just begun with the National Vocational Rehabilitation Project - which was, however, very small in scope.

The overall impression left by the evaluation is that while CPP could benefit from ideas developed in other jurisdictions, it is not fundamentally flawed in comparison with the systems of other countries, nor is it fundamentally different from them in the challenges it faces.

## ii. "Return-to-Work Strategies From Other Systems May Improve Federal Programs"

United States General Accounting Office, Report to the Chairman, Special Committee on Aging, U.S. Senate, SSA DISABILITY. July 1996. (GAO, 1996i)

The GAO has prepared several recent reports on the United States Social Security Administration (SSA) and its Disability Insurance (DI) and Supplemental Security Income (SSI) programs. DI is the pension program for contributors (workers) while SSI is the needs-tested program for non-workers. A major focus of these critical reports is the general failure of rehabilitation and return-to-work. Less than half of 1 per cent of DI recipients ever return to work even though they all have a work history - an observation that can essentially also be made about Canada's CPP disability pension program, although the CPP record is somewhat better.

The GAO correctly emphasizes that rehabilitation and work incentive efforts can show big "dividends" even if there is only a little bit of success. For example:

"SSA estimates that lifetime cash benefit payments are reduced by about \$60,000 when a DI beneficiary leaves the rolls by returning to work and by about \$30,000 when an SSI disability beneficiary leaves the rolls by returning to work."

and

82

"For example, if an additional 1 per cent of the 6.3 million beneficiaries were to leave SSA's disability rolls by returning to work, lifetime cash benefits would be reduced by an estimated \$2.9 billion."

It is *essential* to keep emphasizing that apparently very small, marginal gains in this area can have a very significant long-term effect on the costs of a disability income program. It is *not* necessary to have immediate spectacular successes.

The GAO evaluators suggest that the SSA should adopt the "best practices" of the social insurance programs in Germany and Sweden and of private employers in the United States. These systems are seen as effective in returning persons with disabilities to the workplace. The three leading strategies of these systems are:

- Intervene as soon as possible after an actual or potentially disabling event to promote and facilitate return to work.
- Identify and provide necessary return-to-work assistance and manage cases to achieve return-to-work goals.
- Structure cash and medical benefits to encourage people with disabilities to return to work.

These strategies ideally should be integrated into a comprehensive return-to-work program.

In the United States (and in Canada), however, there is a period of delay following the onset of a disability or major illness.

Instead of an early intervention to facilitate return-to-work, the U.S. and Canadian systems are focussed on long-term eligibility to receive cash benefits. As was seen in connection with the comparative study done for CPP, this leaves the disabled individual in the position of having to document an inability to work to avoid the risk of losing everything. The result is that the critical period for rehabilitation and return-to-work initiatives is wasted. Even though efforts may be made "down the road" to reintegrate the individual into the workplace, they are much less likely to be successful.

The U.S. and Canadian systems are similarly much too slow to provide effective rehabilitation services, including case management. By the time rehabilitation issues are considered, the individual may have lost confidence that there is a job to go to, and in his or her own ability to do that job. Individual planning for return-to-work needs to begin much sooner.

Finally, the U.S. and Canadian systems incorporate a number of disincentives with respect to return to work. In the United States especially, short-term cash benefits are high, and medicare eligibility is linked to remaining on benefits. There is little incentive for the disabled individual to return to the workforce in the short-term especially.

This evaluation, like the previous one, suggests much more attention to the initial period of disability when the crucial decisions are made. Too often, people who still should have a reasonable prospect of returning to work are forced after four months to document a total "inability" to work, so they can qualify for CPP (or LTD). Little if any support is given to the individual to remain in the workplace. More people would get back into the workforce if they were not forced to make the "in-or-out" decision so soon. Our system would serve people better in the long run if there were less emphasis on high benefits in the short term and more emphasis on an *effective* rehabilitation strategy, which allowed people to attempt return to work on an essentially "no risk" basis. And, if more people were returned to the competitive work force in this way, resources would be available to provide more adequately for those who have challenging long-term needs, and who are served most poorly by the current system.

## iii. Annual Report 1994-95

Department of Social Security, Australia. (Department of Social Security, (1995)

What is most noteworthy about this Annual Report is not the specific content (although some of the most important developments in Australia are interesting in themselves), but the manner in which it is organized. The Annual Report is related to objectives that have been set out for the Department by Parliament. Each program, and the developments with respect to it during the year in question, is related to specific objectives set out for the program. As the introductory comments state: "The report is designed to allow an evaluation to be made of the Department's performance during 1994-95, particularly in service delivery and accountability, without reference to other sources." While this may be over-stated, the fact that the attempt is made makes the Annual Report extremely useful and informative.

A chapter in the Report, "Income Security for People with Disabilities and the Sick", covers seven programs: Disability Support Pension; Wife Pension; Carer Pension; Sickness Allowance; Mobility Allowance; Child Disability Allowance; and Postal Concessions for the Blind." Each program has its own identified Strategies and Performance Outcome Objectives.

For example, the Disability Support Pension has as a key strategy "linking of the income support aspects of this Program with employment, education, training and rehabilitation activities to facilitate customers' participation in mainstream social and economic life". According to the Report: "While numbers participating in Disability Reform Package assistance are still modest, some 6,500 returned to work from disability payments in 1994-95, an increase of 70 per cent on the previous year." As noted in our discussion of the GAO study, even this level of success can be significant.

But the objectives of the Disability Support Pension are not limited to "getting people off the rolls". They include: Resource Utilization, Evaluation, Adequacy, Self-Help and Financial Independence; Take-up; Administrative Efficiency and Customer Service; Protection of Customers' Rights; and Control of Incorrect Payment and Fraud. Taken together, this range of

objectives provides a more complete overview of a disability income program than would a narrow focus on caseload size and growth or on costs.

In the Australian system, caregivers are eligible for a separate "carer's pension". While the details are not all available in the Annual Report, it appears that this program is being upgraded as part of a community living strategy, and that the improvements are being "paid for" in part by phasing out the "wife pension", a categorical eligibility program for female spouses. In view of the important role played by "informal" caregivers and the high costs of institutional and residential alternatives to home care, a "carer" pension is something that might usefully be considered in Canada.

Mobility Allowance is another interesting Australian concept — it is lump-sum direct funding to those who need transportation support. The allowance appears to be at a standard level for those who are eligible, expressed in the report as \$52.40 (Australian)/fortnight and indexed to inflation. Interestingly, in the whole country during 1994-95, there were only 54 requests for review presented to Authorised Review Officers, 12 appeals to the Social Security Appeals Tribunal, and 3 appeals to the Administrative Appeals Tribunal. This may reflect a high quality and credibility of initial decision-making or a perception that there is little chance of succeeding on an appeal (or both). What we do not know is whether there are alternative informal appeal routes, for example, through an MP's office.

The Child Disability Allowance, for parents looking after children with disabilities at home, appears to be more significant in scale than comparable programs in Canada. The total payments increased in 1994-95 by 10 per cent to over 87,000 children and a total expenditure of \$188 million (Australian). It appears from these figures that the program is not limited to children with *severe* disabilities (as is usual in Canada) and that the *average* entitlement is more than \$2,000/year (Australian). Health Care Cards are provided as well. The program appears to recognize to some extent the lost income due to a parent's inability (almost invariably the mother's) to work outside the home. The government in its Annual Report indicates that it is moving towards the use of "child disability tables" measuring functional limitations against an age-appropriate standard. It is not stated whether cost control is a motivation for this change but it seems likely in the circumstances that it is.

### iv. "Social Insurance, Disability and Personal Injury: A Retrospective View"

John F. Keeler (Law School, University of Adelaide). University of Toronto Law Journal, 44, 275. 1994 (Keeler, 1994)

Professor Keeler reviews developments in Australia, New Zealand, Canada, the United States, and Great Britain over the past 25 years. He notes that, while there has been great interest shown in moving in the "no-fault" direction over that period, particularly due to the influence of Professor Terence Ison's seminal work *The Forensic Lottery*, only New Zealand attempted a truly comprehensive reform. Even then, New Zealand's scheme dealt only with accidents, not with all causes of disability, and its plan has been significantly cut back in recent years.

The value of Professor Keeler's paper is that it reviews in detail the legal debate about "no-fault" over the past 25 years, providing an assessment of the main arguments. But he also identifies a separate type of analysis of disability compensation programs, which looks at their social insurance and social policy aspects, and which is generally ignored in legal approaches to the problem. However, he cites Patrick Atiyah (Atiyah, 1993) in England and Stephen Sugarman (Sugarman, 1989) in the United States as significant exceptions - legal experts who know and discuss the social policy implications of disability compensation proposals as well.

Keeler discusses at some length the range of design and policy features that would have to be considered in a "universal" or "comprehensive" plan:

- · Coverage accidents only, or also injury and sickness?
- · Should there be death benefits, too?
- Should disabilities ever be excluded because of conduct: criminal? self-inflicted harm? recklessness?
- · Should there be a flat-rate minimum?
- · Should there be provision for partial disability?
- · Should there be an "earnings-loss" basis for compensation?
- · Are there "minor" disabilities which should be excluded?
- · What about medical, rehabilitation and long-term care benefits?
- · Should there be provision for non-economic loss (pain and suffering, nervous shock)?
- · How should the plan be funded?

and many other issues. This paper is a valuable overview of what is involved in carrying out an actual thoroughgoing reform of disability income systems.

# v. *Disability, Work and Cash Benefits* (Edited by Mashaw, Reno, Burkhauser, and Berkowitz, 1996)<sup>2</sup>

This important recent publication from the United States contains a series of studies relating to the increase in recent years in disability program claims and to the prospects for return to work for persons with disabilities.

In their overview, Jerry Mashaw and Virginia Reno draw the following main conclusions (at pp. 27-29):

• The recent rapid growth in disability benefit claims is to a large extent a temporary phenomenon, created by the economic downturn of the early 1990s, the aging of the

(working-age) population, and the expansion of the working population due to women entering the workforce in increasing numbers.

- Private sector providers should be involved in rehabilitation in innovative ways.
- There should be a wage subsidy program for persons with disabilities (to supplement the American barrier-removal initiatives under the Americans with Disabilities Act).
- The effectiveness of both return to work initiatives and fairness of claims medical reviews could be improved by a better disability management approach.

## E. Summary: Major Findings and Lessons Learned

- There is potential for a significant proportion of CPP beneficiaries to return to work with appropriate vocational rehabilitation services. Early intervention has been identified in other systems as a critical factor associated with successful return to work but may require coordination with provincial governments, insurers, employers, and so forth.
- Even very small, limited successes in enabling a small proportion of claimants to return to work can result in very significant long-term cost savings. It is *not* necessary to have immediate spectacular successes.
- The all-or-nothing definition of "employability" used in CPP and other income security programs creates a powerful disincentive to rehabilitation and employment, and results in what has been termed the "welfare trap."
- Many people with disabilities are capable of some degree of work, but still would require income support. The all-or-nothing approach makes this problematic, to the disadvantage of both individuals and lack of potential cost savings to the taxpayer.
- There are no clear answers as to how best balance opportunity and security. A starting point may be to concentrate on recognizing the ongoing disadvantage of those who have disabilities but who nevertheless are working.
- The complexity of the income security system, or systems, as they affect people with disabilities in Canada, which results in inconsistencies and overlapping jurisdictions, has been well documented. A starting point for any reform would be a centralized information data base, so that at least the facts about all of the programs would be known.

- There are potential pros and cons to a comprehensive disability income system. There are high administrative and practical barriers. There is no evidence that a unified system will be more effective than the present situation. It could present new problems, as existing schemes are set up to address specific needs which could be lost in a one-size-fits-all approach.
- Alternatives that have been suggested include consideration of intermediate steps that might be taken towards a more coordinated approach.
- CPP could benefit from ideas developed in other countries. Nevertheless, it is not fundamentally flowed in comparison with other systems, nor is it fundamentally different from them in the challenges it faces.

## 6. Independent Living and Community Support Services

by Patti Bregman

### A. Introduction

As we discussed earlier, there is evidence indicating that with the availability of appropriate services and supports, people with disabilities can be capable of employment. Services and supports can also enable people with disabilities to take part in a wide range of community-based services, on the same basis as others. A variety of services, however, may be needed to make this possible. This chapter discusses some of the relevant evidence, and indicates what is known about which types of approaches appear to be most effective.

As this chapter indicates, personal supports and community-based services are critical to the ability of people with disabilities to live independently and to participate in the community. They are essential to the citizenship and equality of people with disabilities. They are essential to employment. Numerous studies and reviews, including various parliamentary committees and most recently the Federal Task Force on Disability Issues (1996), have spoken of this.

Thus personal supports and community-based services can contribute to federal government objectives. Nevertheless, direct responsibility for these services is primarily under the control of the provinces and territories. It should be noted, however, that significant federal funding goes towards these services, through the Canada Health and Social Transfer, VRDP, the *Canada Health Act*, and through other vehicles. Some services may also be directly funded by the federal government. There are various mechanisms by which the federal government, if it wishes, could support activities in this area.

### i. Types of Services

The two types of services most often associated with support or community services for people with disabilities are:

- attendant care (often referred to in the United States as personal assistance services PAS);
- the more traditional home care services, which include professional services such as nursing and non-professional services such as homemaking assistance.

There are many ways in which these two categories of service can overlap. But they have been based on philosophically different models. Attendant care services are closely related to the independent living movement. The goal of most attendant care programs is to enhance independence and autonomy. The concept of self-direction is central to the delivery of attendant care services.

In contrast, community or home care services have traditionally been seen as part of the more professional or medical approach to the care of people with disabilities. However, there may now be a coming together of the two models. In some jurisdictions, home care programs are now beginning to adopt the philosophy of independent living.

## **Attendant Care Services: A General Description**

Attendant care services provide to people with disabilities assistance with activities of daily living. This can include, but is not limited to, assistance with bathing, dressing, and eating, as well as with work-related activities. Attendant services may be provided in a variety of different settings and by different types of service providers. For example, services may be provided as:

### Outreach attendant care

These services are provided through an agency to persons with a disability in their homes, workplaces or other settings. A set number of hours and services is contracted for. Services are not available 24 hour a day.

• Support Service Living Units, or similar congregate living settings

In this type of arrangement, a cluster of apartments enables attendant care services to be provided to a group of people with disabilities in their homes. Specific hours of service are contracted for, but attendants are present on site 24 hours a day and are available on an emergency basis. Services are provided only in the home setting, not in school or at a workplace.

### Direct Funded Services

A program that provides individuals with the funds necessary to purchase their own attendant care services is growing as an option in many jurisdictions. Alberta, Manitoba, Ontario and Quebec have direct-funding projects. Attendant care services in the United States are also provided to people through the Medicaid System on a direct funding basis. Increasing attention is being paid to the direct funding approach. Later in this chapter, we discuss findings from evaluations of various direct funding models.

## **Home and Community-Based Services**

Home and community-based programs serve a wide variety of needs, ranging from home nursing services to support for individuals with disabilities in the workplace. In some cases, they can include attendant care services. These programs are receiving increasing attention for a variety of reasons (see Nosek, 1991; Spillman and Kemper, 1992):

• to enable people who are being released from institutions to live in the community;

- as a result of demographic pressures, resulting in an increasingly aging population who will need more services in order to be able to remain in the community, with the alternative being costly and undesirable institutionalization;
- a means of providing essential services to people with acute or chronic illnesses who would otherwise have to be hospitalized.

## ii. Other Characteristics of Community Support Services

These services can vary across a number of different dimensions, such as:

- Purpose of the service: does it support employment, recreation, health, education, social activity or a combination?
- Population served: are they seniors, people with mobility disabilities, people with intellectual disabilities?
- Location of delivery: community-based or operated out of an institutional setting;
- Type of provider: services provided by family and other informal caregivers, professionals, paraprofessionals;
- Direct or indirect service: services providing care or services directly to individuals, or services directed at improving the quality of care offered by agencies;
- Funding source: federal government, provincial government, private payment, or insurance.

### **B.** Policy Direction for Services and Implications for Evaluation

As Chapter 1 indicated, there have been important policy evolutions in recent years with respect to how people with disabilities are viewed. Canada, along with most other western governments (Lunt and Thornton, 1993) has moved from a philosophy of paternalism to one which encourages and respects independence and responsibility. The Mainstream 1992 Review (Mainstream, 1993) indicates the evolution from a "warehouse" to "greenhouse" to an "open house" vision. This "open house" vision has been endorsed by all federal, provincial and territorial social service ministers, as well as by the federal Standing Committee on Human Rights and the Status of Disabled Persons.

This shift in vision has particular implications for the manner in which community-based services are provided, and for how they need to be evaluated. This evolutionary shift, with its major implications for community-based services, is discussed briefly below.

### **Stage 1: Deinstitutionalisation (Warehouse Model)**

The institutional approach was based on an assumption that people with disabilities, for the most part, are permanently incapacitated and therefore deemed to be incompetent. This approach focused on the care and protection of people with disabilities, primarily in institutional settings.

Over the last 20 years in particular, community living has replaced the institution as the model of choice for delivering services to support people with disabilities. Early evaluations demonstrated that people with disabilities could live in the community. A wide range of research studies identified the types of services needed and looked at different delivery models for the services. (For a review of evaluations of early programs analysing home care services in the United States, the United Kingdom, Canada, the Netherlands, Sweden, France and Italy see Lesemann and Martin, 1993; also see Weissert, Cready, and Pawelak, 1988; and Lord, Hutchison, and Farlow, 1988).

These and many other studies have documented the benefits of community-based services for a number of reasons, including the ability to live and to become part of the community as well as increased abilities in self care (Boschen, 1995; Lakin, Blake, and Prouty, 1991; and O'Brien, 1996). Evaluation approaches are now examining the factors associated with successful transitions from institutional to community settings (Lemieux-Brassard, 1995). Evaluations in progress are examining deinstitutionalisation in a number of provinces, including Saskatchewan, Prince Edward Island, Newfoundland, Ontario, Manitoba and Alberta. Problems in deinstitutionalisation arise when the necessary community services are not available.

The seminal Pennhurst Longitudinal Study (Conroy and Bradley, 1985) of a court-ordered deinstitutionalisation of the Pennhurst State School and Hospital in Pennsylvania, tracked residents over five years after they moved to community settings, and documented many benefits. This study, in common with others, documented changes in attitudes of families towards community placements. Initially, family members were concerned about the possible impact on their relatives, with 55 per cent strongly opposed. But after community placements, opposition dropped to 4 per cent. Conversely, before placement, only 19 per cent agreed strongly, whereas this increased to 66 per cent afterwards.

## **Stage 2: Acceptance of Community-based Services (Greenhouse Model)**

More recently, the use of community-based services has become the preferred model, leading to the closure of institutions. The "greenhouse" approach recognizes the capacity of individuals. It has a strong focus on rehabilitation, on supporting individuals in caring for themselves and adapting to the existing environment. This change in countries around the world has been well documented (GAO, 1994d; Mainstream, 1993). The GAO study identified the need for long-term care services and explored how various countries are dealing with this and the increasing cost.

## **Stage 3: Independent Living Principles (Open House Model)**

The "open house" vision, as expressed by the Mainstream review, "reflects a society which is accommodating to its citizens with disabilities, which understands the importance of empowerment of people, which recognizes the necessity of adapting the social and physical environments, and which acknowledges the rights and responsibilities of people with disabilities." An important development, consistent with the open house vision, is the growth of the independent living movement (which as Boschen, 1995 indicates, emerged some 20 years ago, but has become more widely accepted over the last few years).

Independent living has been defined by the Canadian Association of Independent Living Centres as "a process whereby disabled citizens achieve their desired individual lifestyles by assuming responsibility for the development and management of personal and community resources." Key principles of an independent living philosophy include autonomy and control by an individual over her or his own services. This differs from the traditional model where these decisions tend to be made by professionals.

The role of the disability movement in articulating and establishing the independent living philosophy as the core of personal support services is well documented in a recent evaluation of the Independent Living Resource Centres in Canada (Hutchison, Dunn, Lord and Pedlar, 1996).

Independent living has been valued for its own inherent recognition of the autonomy and independence of people with disabilities. A review of programs, policies and legislation in many different jurisdictions suggests that independent living principles are now broadly accepted and incorporated into programs providing personal support services. It is being incorporated into the legislation in a number of different countries - the Rehabilitation Act in the United States, the Disability Services Act in Australia, and recent legislation in the United Kingdom for example.

There is, however, some evidence that a fundamental shift from a medical model which encourages dependence to one based on independent living principles may exist more in name than in practice. This is discussed later in this chapter.

### **Implications for the Focus of Evaluation of Community-Based Services**

Thus there is no longer any question about whether or not people with disabilities can live in the community. More recently, there has been acceptance of the principles of independent living for community-based services. The focus of evaluation accordingly has shifted to addressing questions such as: to what extent are programs and services consistent with these principles and how effective are these services at improving the quality of life of individuals? Evaluation is also exploring the appropriateness of various service models, and in particular, which approaches are most effective at addressing the needs of specific populations or in different settings.

Many of the personal support services based on a philosophy of independent living have historically served people with particular types of disabilities or needs, such as people with spinal cord injuries. The early programs required persons with a disability to be able to direct their own services and to live independently in the community thereby limiting the services to people with a minimum level of intellectual capacity and experience.

In recent years, services based on the independent living philosophy have been expanded to serve other population groups, such as people with intellectual disabilities. The focus of evaluation and research has been shifting to examining the best means of delivery to different client groups. Evaluation is also examining the impact of external factors, such as living conditions and locale, and implications for program design. Both the National Institute for Disability Research in the United States and the National Health Service in the United Kingdom have major research funding programs underway looking at a range of different types of programs to evaluate their effectiveness and how they can be improved. These research studies are also examining ways of serving people with varying needs, such as people who are deaf or hard of hearing, people with brain injuries, visible minorities, people with psychiatric disabilities, and people with developmental disabilities.

As discussed in more detail in Chapter 7, there is also increasing recognition that evaluation needs to focus less on the quality of the service being delivered, and more on the impact of service on the overall quality of life of the persons receiving the service. For example, the outcome-oriented evaluation process developed by the Accreditation Council (1993) for community agencies is based on this philosophy.

The variety of service types and models of delivery reflects the diversity of needs in the community and makes the creation of "the ideal" program impossible. This does not mean that evaluation research is not relevant or necessary. New and more responsive services and methods of service delivery have evolved over the years in response to individual needs, government budgets, and changes in the underlying philosophy from a system in which the medical model prevailed to one in which autonomy and independence are seen as central.

### C. Impact and Effectiveness of Alternative Approaches to Independent Living

### **Some Overall Findings**

Following are some overall findings emerging from the literature on the overall impact of independent living, as well as its contribution of related factors.

### Acknowledging and Responding to Differences

Personal support services can involve every aspect of a person's life. For many people with disabilities, the attendant or service provider enables them to get out of bed in the morning, get dressed, eat and work or otherwise participate in their community. The attendant responds to the individual's need for physical assistance to carry out routine activities of daily living.

However, these needs are not the only barriers preventing people from fully participating in society. While support services play a significant role in enhancing the autonomy and independence of people with disabilities, there are also other factors that may impact and affect the success of the support service programs.

In addition, the differing goals and populations served by these programs make it impossible to identify one specific model that will work for all, or even to identify only one goal to use as the basis of evaluation. For example, some programs may be intended to assist people attending university, while others may assist people to live in the local community instead of an institution.

In a study funded by the National Welfare Grants Program, Kathryn Boschen (1995) examined personal support service delivery in Canada. This study identified a number of factors that lead to successful outcomes in terms of enhanced independence, participation in society and quality of life. These factors include: type of disability, psycho-social considerations, infrastructure of the programs including involvement of consumers in their operations, and environmental factors such as transportation and housing. The study also found that responsiveness of the program to individual needs was important.

The study indicated the need for a broad range of services which are sufficiently flexible to meet varying needs of individuals. It also identified the need for integrated education as well as vocational and life skills, financial support and access to transportation as well as housing and assistive devices. Advocacy was also identified as an important component. The study recommended the use of direct funding approaches, as well as peer counselling such as that provided by independent living centres, along with family and community supports strengthened through government and community agencies.

The study found that the more productive people tended to be more educated and have higher levels of functional ability, as well as an accessible home environment and transportation. It left open the question of whether increased productivity precedes the ability to make effective use of the services or whether it follows the other factors.

A United Kingdom study (Department of Health, UK, 1996) which explored factors contributing to the social isolation of people with arthritis also identified the importance of considering a range of factors. This study indicated that rural living, lack of access to transportation and inappropriate housing are major factors contributing to isolation. Poverty, lack of education and employment are also factors. In addition, this study identified, as did Boschen (1995), other factors contributing to isolation, including lack of information about available services, restrictions on the types of services available, inappropriate housing, lack of transportation, education, equipment and employment. The study also found that assistance in the support of self-advocacy was effective and enabled people to identify and to obtain the services to meet their needs.

A Dutch study (Kuyvenhoven, Marijke et al., 1996), looking at an innovative home care project for severely disabled elderly people with difficulties in carrying out many activities of daily

living, found that single people living alone had poorer health and a lower state of wellbeing than married people. The study indicated that there is a need for more architectural adaptations and personal attention by care providers for people living by themselves.

## **Respect For Individual Needs**

A theme emerging from evaluations of many different types of programs is the importance of respect for individual needs and desires. For example, a recent Swedish study (Sjogren, 1996) led to the recommendation of a model for individualizing service delivery. The suggested process looks both at the individual's own objectives and how the person experiences his or her own situation. The goals of the process are also reflective of many studies: full participation and equality and a good quality of life for the individual and his or her family.

## **Value of Independent Living Principles**

As noted above, the focus on independent living principles of autonomy and independence is not just a philosophical preference. Studies of various types of support services consistently identify a strong preference by clients for those services that enhance independence and autonomy (Lord, McGeown, Taylor, and Young, 1992; Marrone, Hoff and Helm; O'Brien, undated; Smith et al., 1993).

This applies not only to the actual support services provided to individuals, but also to the indirect services, such as peer counselling and training provided by agencies and consumer-run organizations such as Independent Living Centres. Consumer autonomy and control, as well as the responsiveness of programs to individual needs, is consistently ranked high in evaluations of support service programs. It is also used as the basis for evaluating these programs, such as Nosek's (1991) review of the literature of personal support services and in the evaluation of Independent Living Resource Centres in Canada (Hutchison et al., 1996).

Boschen's (1995) study also documented the importance of *control* to satisfaction and other measures of successful independent living. She found that those who ranked their lives as more independent had greater functional ability, access to affordable and accessible housing, support of family and friends, and a feeling of personal control over their lives. Locus of control was found to be more closely related to quality of life than other factors, such as the level of independent living or productivity.

### **D.** Direct Funding

Direct funding (also often called "individualized funding", "service brokerage", or "self-managed attendant care") provides funding directly to the consumer to hire his or her own attendant or support staff. This contrasts with the typical model where funding is provided directly to an agency. Direct funding permits individual clients to hire and to direct their own

staff. Depending on the model, the consumer may be completely responsible for all financial aspects of their services including payroll duties. Staff hiring and training can also be part of the program.

The direct funding model puts control directly in the hands of the consumer, rather than the service provider. It enabled individuals to purchase their own services and to arrange them in a way that is most suitable to them. The service provider is dependent on the consumer for continued funding, rather than vice versa as with traditional funding models where agencies have control over which services will be provided and how. Direct funding is consistent with the independent living philosophy. It is also consistent with the general societal trend towards making public and private services more responsive and accountable to the end user.

Various models of direct funding are becoming increasingly common in Canada and in the United States. There is also interest elsewhere, such as in Australia, where the Law Reform Commission (Baume and Kay, 1996), in a recent evaluation of the *Disability Services Act* in that country, strongly recommended the incorporation of provisions for direct funding in the legislation.

There has been a significant amount of evaluation of direct funding approaches, in Canada and in the United States, demonstrating the effectiveness of the model. Other evaluation studies, exploring the model in more depth, are currently in process.

For example the Self Managed Care Pilot Project in Alberta (Alberta Health, 1993) was recently evaluated. The project had three goals:

- To provide home care clients with the opportunity to manage their own support services through funds received.
  - The evaluation indicated that this goal was met, with nearly 70 per cent of the clients feeling comfortable about managing their own funds.
- To enable clients who select the self-managed project to increase control over their lives and enhance their personal dignity and well being.
  - This occurred for all of the recipients. In addition both case managers and staff also believed that the goal was achieved.
- To demonstrate the effectiveness in both quality and cost of self managed care.
  - Because of the short time frame, the evaluation was not able to make conclusions about cost effectiveness. However, it identified significant improvements in quality of life, along with improved health and social status and few risks or problems arising.

Indeed, the evaluation found that the greatest positive impact of the project was on the clients' sense of personal control. The program also had positive effects on families. As discussed

below, this is important, because the increased stability and financial situation provides for more positive interactions among family members.

A recent evaluation of a Quebec direct funding program (Tourigny et al., 1996) similarly found that the model had a number of strengths. These included the flexibility of being able to adapt services to individual needs, freedom to chose, and the ability to pay those providing the service (as opposed to being forced to rely on inconsistent help in some cases from volunteers or family members). Consistent with other studies of direct funding, this project documented enhanced feelings of dignity and self esteem.

The Roeher Institute (1991) evaluated a service brokerage and individualized funding approach in British Columbia for individuals with intellectual disabilities. This evaluation similarly found that the approach led to increased self determination, autonomy, and dignity. It provided greater flexibility to be able to respond to changing needs. A factor of particular importance was the contractual status provided to individuals, which they could use to negotiate agreements with service providers about the services they were to receive and under which conditions. The Roeher Institute (1993) has also issued a publication reviewing approaches to individualized funding in Canada.

An evaluation of Enable in the United States (O'Brien, undated) compared a variety of delivery methods including direct funding and agency provided services. The evaluation found that clients believed they had greater control over their lives, particularly the ability to choose their own attendants. Most indicated that they were able to receive the services they needed, although those with high needs did not think they received sufficient services or the proper services because of restrictions by Medicaid. Clients expressed dissatisfaction with traditional agency services where the person has no choice. They considered the quality of staff hired through direct funding to be higher than those provided by agencies.

There are also some evaluations under way. For example, the Roeher Institute, in conjunction with the Centre for Independent Living Toronto, is completing an evaluation of Ontario's direct funding and self-managed attendant services project. The study includes a comparative financial analysis of different systems of funding attendant services, including direct funding, outreach programs, and supported living arrangement.

Despite the overall success of the direct funding models, evaluations discussed above have identified a number of issues that still need to be resolved. Key issues include:

- Management of the administrative tasks: In some programs, the individual is responsible not only for hiring the employees, but for filing appropriate taxes and doing all bookkeeping. The Alberta study found that 54 per cent of clients needed some assistance from family, friends or agencies to assist with financial duties.
- Availability of alternatives: Choice can be more illusory than real if there is not a selection of service providers to choose from. This can be a particular problem in rural

areas. The Quebec evaluation indicated that a number of people had difficulty with recruitment.

Ability and skills of clients to make hiring decisions and to provide ongoing direction: It
is not clear whether some types of individuals may be better suited than others to benefit
from direct funding.

With respect to the last point, it should be noted that one of the functions of independent living centres in the Canada and in the United States, as discussed below, is to provide training and assistance to clients to help them in carrying out their new duties.

## E. Impact of Independent Living Centres

Independent living centres (ILCs) aim to enable participation and integration by empowering individual clients to take charge of their lives. Rather than provide technical aids or advocating on behalf of clients, ILCs encourage and enable clients to do this for themselves (Hutchison et al., 1996).

A recently released report, based on a two-year study of Independent Living Resource Centres in Canada (Hutchison et al., 1996) evaluated the impact of independent living resource centres on individuals with disabilities. The evaluation documented the efficacy of the independent living model, as expressed above. The evaluation indicated that ILCs had significantly assisted individual clients with their feelings of self esteem and assisted them in the development of useful skills which in turn led them to take charge of their own lives. This significantly assisted individual clients in improving their lives, sometimes dramatically so, and in participating in regular life in their communities. The evaluation also found evidence of growing acceptance of independent living principles and practices in the wider community.

The evaluation indicated that ILCs had more positive impacts on people's lives than rehabilitation and other formal services. It indicated that the individual empowerment model was particularly important in enabling people to become competent in a variety of community living skills.

Thus ILCs can provide the support necessary for an individual to be successful in making use of support service programs. A wide body of research from a variety of jurisdictions (Boschen, 1995; Dept. of Health, 1995; Hulgin, Shoultz, Walker and Drake; 1996) documents the importance of services such as peer counselling and advocacy in making support service programs, particularly those responsive to and accountable to the individual, effective. Learning how to direct services, finding and successfully obtaining services takes skills that people may not have when the need for the service first arises. Yet, without these skills, it is unrealistic to expect a person to be successful in independent living.

An evaluation of United States independent living centres administered under the *Rehabilitation Services Act* confirmed the importance of advocacy and skill development services. This evaluation identified a wide variety of services and advocacy available at the local level through programs that respond to and reflect individual and local needs. For example, programs worked on improving accessible transportation, improving access to buildings, educating the public about people with disabilities to break down barriers, and supporting employment opportunities.

In addition to providing funding for services to support independent living through various state initiatives, including independent living centres, Title V of the *Rehabilitation Act of 1973*, also provides for advocacy assistance for individuals who need services beyond the scope provided by other programs. This can provide for the protection of both legal and human rights of people with disabilities. Individual advocacy services include assisting with the enforcement of the *Americans with Disabilities Act*, the *Fair Housing Act* and other human rights legislation.

The availability of training also seems to improve the ability of people with developmental disabilities to live successfully in the community. A United States study (Lakin, Blake, Prouty et al., 1991) indicated that individuals who received a greater amount of independent living skills were more likely to continue living independently. The amount of training, however, did not seem to have an impact on the probability of improving or maintaining skills, suggesting an ongoing need for skills training.

Other settings, such as those which deliver health care services, have also recognized the importance of having independent supports available for people living in the community. These are necessary so that individuals can both obtain the necessary services and use them appropriately (GAO, 1996c).

### F. The Role of the Family

Evaluation studies are documenting the role and impact of the family on the ability of people with disabilities to participate fully in society. The family can take on a number of different roles.

## i. Family Caregivers

Despite the availability of government or privately-funded support services, families still provide the overwhelming majority of support services to people with disabilities regardless of their age. It is estimated that in Canada and the United States family members provide between 70 and 80 per cent of all personal support services (Metro Toronto District Health Council, 1995).

A number of societal changes are raising questions about whether families can and should continue to provide this level of support. Since most of this informal care is provided by women, the return of women to the workforce has had an impact on the availability of family

101

caregivers, particularly for seniors where the woman may have obligations to care for elderly parents and younger children (see Kemper, 1989; Lesemann and Martin, 1993; and Spillman and Kemper, 1992).

In addition, evaluations are indicating that the added stress of care giving may have a negative impact on the caregiver's own health and ability to cope. The growth in respite services is just one means of dealing with this reality and reducing the risk that caregivers will require health or social services themselves.

For example, one study ("Patterns of Caring for People with Dementia in Canada") compared the health of caregivers caring for people with dementia and the health of caregivers for people who are not demented. It found that 94 per cent of the caregivers for people with dementia are unpaid family members, relatives or friends. They rarely use community support services. Those caring for a person in the community are more likely to experience chronic health problems and depressive symptoms, to feel more burdened than those whose family member is in an institution.

It is interesting to note that the evaluation measures of a number of the projects reviewed (Silverman et al., 1995, for example) include the impact of the program on caregivers, family and other informal support. The types of programs considered range from independent living to geriatric assessment programs that enhance the rehabilitation of aging seniors.

### ii. Family Support of Independence

Caregiving is not the only role that families play. Boschen's (1995) research, as well as other studies, show how the support of families may also play a role in successful independent living. It is therefore important to recognize that families, like service providers, may view the world differently than the consumer. A United Kingdom study (Heyman and Huckle, 1993) compared the attitudes of adults with intellectual disabilities and their carers towards dangers and risks they might face in the community. The researchers looked at a variety of topics including relationships with the opposite sex and independent living. It found that the category of risks and dangers was the most important aspect of independent living discussed. The areas of most concern were physical hazards, hazards from other community members, sexual hazards and hazards from other adults with learning disabilities. They found that family members had a different perspective towards the hazards than those of the adults with disabilities or their caregivers. In particular, professionals and people with manual skills seemed most concerned with hazards as dangers to be avoided.

Independent living centres and similar types of programs providing broad-based advocacy, training and education have been proven useful in providing the individual with support when making the transition from the family home to independent living in situations where the family may be apprehensive about the move. The programs give the individual the self-confidence and support to make the move successfully.

## **G.** Impact of Other Factors

## i. Housing and Living Arrangements

A number of studies have identified the importance of housing to the success of community-based programs. Issues related to housing involve affordability and accessibility. The relationship between environmental factors including access and independent living has been recognized early in the development of independent living (DeJong, 1981).

Both Canada and the United States have a housing crisis for people with disabilities who have low incomes. In the United States it is estimated that many people with disabilities are living in substandard housing, paying 50 to 75 per cent of their income on rent or else living at home with elderly parents, or are homeless. A report by Housing and Urban Development (1994) identified people with disabilities as the group most likely to live in severely inadequate housing.

Inappropriate housing was also identified as a barrier to participation in the community and a factor in increasing dependency rather than independence. One study (Sapey, 1995) looking at the housing needs of people with disabilities who use wheelchairs, indicated the importance, yet again, of involving clients in planning. This study found that when the people were not involved in the process of planning for housing, the final result was inappropriate housing and increasing dependency. Ratzka (1994) indicated that while renovating existing buildings to make them accessible can be costly, building the same structure with barrier-free design from the beginning requires only minor increases in construction costs.

At least one study (Kemper, 1989) has found that living and demand for care arrangements, including formal and informal care, are related and that changes in one will have an impact on the other. In addition, family availability and employment as well as income and price also determine whether paid or unpaid care is used.

## ii. Effectiveness of Assessment Services

Support service programs depend on effective assessment of needs, not only for the services, but for the other components of a person's life such as housing and transportation. Traditionally, assessment programs revolve around professionals and ignore the input of people with disabilities and their families. One of the most attractive components of independent living programs is the degree of self-assessment it entails. Assessments not only involve the need for particular types and quantities of personal services, but also take into account environmental factors.

Evaluation studies have found mixed results in the effectiveness of assessment programs, whether done by the individual or professionals. This suggests that it is an area requiring further study. For example, the Alberta evaluation of the direct funding pilot (Alberta Health, 1993) found some variations in the actual hours and types of services used from those identified in the initial assessment. The effectiveness of self-assessment of barriers and needs seems to be more

problematic for people with disabilities who are elderly. The enhanced expertise of a geriatric assessment program was found to identify a higher level of cognitive impairments, depression and other functional problems than the community health program assessment, leading to an increase in satisfaction for the individual and less stress for the caregivers (Silverman et al., 1995).

One study indicated that expert assessment of barriers in the home setting which might impede independence could be useful (Steinfeld and Shea, 1993). This study found that the self-report did not always provide an accurate picture of the actual living situation. It also found that many of the needs were unique to the individual, requiring flexibility in the assessment.

## iii. Services for Different Populations

As noted earlier, a number of research programs are underway looking at how support services based on the independent living principles can be adapted to serve a variety of specific populations. Because the fundamental principle of independent living is flexibility and responsiveness to the needs of the consumer, adaptation is possible in most cases. But while independent living principles can be widely used, there are a number of differences which must be taken into account when developing specific programs and services for individual populations.

There is an increasing number of research and evaluation programs focusing on services for special populations including people with intellectual disabilities, psychiatric disabilities, frail elderly and minorities. The results that are available suggest that:

- Specialized programs may be useful to deal with specific types of problems such as alcohol addiction, violent behaviour and language or culture issues.
- Programs should take special needs into account with respect to training, consumer education and methods of providing service.

In order to meet the unique needs of individuals, some variation will be required and special programs established to target particular needs (Boschen, 1995). While generic programs might be effective, some studies indicate that specialized programs will be even more effective. For example, a United Kingdom study looked at costs and qualities of four different models for 40 adults with multiple disabilities. The models were: institution-based units, campus-style further education service, community-based group homes and "ordinary" community-based group homes. The study found that on almost all measures of outcome, the specialized group-home model was the preferred model although it was not necessarily the highest cost.

#### **Seniors**

A great deal of interest is being shown in the use of community support services by seniors with disabilities. Seniors have the highest rate of disability in Canada. Research into the use of these

programs is resulting in the recognition that services such as attendant care services can benefit people with a broader range of disabilities including seniors with disabilities.

The rapid growth of interest in this area is evidenced by a recently published literature review which looked at publications between 1992 and 1995 relating to assisted living for the frail elderly, a fairly narrow topic. The review, part of a national study of assisting living in the United States which is currently in progress (Hawes, Manard, Kane and Molica, 1997) found 175 articles published between Jan 1992 and Sept. 1995.

The authors indicated that they were surprised to find little consideration in the published research of the potential of assisted living to decrease the use of institutional placing in higher care settings, such as nursing homes. The review did show that assisted living improved the quality of life to some extent. The authors suggest that further study is needed to find out why the substitution of assisted living for nursing homes is not occurring at the expected rate. There is a current study in the United States looking at assisted living to identify where it fits into the system and how it addresses the needs of seniors. Among the criteria to be evaluated is the extent to which assisted living embodies the principles of consumer choice and autonomy as well as affordability and its potential substitution for institutional care.

Findings from evaluation are also challenging assumptions that programs that promote autonomy are inappropriate or not wanted by seniors. For example, one study (Sheets, Wray and Torres-Gil, 1993) examined the differences between programs offered to people with disabilities under the *Rehabilitation Act of 1973* and programs funded through the *Older Americans Act* for seniors with disabilities in the United States. It found that programs for people with disabilities promote the enhancement of personal choice and support independence. The programs for the elderly disabled emphasize services that diminish personal autonomy and increase dependancy. The study suggested that advocacy efforts should focus on educating the professionals and clients of the value of geriatric rehabilitation.

This is consistent with other studies that show that consumer control for the elderly with disabilities is as important as for younger people. One of the studies done for the National Forum on Health reviewed the factors that resulted in seniors maintaining independence and identified programs it considered successful. Control was considered an important factor in leading to success.

#### iv. Cost Effectiveness and Decreased Institutionalization

The growing population of seniors and people with disabilities, as well as shrinking budgets for social and health care services in many jurisdictions has once again turned the attention of government and researchers to the cost effectiveness of programs. There are several studies comparing long-term care programs in different countries, considering both costs as well as strategies to deal with increasing costs (GAO, 1994c,d). These studies identify three trends in methods used in the countries studied to control costs and improve administration of services:

• Fixed budgets along with cost controls

- Consolidating delivery of long-term care through coordinated administration
- Increased public support for long-term care

While these trends are evident in long-term care services in Canada, as reflected in the increasing regionalization of health and long-term care services, it is too early to assess whether these changes have actually succeeded in reaching the goal while maintaining the level and quality of service. This is particularly true in the United States, which has seen a significant rise in the cost of long-term care services. Among the results expected were a decrease in hospitalization as well as overall reduction in costs.

One of the justifications for the expansion of home based programs was deinstitutionalisation and expected cost savings. There is some work suggesting that these goals have not been achieved. Research is currently underway to determine the extent to which it occurs. Some studies show that it does not necessarily decrease institutionalization, but researchers analysing those studies hypothesize that this may result from the fact that the programs may not be targeting those at highest risk of institutionalization. Results from a small project in Tennessee show that providing these services can be successful in moving people into the community who have been in an institution for a number of years.

Cost effectiveness in the United States is also used as the basis for determining whether a service is provided in an institution or at home. In reviewing these evaluations, a number of factors must be considered before making policy decisions based on the results.

Simply comparing costs of services does not reflect the true cost of a program. While delivering some services in an institutional setting may be less expensive, if the person receiving the service cannot participate in the community, obtain employment and generally have a higher quality of life, the actual financial as well as social cost may be higher. Again, there is a need for an understanding of the full costs of disability in order to be able to carry out effective cost analyses.

It is also important to identify the characteristics of those served by various programs. For example, community services programs may not be serving the same type of people as those being served in institutional settings. It is important to determine if this is in fact the case since it may help target assistance more appropriately where the goal is reduction of deinstitutionalisation.

## H. Incorporation of the Principles of Independent Living into Agencies

Finally, as indicated at the beginning of this chapter, questions have been raised about the extent to which agencies in particular have actually incorporated the independent living principles into their day to day practice. Despite formal support for principles of independent living, some research suggests that organizational change is still a major challenge.

Principles of independent living are increasingly being incorporated into policies, programs and legislation. For example, in the United States Title VII, Ch. 1, Part B. of the *Rehabilitation Services Act* authorizes grants to assist states to promote the philosophy of independent living which includes consumer control, peer support, self-help, self-determination, and equal access to an individual system of advocacy to maximize leadership.

Similarly, in Australia and the United Kingdom, there is new legislation providing for the direct funding of services and expanding the role of community services. In Canada, all provinces have programs for attendant care services and other long-term care services in the community. Ontario, Manitoba and Alberta have established programs to provide direct funding to people with disabilities to enable them to purchase their own services without going through the more traditional agency. However, evaluations are raising questions as to whether these principles are in fact operational in a real sense, or in form only.

Two recent studies suggest that organizational change is still necessary to deliver services based on an independent living model, even in institutions which "nominally" adopt these principles.

A study done by Whitney-Thomas and Thomas for the Institute for Community Inclusion looked at the impact of the 1992 *Rehabilitation Act*, which established standards for rehabilitation practice, on the administrators and counsellors of those agencies. Both administrators and counsellors believed that some change had occurred, with administrators thinking more change had taken place. These changes occurred in the context of consumer empowerment and choice, assistive technology and accommodation, consumer involvement and services for people with HIV and AIDS. Neither of those groups believed any real change had occurred.

Hagner and Marrone (1995) also looked at this issue and found that fundamental agency changes have not yet occurred. It is apparent that while the language and philosophy of independent living is often used as the model for developing services, many agencies have not yet incorporated these principles and philosophy into the actual delivery of services.

The Roeher Institute (1992), in a study of personal supports in Ontario, identified a number of problems with how they are delivered. For example, they tend to be framed on a medical/rehabilitation model, rather than on principles of independent living. Clients have little control over how services are delivered, and few opportunities to making choices and provide direction for services. Personal supports also appear to be geared toward persons with ongoing stable conditions, and cannot respond well to individuals with high care needs.

The evaluation of ILCs in Canada (Hutchison et al., 1996) did not look specifically at consumer responses to community-based services and attendant care. But it did identify characteristics which people using those services find important. For example it indicated that clients valued being independent, being respected and considered part of the community. It also indicated that these factors are key to success in achieving community integration. This suggests that these characteristics would also be fundamental to the delivery of an effective community-based service.

In particular, it suggests characteristics for organizations that can improve their effectiveness in serving people with disabilities. For people with disabilities, the structure of service delivery may be as important as the actual service itself in determining satisfaction.

One of the core values of the independent living movement is empowerment. This is a term that is also used to describe the purpose of attendant care and other community-based services that support people living in the community. Some researchers have suggested, however, that while the term is frequently used, there is little agreement as to how it can be implemented in practice.

## I. Summary: Major Findings and Lessons Learned

- The policy direction in Canada, as well as in many other countries, has changed from a philosophy of paternalism to the "open house" vision which respects and supports independence and responsibility. This is in keeping with the independent living philosophy and is incorporated in legislation in several jurisdictions.
- The evidence overwhelmingly indicates the capacity of individuals to live in community settings, given proper supports. Clients strongly favour community living, in particular the independent living models which give them control over their lives. Family members are also generally supportive of the relocation of their relatives with disabilities from institutional to community settings.
- Individual control and empowerment has been identified as one of the
  most important factors to the success of community services in
  facilitating skills development, community and social integration,
  consumer satisfaction and improvements in quality of life. Services based
  on independent living principles are more effective in this regard than
  traditional, professional-driven services.
- A number of evaluations have documented the effectiveness of direct funding, which gives individuals the funding and the control to select and to direct their own attendant care and other personal services.
- Advocacy, support and training services provided by independent living centres have been shown to be effective in enabling people to take control of their services and their lives.
- Effective community-based programs, in addition to personal support services, also need to take into account a broad range of needs including housing, transportation and support for families.
- Program design needs to consider the special needs of different populations, such as seniors, people with developmental disabilities and people who are deaf or hard of hearing. They should also take into account the setting, such as rural or urban, in which the service will be delivered.
- While principles of independent living are increasingly being incorporated into policies and program philosophies, the manner in which many agencies actually deliver services does not reflect these principles.

# 7. Implications for How and What to Evaluate in the Future

This chapter discusses lessons learned for the approach to evaluation itself, along with implications for future approaches to evaluation in the area of disability policy and programs. These lessons are based on our review of evaluations and related research that has been carried out - a somewhat different focus from the rest of this report.

#### A. Status of Current Information Base

This study has identified an impressive amount of evaluation information which *is* available, more than was expected when the study was initiated. Many of these studies are listed in the extensive bibliography to this report, which by no means represents a complete list. As this report has indicated, considerable information is available about many basic questions and issues — such as the capability of many people with disabilities to work given the appropriate supports and elimination of barriers. We know about the potential of many approaches, such as vocational rehabilitation and supported employment.

As the Federal Task Force on Disability Issues (1996) noted, there has been no lack of information, documented in a series of reports, about the need for action. But, as the Task Force said, "implementation of these visions and the recommendations has left a lot to be desired." This has significant implications for evaluation. Evaluation is needed to assess the extent to which policies and proposals are acted upon and to what effect. In addition to a monitoring function, evaluation can be very useful in aiding in program development and improvement. It can play a very useful role by identifying, as early as possible, where implementation runs into difficulties or takes a different course than anticipated, why, and what is needed, where applicable, to put things back on track.

Nevertheless, there is still a lot we do not know about what works or does not work, and why. Even when we know about the potential of an overall approach, such as supported employment or home care, there is not enough known about the effectiveness of specific programs or what makes a difference. Evaluation is frequently lacking, along with even basic data, such as who is being served and in what ways. Comparative evaluation examining the strengths and weaknesses of various potential approaches has been limited. Such information is critical for meaningful planning and priority setting, not to mention for delivery of services that really *will* be effective and make a difference.

As indicated in this text, lack of adequate consideration of interactions and relationships represent a major policy weakness. There has been insufficient coordination across jurisdictions, departments within the federal government, and even across program areas in the same department. Similarly, there has been limited evaluation exploring the interactions and linkages across different areas. This is a major field where more evaluation is needed.

Ongoing valuation will be needed to explore the effectiveness of emerging program and policy directions, as well as to provide guidance in the best ways of implementing new directions. For example, some exciting research is currently in progress, particularly exploring the effectiveness of initiatives in areas like direct funding, home care, consumer involvement, and deinstitutionalisation, in Canada and in the United States. With an aging population and more recognition of the key role of consumer control in a range of outcomes, these are all "growth" areas. The United States government has set up a number of research funding programs, sponsoring major evaluation and research studies in these and related areas. A number of evaluation activities are also under way exploring the impact of various aspects of ADA in the United states.

## **B.** Evaluation Paradigms and Approaches

## **Participatory Approaches to Evaluation**

There is a strong trend in the evaluation field towards greater emphasis on participatory and empowerment evaluation. This, for example, was a major theme at the 1995 International Evaluation Conference in Vancouver. A major lesson learned is the need for greater involvement of people with disabilities in research and evaluation of programs and policies that affect them. This is consistent with the current partnership approach to disability within the federal government, which has been highlighted by previous reviews (Mainstream, 1993; Federal Task Force, 1996).

There has been particularly strong condemnation within the disability community of the traditional research paradigm that leaves out the perspective of the subject in any consideration of the design, methods or results of research (see especially Oliver, 1990, 1992; Rioux and Bach, 1994: and Woodill, 1992). People with disabilities have expressed their resentment of others deciding what is best for them. They are just as upset by researchers and evaluators who decide research questions and approaches without involving them in the process.

The above writers, and others, have pointed out the major differences in the assumptions and questions used in research directed by professionals from those when consumers have been involved. When professionals determine the research questions, they tend to focus on problems with the individual. But when consumers help set the agenda for research, they identify questions related to the barriers and environmental factors that preclude participation, integration, productivity, socially accepted behaviours, and a host of other outcomes. These writers indicate that research needs to be based upon a socio-political model, one which takes into account the social and political context and factors in "our disabling society" (Rioux and Bach, 1994).

There is ample evidence documenting the practicality of involving consumers in evaluation (Byrnes, 1996; Cantrell and Walker, 1993; the Mainstream 1992 review). One of the themes at the 1996 International Association for the Scientific Study of Intellectual Disability conference in Helsinki was the active involvement of people with intellectual disabilities in research and

112

evaluation. There is evidence that the involvement of people with disabilities in the evaluation process can improve the quality and credibility of the evaluation, lead to more relevant and improved policies and programs, and result in improved outcomes for the participants themselves.

## **Practical Approaches to Evaluation Which Helps Programs Improve**

A major theme in the evaluation field concerns the use or "utilization" of evaluation. There is little point in carrying out evaluation if it will not be used in some way. A focus on utilization requires evaluation to take a practical focus, providing timely, useful advice and guidance about how programs and policies can be improved. There is an extensive literature documenting the characteristics and features of evaluation most likely to result in use (Patton, 1997).

Evaluation most likely to result in action provides information about program effects and impacts, as well as implications for future directions, and is seen as credible and valid by key stakeholder groups. It needs to take into account the interests and concerns of various constituencies, ranging from senior executives and politicians in government to people with disabilities at the community level.

Evaluations that are most useful address questions such as: Are programs and policies indeed being implemented as intended? What are the reasons for these findings, and what are their implications for future directions? What are some of the barriers? How can these be overcome? When evaluation is implemented early, and is built into the program as much as possible, it can be used to provide feedback and guidance for programs.

A range of methods is needed, in particular qualitative, as well as quantitative evaluation data. Qualitative data can be especially valuable in providing an understanding of *why* given strategies are working or not. As Lutfiyya (1991) has noted, qualitative information can be used to "ground" findings in context, and to look at interactions and reasons. Given the importance of interrelationships and linkages that this report has identified, this seems particularly relevant. Qualitative methods also can provide a way of involving individuals in research and evaluation who are unable to answer questionnaires or to perform certain tasks, but who can still share their thoughts in other ways.

## C. Success Indicators for Disability Policies and Programs

The literature indicates the need for a greater consensus on what are appropriate outcome measures for the evaluation of disability policies. Without agreement on at least the general goals of policies, it can be difficult or impossible to focus evaluation efforts or to identify appropriate measures of outcomes. As a result, evaluation findings may not be considered relevant or acceptable.

Two broad areas have emerged that may prove useful for thinking about the outcomes of disability policies: quality of life, and cost effectiveness. These are discussed below.

## **Quality of Life**

The concept of quality of life is part of the public lexicon, with potential for considerable face validity. It can encompass other outcome measures frequently mentioned in the disability area, such as equality and social integration, consumer participation, and citizenship. It can be assessed through a variety of quantitative and qualitative measures. It can refer to living conditions of individuals, as well as of larger communities.

Quality of life can provide a way of thinking about outcomes from the perspective of the consumer, rather than in terms of what services are provided or what "experts" feel is appropriate. Quality of life now represents a major field of study in the disability area, with a strong and growing body of literature. It has been receiving extensive attention internationally. It is increasingly being viewed as an important, perhaps *the* key outcome indicator. For example, the title of a session at the forthcoming (May, 1997) American Association on Mental Retardation is entitled: "Quality of Life: Could It Be an Emerging International Disability Policy Concept for the 21<sup>st</sup> century?" The concept of quality of life is generic in nature, but considerable research in recent years has focused particularly on people with intellectual disabilities. Attention is now broadening to look at other population groups as well.

The concept of quality of life reflects two dimensions: objective indicators of life conditions, and individuals' subjective assessments about their life experiences (Borthwick-Duffy, 1996, Schalock, 1994). The literature indicates that both dimensions are important in assessing the impact of public policy.

A number of different approaches are being used in investigations of quality of life. They tend to share the above dimensions in one way or another. There is broad agreement that quality of life is a multi-dimensional construct. There have been various attempts to develop quantitative "quality of life" instruments. Some of the most extensive work in the area of assessment of quality of life has been by a University of Toronto group (Renwick, Brown and Raphael, 1994; Woodill, 1996).

The Accreditation Council (1993) has developed an approach to the evaluation of agencies serving people with disabilities that is consistent with the quality of life perspective. It has identified 30 outcome measures that can be used in shifting the perspective of social agencies from process to outcome and as a means of assessing impact and effectiveness. It has developed and is using a protocol for the accreditation of social agencies in the United States, which is now starting to be applied in Canada as well.

As Borthwick-Duffy (1996) indicated in a discussion about the evaluation and measurement of quality of life, there is a lack of consensus over basic conceptual issues (such as the legitimacy of subjective vs. objective assessments) as well as operational definitions and models. Despite various attempts to operationally define quality of life, "the fact remains that it is essentially a subjective construct that resists numerical scoring." *The* definitive measurement instrument probably has not yet been developed (although some may disagree). As Borthwick-Duffy indicated, it may be premature. Schalock (1996) indicates that quality of life is most useful as a

construct rather than as a score, and that it may be dangerous to attempt to quantify it at the present time.

Despite the promise of quality of life as a means of identifying outcomes from the perspective of the individual, it has been subject to criticisms from consumers (Byrnes, 1996; Pfeiffer, 1995). These concerns seem to be less with the concept *per se* than how it is operationalized and applied. For example, there is concern that quantitative ratings by "experts" which reduce quality of life to a number do not reflect reality or consumer perspectives about what is important to them. Approaches that have not provided for consumer participation have been criticized. Some measurement approaches are seen as overly intrusiveness. And while the multi-dimensional nature of quality of life and its measurement can be a strength, there is also a feeling that use of standardized instruments may not reflect issues germane to a particular situation. There is also a concern about how quality of life ratings can be used or misused. Pfeiffer (1995), for example, says that ratings of quality of life can be used to devalue people with disabilities whose ratings are too low, and even serve as grounds for euthanasia.

#### **Cost Effectiveness**

There is little disagreement, in principle, on the legitimacy of cost effectiveness as *one* measure of effectiveness. In a time where government programs are under increased scrutiny, value to taxpayers of public expenditures is receiving increased attention. Many people with disabilities, for example at hearings of the Federal Task Force, have also indicated their interest in seeing the limited amount of funding available for disability spent as effectively as possible. This report has cited evidence of cost effectiveness where available regarding the payback of various program approaches (of various employment strategies, for example).

Given that cost effectiveness is recognized as one key variable for assessing the impact of programs, it is striking that there has been so little conceptual attention paid to what this concept means in practice for assessing the costs and benefits of disability policies and programs. Because there has been only limited work (Federal Task Force, 1996; Perrin, 1991; Roeher, 1994) considering the costs of disability, including extraordinary costs to individuals and families, to government and society, there is limited understanding of what disability really costs. It is clear, from work done to date, that there can be a range of both direct and indirect costs. Costs can occur in one program area, with savings and benefits in another. Without consideration of the nature of costs resulting from disability, it is not possible to do meaningful cost-effectiveness analysis.

Policy decisions made in the absence of understanding of what it costs *not* to take action can be flawed. For example, the evaluation of the federal disability management pilot project found very limited understanding among senior managers about what it actually costs the government for employees who are off work due to disabilities.

Also, what is the best way to measure costs and benefits? What time frames are appropriate? The limitations of taking too short a time frame have been discussed. Perhaps more than with other target groups, initial expenditures in the disability area may be required where the

payback may not be until subsequent years. This can result in a reluctance to look at expenditures as investments. Nevertheless, as indicated above, there can be potential for major payback if an appropriate time frame is chosen. To give one example, a United States GAO report (1996i) has been cited indicating the potential for savings in Social Security Disability payments even if just a small number of people can be assisted in returning to work. Conversely, some evidence has been cited in this study to suggest that initial benefits or outcomes may not endure, at least under some conditions, and initial gains may be illusory.

Given the importance the federal government appears to be placing on cost effectiveness as a key evaluation outcome, it would appear that a priority area for additional research and evaluation attention would be in identifying effective and appropriate means of conceptualizing and assessing cost effectiveness. As noted above, a prerequisite to this is additional research to identify the full costs of disability, taking into account both direct and indirect costs and benefits.

#### **Outcome Measures Tailored to Individual Circumstances**

While one can set general parameters for considering outcomes, it is important that they not be applied in an overly rigid fashion. As this report indicates, there are many different types of programs serving people with widely ranging characteristics, needs and desires. One size cannot fit all. Evaluation designs need to be sensitive to differences and adapted to individual situations.

This paper has shown that employment is not an all-or-nothing matter. Many people are capable of some degree of work, for example on a part-time or episodic basis. Outcome indicators for the evaluation of employment programs may need to develop more specific indicators of employment status, to reflect the nature and characteristics of employment and to take into account the target audiences and specific program objectives.

While employment may be a goal for many people, evidence indicates that it is not appropriate for everyone, and it can be dangerous if it is used as an outcome measure where it may not be appropriate. As has been noted, there is strong potential for many CPP beneficiaries and recipients of other forms of disability income to be employed. But for others, this would not be a realistic outcome. Quality of life, including considerations such as independent living, community inclusion and integration, are also important goals. For many people, they are just as important as financial or economic goals.

## **D.** Areas for Future Evaluation

Following are areas for which additional evaluation and research programs may be worthy of consideration. This is not meant as a complete list, but intended as a starting point for development of an evaluation and research strategy for disability. It should be noted that the appointment of the Minister of Human Resources Development as lead Minister for disability

issues provides an opportunity to develop a new research and evaluation strategy for disability that cuts across all government programs and departments.

## **Conceptual and Methodological Explorations**

- Developmental work in defining and developing a consensus on appropriate and realistic outcome measures for the impact of policies and programs on people with disabilities.
- In particular, exploration of how the concept of quality of life can be used as a basis for evaluating the effectiveness of programs and services, building on the current Canadian and international work on quality of life.
- Exploration of the full costs of disability, including both direct and indirect costs and benefits, extraordinary costs borne by individuals and families, costs to government, to society and to others.
- Identification of considerations required for more meaningful cost effectiveness analysis in the disability area, taking into account considerations related to the identification and measurement of both costs and benefits.
- Longitudinal research in a variety of areas, in order to be able to identify long-term impacts, benefits and potential cost savings.
- Coordinated, multi-dimensional and multi-site research, such as is now under way in the United States and the United Kingdom.
- Use of evaluation models which provide for consumer involvement.
- Exploration of means of encouraging more attention to outcomes and evaluation by programs at all levels. This may involve the development of evaluation tools and resources to support more local level evaluation activity.

#### **Policy and Program Directions**

- Evaluations and research studies examining linkages and interrelationships across programs and across federal departments, as well as exploration of relationships with programs provided by other jurisdictions, considering areas where synergy is important as well as identifying gaps and overlaps.
- Evaluations identifying the extent to which the government response to the Federal Task Force has been implemented and the impact of these initiatives.
- Evaluation of the impact of the new EI Act on people with disabilities, taking into account the impact of the overall policy approach, as well as of services provided by

HRCCs directly, by third party agencies, and services provided through agreements with the provinces.

- Evaluation of the impact of the new Employment Equity legislation.
- Exploration of how other HRDC programs and policies are affecting people with disabilities.
- Research identifying the impact of the "new economy" on people with disabilities, as well as future challenges and opportunities.
- Greater use of demonstration or action research approaches which have non-token evaluation built into the design and budget, in order to test out new models of programming. CPP has been using this approach in exploring new ways of partnering with other programs with mutual beneficiaries.

#### Other Areas for Additional Evaluation and Research

- Tracking the size and characteristics of the disability population in Canada for example, through reinstatement of the HALS survey or through alternative approaches, such as piggybacking on other survey vehicles or small-scale dedicated surveys with results reported on a more timely basis.
- Identifying characteristics of people served by various types of disability programs across Canada for example, characteristics of people served by supported employment and independent living programs, as well as identification of which program approaches (alternative approaches to rehabilitation, the effectiveness of various approaches to barrier removal) work best for whom and under which circumstances.
- Evaluation identifying the extent to which the independent living model and a consumer perspective is being incorporated into the philosophies and operations of governmental and non-governmental programs and services.

## Dissemination of Information about Findings and Implications from Evaluation Studies

A process is needed for information sharing about what is learned from evaluation and research findings, as well as identifying implications of this information for policy and practices. This study represents a good first step. Information sharing is most likely to be effective when it is planned as part of the evaluation design, and when the manner and style of reporting is tailored to the needs of specific stakeholder groups. There appears to be a need for information sharing among HRDC and other federal government staff, as well as with stakeholder groups.

Methods of information sharing used in other jurisdictions may be worthy of consideration. For example, the European Commission (1996) recently published a *European Guide of Good* 

*Practice Towards Equal Opportunities for Disabled People*. What is particularly noteworthy about this guide is that it organizes best practices examples around principles in a number of different areas (work and preparation for employment, educational integration, social integration and independent living, functional rehabilitation). A participatory approach was used in the development of the principles.

The United States National Institute on Disability and Rehabilitation Research (NIDRR, 1995) has gone one step further. It has not only put a high priority on information dissemination ("to ensure that rehabilitation knowledge generated from projects and centres funded by NIDRR and others is utilized fully to improve the lives of individuals with disabilities and their families"). It is also establishing an infrastructure to share knowledge in a very practical way, through the creation of ten regional Disability and Business Technical Assistance Centers.

## E. Summary: Major Findings and Lessons Learned

- More attention to the development of appropriate outcome measures for the evaluation of disability policy, and the development of a consensus about these, can enhance the credibility of future evaluation.
- There is particular potential for using the concept of quality of life as a primary outcome indicator.
- More research to identify the full costs associated with disability, and the
  costs and benefits of disability related policies and programs is a
  prerequisite to meaningful cost effectiveness analysis.
- Participatory models of evaluation, involving consumers in the process, can enhance the relevance of evaluation questions and develop credibility and support for findings. Use of a formative, practical, action-oriented approach to evaluation can aid in the use of evaluation in the design and improvement of policies and programs.
- The text discussed a number of areas where additional evaluation and research can be helpful.
- Mechanisms for dissemination of information about the implications of evaluation findings, both internally and externally to government, can aid in more effective use of evaluation.

## 8. Conclusion: Overall Lessons Learned

This report has identified a number of lessons learned under each of the specific topic areas discussed. Following is a summary of overall findings and lessons learned which cut across specific topic or program areas. A more detailed summary of both general and specific lessons learned is presented in the overview of this report.

## Disability is likely to continue as a major public policy issue.

• The number of Canadians with disabilities (4.2 million in 1991, representing 15.5 per cent of the population) can be expected to increase as our society continues to age. Without effective interventions, this is likely to lead to increased pressure for services and to increased costs.

## Disability issues and topics are inter-related.

- There is substantial evidence, from Canada and abroad, indicating relationships and linkages across seemingly separate areas such as: employment, transportation, housing and living arrangements, personal support, income support, access to rehabilitation and training, and others. This in turn leads to related findings or lessons from the evaluation literature.
- A coordinated, holistic approach across federal government departments and with other jurisdictions is needed to address the complexity of issues and their linkages and to provide for coordination of policies and approaches. Without a coordinated approach, the result is frequently fragmented, with inconsistencies, overlaps, and gaps within and across jurisdictions.
- The all-or-nothing approach inherent in eligibility requirements of some income security programs, which labels people as either fully employable or unemployable, acts as a major disincentive to employment and reinforces the welfare trap.
- Environmental barriers have been found to be a greater impediment to employment and participation in society than functional limitations. Barrier removal, through legislative approaches, the availably of workplace accommodations, and through other means, has been identified as the key to the social integration of people with disabilities.

A variety of approaches that facilitate the participation of people with disabilities in employment and in independent living have proven cost effective in reducing the massive direct and indirect costs of disability.

• The text of this report has identified the cost effectiveness of a number of interventions, such as disability management, workplace accommodations and supported employment, as well as measures intended to permit independent living in the community.

Many approaches that permit the participation of people with disabilities can be low cost, but require the involvement of different sectors in society, frequently on a partnership basis.

- Most workplace accommodations are low in cost and many, such as modifications to the
  work task or rearrangement of the work site, require no additional expenditures. Costs
  are also minimal when accessibility is built into the design of facilities and generic
  programs, as opposed to retrofitting.
- Attitudinal barriers and lack of information about the capabilities of people with disabilities have been found to be among the biggest obstacles to independent living and to employment. These can result in systemic discrimination and the lack of opportunities for people with disabilities.
- The United states (ADA) experience has demonstrated that changes in attitudes can effectively follow changes in behaviour when standards and expectations are clearly set out. The Canadian approach has not been as effective: vague standards and lack of effective enforcement mechanisms often force individuals to litigate to prove discrimination, resulting in an adversarial environment.
- The provision of clear standards and the availability of technical information and support has led in the United States to increased compliance by employers and private and public facilities with the accessibility provisions of ADA.

Mainstreaming is necessary to ensure equality and to achieve the integration of people with disabilities.

• There is ample evidence that there is higher individual satisfaction and quality of life when individuals are able to make use of generic programs and services in the community, rather than having to use special or segregated programs and facilities.

Direct involvement of individuals with disabilities with programs, services and policies affecting them, as well as with evaluations, has been found to be associated with increased satisfaction and support, higher levels of functioning and greater success at community integration.

- Direct funding to enable people with disabilities to purchase and direct their own attendant care and personal support services has been shown to be more effective and to result in greater satisfaction than traditional approaches, where professionals decide on needs without involving the individual.
- Direct involvement of people with disabilities in policy reviews, such as with Mainstream 1992 Review and the 1996 Federal Task Force on Disability Issues, has demonstrated how this can lead to the development of a consensus and support for the directions taken.
- Similarly, there is evidence that the involvement of people with disabilities in evaluation studies can improve the quality and credibility of the evaluation, and lead to more relevant and improved policies and programs — and also lead to improved functioning of the individuals involved.

## There is insufficient information about the full costs of disability, which can lead to inappropriate and costly decisions.

- More research is needed to clearly identify the full range of direct and indirect costs associated with disability. This in turn will permit more meaningful determination of the cost-effectiveness of various programs and measures.
- Benefits from program measures and expenditures may accrue to different cost centres
  than those incurring the costs. This is one reason why, in spite of an impressive returnon-investment, disability management interventions have sometimes failed to obtain the
  support of senior management and human resources officials.

## Not enough is known about the reasons for the effectiveness of various types of approaches intended to aid in the community integration of people with disabilities.

- There is a need for more evaluation at various levels, ranging from the individual project level to consideration of the inter-relationships of programs cutting across different program areas and jurisdictions, in order to identify which approaches are most effective under given circumstances and to better target programs where they can be as effective as possible.
- There is a particular need for evaluation that identifies the reasons for effectiveness and facilitates learning from experience.

## Solid data about the disability population in Canada is needed for effective planning, policy development and evaluation.

 HALS has proven very useful in many different respects. It was discontinued, however, by Statistics Canada, after the 1991 survey. Lack of basic descriptive data can result in policy decisions being made in a vacuum. There appears to be an need for either the reinstatement of HALS, or the creation of an alternative means for identifying the size and characteristics of the disability population in Canada, such as piggybacking questions about disability on other survey vehicles.

## **BIBLIOGRAPHY**

Aarts, L.J.M. and De Jong, P.R. European Experiences with Disability Policy. In Mashaw, J. L. et al.(eds.). *Disability, Work and Cash Benefits*. Kalamazoo, Michigan: W.E. Upjohn Institute for Employment Research. (forthcoming as of 1996).

Aarts, L.J.M. and De Jong, P.R. *Contributions to Economic Analysis: Economic Aspects of Disability Behaviour.* Amsterdam, North-Holland. 1992.

Abrams, D., Barker, L.T., Haffey, W. and Nelson, H. The Economics of Return to Work for Survivors of Traumatic Brain Injury: Vocational Services are Worth the Investment. *Journal of Head Trauma Rehabilitation*. Vol. 8, No. 4. December, 1993.

Abt Associates of Canada. Evaluation of the Industry Canada Component of the National Strategy for the Integration of Persons with Disabilities. Federal-Provincial Relations, Planning and Evaluation Directorate, Industry Canada. 1995.

Abt Associates of Canada. Final Report: Evaluation of the Federal Contractors Program. Employment and Immigration Canada. June, 1992.

Abt Associates of Canada. Final Report: Evaluation of the Legislated Employment Equity Program. Employment and Immigration Canada. June, 1992.

Accreditation Council on Services for People with Disabilities. *Outcome Based Performance Measures*. Towson, MD.: The Accreditation Council. 1993.

Advisory Group on New Social Assistance Legislation. *Disability Determination Project Team Final Report*. Toronto, Ontario: Advisory Group on New Social Assistance Legislation. 1992.

Advocacy Resource Centre for the Handicapped. *Proceedings of the National Disability and Poverty Strategy Session*. 1989.

Agency for Health Care Policy and Research (U.S.). *Getting Better? Change or Error in the Measurement of Functional Limitations*. AHCPR Article Reprint Number 95-0040. Undated.

Alberta Health. Self Managed Care Pilot Project Evaluation. Alberta Home Care Program. 1993.

Allingham, Richard and Hyatt, Douglas. Measuring the Impact of Vocational Rehabilitation on the Probability of Post-Injury Return to Work. In Thomason, Terry and Chaykowski, Richard P. Eds. *Research in Canadian Workers' Compensation*. Kingston: IRC Press. 1995.

Altman, B. *Elderly Persons with Developmental Disabilities in long-term care Facilities* (AHCPR Pub. No. 95-0084). National Medical Expenditure Survey Research Findings 25, Agency for Health Care Policy and Research, Rockville, MD: Public Health Service. July, 1995.

American Council on Education. *College Freshman with Disabilities: A Triennial Statistical Profile*. Washington, D.C. 1995.

Andrews, Frank M. (Ed.). *Research on the Quality of Life*. Ann Arbor: Institute for Social Research, The University of Michigan. 1986.

AQEHPS. *Problématique reliée à la situation des étudiants ayant des limitations fonctionnelles majeures*. Mémoire présenté au Groupe de travail sur le Régime d'aide financière aux étudiants. Montréal. 11. Février 1995.

AQEHPS. Accessibilité et services aux étudiants(es) handicapés(es) dans les institutions d'enseignement post-secondaire au Québec. Rapport préliminaire. Association québécoise des étudiants handicapés au post-secondaire. Montréal. Octobre 1994.

Arruda, Maria, et al. *Position Paper on Case Management*. International Association of Psychosocial Rehabilitation Services, Ontario Chapter. 1991.

Association Québécoise de Loisir pour Personnes Handicapées et son Réseau Régional. Les centres de service en adaptation/readaptation au Québec pour les personnes vivant avec une déficience intellectuelle: le loisir au programme? Rapport d'enquête. Montréal. 236. Juin 1994.

Athanasou, J.A., Brown, D.J. and Murphy, G.C. Vocational Achievements following Spinal Cord Injury in Australia. *Disability and Rehabilitation*. 18, 191-196. 1996.

Atiyah, P. S. *Accidents, Compensation and the Law.* [5th edition authored by Peter Cane]. London, Dublin and Edinburgh: Butterworths. 1993.

Baer, Marvin. *The Legal Aspects of LTD Insurance*. Draft Report to the Ontario Law Reform Commission. 1996.

Bak, Lori and Dempsey, Stephen. Benefits Canada. Rising to the Challenge. *Industry Trends*. (On-line service). 1996.

Baker, David. *Tax Reform and People with a Disability*. Coalition of Provincial Organizations of the Handicapped. 1991.

Baker, David. Learning from the Best: Lessons for Ontario from the Employment Programs for Disabled People in Germany and Sweden. Report to the Chief Commissioner, Office of the Employment Equity Commission. 1992.

Barker, Linda Toms, Altman, Maya and Youngdahl, Andrea. Dimensions in Peer Counselling: Observations from the National Evaluation of Independent Living. *Issues in Independent Living*. No 7. Houston: ILRU Program. 1987.

Baume, Peter, and Kay, Kathleen. *Working Solutions*. Report of the Australia Law Reform Commission. 1996.

Beatty, Harry. *The Case for Comprehensive Disability Reform: A Report Prepared for Mainstream 1992*. Background paper prepared for the Canadian federal/provincial/territorial review of disability policy. 1992.

Beatty, Harry. Comprehensive Disability Compensation in Ontario: Towards an Agenda. *Journal of Law and Social Policy*. Volume 7, 100-142. 1991.

Benjamin A.E. Long-Term Care and AIDS: Perspectives from Experience with the Elderly. *The Milbank Quarterly*. 66(3):415-443. 1988.

Bergeron, Hélène, Cloutier, René, Fougeyrollas, Patrick and St.Michel, Ginette. *Quebec's Classification Proposal: Handicap Creation Process.* (QCICIDH - CSICIDH). 1991.

Berkowitz, M. (ed.) Forging Linkages: Modifying Disability Benefit Programs to Encourage Employment. New York: Rehabilitation International. 1990.

Berkowitz, M. Domestic Politics and International Expertise in the History of American Disability Policy. *The Milbank Quarterly*. Vol. 67, Supplement 2, Pt. 1, 195-227.

Berkowitz, M. and Hill, M.A. *Disability and the Labour Market: Economic Problems, policies and programs*. Cornell: ILR Press. 1986.

Berkowitz, M. *Disabled Policy - America's Programs for the Handicapped.* New York: Cambridge University Press. 1987.

Birch, Gary. Creative Employment Options: Economic Benefits. Burnaby: Neil Squire Foundation. 1996.

Blanck, Peter, David. Assessing Five Years of Employment Integration and Economic Opportunity under the Americans with Disabilities Act. *Mental and Physical Disability Law Reporter*. #19, 384. 1995.

Blinder, Alan S., Gordon, Roger H., and Wise, Donald E. Reconsidering the Work Disincentive Effects of Social Security. *National Tax Journal*. V. 33(4), 431-442. 1980.

Blue Ribbon Panel on National Telecommunications Policy. *Telecommunications and Persons with Disabilities: Building the Framework*. Oakland, California: World Institute on Disability. 1994.

Bolduc, Mario. *Making Equal Opportunity a Reality: An Insurance Plan Covering the Additional Costs Associated with Impairments, Disabilities and Handicap Situations*. Paper prepared for Independence '92 Conference, Vancouver, B.C. 1992.

Bolduc, Mario. Analyse de la cohérence de politiques québécoises à l'égard des personnes ayant des incapacité; Les clientèles ou problématiques considérées. Service de l'évaluation des programmes. Ministère de la Santé et des Services sociaux, Québec. 31.1986.

Borthwick-Duffy, Sharon A. Evaluation and Measurement of Quality of Life: Special Considerations for Persons with Mental Retardation. In Schalock. Vol. I, 105-119. 1996.

Boschen, Kathryn A. *Variables Affecting Independent Living for Persons with Physical Disabilities*. Project funded by National Welfare Grants Program. March, 1995.

Bound, J. The Health and Earnings of Rejected Disability Insurance Applicants. *The American Economic Review*. 79(3), 482-503. June, 1989.

Bousquet, Francine et Isabelle Guertin. Loi visant à assurer l'embauche de personnes handicapées; projet de loi. *CAMO pour personnes handicapées*. Montréal. 7.

Boxer, E., Young, S.M. and Kirchner, D. *Disability Case Management: a Review of the Literature*. Washington Business Group on Health. Washington, D.C. 1994.

Boydell, Katherine. *Measuring Health Outcomes*. Presentation at the Health and Mental Health Evaluators' Interest Group. Toronto. Oct. 29, 1996.

Brehm, Henry P. and Rush, Thomas V. Disability Analysis of Longitudinal Health Data: Policy Implications for Social Security Disability Insurance. *Journal of Aging Studies*. 2(4), 379-399. 1988.

Breslaw, Jon A., and Stelcner, Morton. The Effect of Health on the Labour Force Behavior of Elderly Men in Canada. *The Journal of Human Resources*. V. 22(4), 490-517. 1987.

British Columbia Association for Community Living, B.C. Coalition of Persons with Disabilities, and Canadian Mental Health Association (B.C. Division). *Income Security for People with Disabilities: A New Philosophy of GAIN for the Handicapped.* June, 1993.

Brown, Craig and Seto, Elizabeth Cummins. *No-Fault Automobile Insurance in Canada*. Toronto, Calgary and Vancouver: Carswell. 1988.

Brown, Joan C. The Disability Income System. London: Policy Studies Institute. 1984.

Burkhauser, Richard V., and Haveman, Robert H. *Disability and Work: The Economics of American Policy*. Baltimore: The Johns Hopkins University Press. 1982.

Butler, Richard J., Johnson, William G. and Baldwin, Marjorie L. Managing Work Disability: Why First Return to Work Is Not A Measure of Success. *Industrial and Labor Relations Review*. Vol.48, No.3. April 1995.

Byrnes, Mary. Outcome Evaluation Guidelines. Gage Transition to Independent Living. 1996.

C.S.N. *Le moment d'agir, c'est maintenant*. Document du Groupe de travail pour l'intégration et la réintégration en emploi des personnes handicapées. Montréal. 49. 1994.

Canadian Human Rights Commission. *Unequal Access: Accommodation of Employees with Disabilities in the Federal Public Service: A Case Study Approach.* 1996.

Canadian Human Rights Commission. *Unequal Access: An Accessibility Survey of Automated Banking Machines.* 1995.

Canadian Human Rights Commission. *Unequal Access: Availability of Federal Government Publications in Alternate Formats.* 1995.

Canadian Human Rights Commission. *Unequal Access: An Accessibility Survey of Selected Federal Offices*. 1995.

Canadian Human Rights Commission. Annual Report. 1995.

Canadian Human Rights Commission. *Unequal Access: Availability of TTY Services from Federal Departments and Private Sector Organizations*. 1994.

Canadian Human Rights Commission. *Unequal Access: An Accessibility Survey of Canadian Banking Facilities*. 1992.

Canadian Human Rights Commission. *Unequal Access:An Accessibility Survey of Canada Post Outlets*. 1992.

Canadian Human Rights Commission. *Unequal Access: Availability of TDD Services by Federal Departments.* 1991.

Canadian Human Rights Commission. *Unequal Access: Availability of Federal Publications in Alternate Format*, 1991.

Canadian Life and Health Insurance Association. Survey of Health Insurance Benefits in Canada. 1990-93.

Canadian Life and Health Insurance Association. *Disability Insurance: Where Will the Money Come From If You're Disabled?* Toronto, Ontario: CLHIA. 1992.

Canadian Mental Health Association. *Diversity Works: Accommodations in the Workplace for Employees With Mental Illness.* 1993.

Canadian Paraplegic Association and Manitoba League of the Physically Handicapped. *Securing Our Future: Renewing Canada's Social Security System.* May, 1994.

Canada Pension Plan Advisory Board. Report of the Committee on Disability Issues. December, 1994.

Cantrell, R. P. and Walker, M. L. Participatory Action Research and Policy Making: Consumers and Nonconsumers Deliberate the Reauthorization of the Rehabilitation Act. *Rehabilitation Counselling Bulletin*. Vol 37, No 1. Sept 1993.

Carpentier, René. *L'évaluation en éducation: situation et developpements*. Etude réalisée pour le Conseil supérieur de l'éducation dans le cadre de son rapport annuel 1992-1993. Montréal, 49. 1993.

Catalyst Research and Communications. Evaluation of the Special Initiative. 1995.

Cnaan, Ram A., Blankertz, Laura, Messinger, Karlyn W., and Gardner, Jerome R. Psychosocial Rehabilitation: Toward a Definition. *Psychosocial Rehabilitation Journal*. Vol 11 (4). April, 1988..

Comité ad hoc Rop-Cophan. *La sécurité sociale dans le Canada de demain: Une reconnaissance du droit à la différence*. Mémoire présenté au Comité permanent de développement des ressources humaines de la Chambre des communes. Montréal. 25. Décembre 1994.

Conley, Ronald W., Rusch, Frank R., McCaughrin, Wendy B. and Tines, Jeffrey. Benefits and Costs of Supported Employment: An Analysis of the Illinois Supported Employment Project. *Journal of Applied Behavior Analysis*. No. 4, 22, 441-447. Winter 1989.

Conley, Ronald, Assam, Riza and Mitchell, Arthur. *Evaluation of Transition to Work Demonstration Projects Using a Natural Supports Model*. Pelavin Research Institute. March, 1995.

Conroy, J.W., & Bradley, V.J. *The Pennhurst Longitudinal Study: A report of five years of research and analysis.* Philadelphia: Temple University Developmental Disabilities Center, Boston: Human Services Research Institute. 1985.

Conseil de L'Europe. *Note de la Délégation de la France relative au Plan d'action pour une politique d'insertion sociale et professionnelle en faveur des personnes handicapées*. Comité pour la réadaptation et l'intégration des personnes handicapées. 19e session à Strasbourg, 11 au 14 juin 1996, 11.

Consortium for Citizens with Disabilities Housing Task Force. *Opening Doors:* Recommendations for a Federal Policy to Address the Housing Needs of People with Disabilities. The Technical Assistance Collaborative, Inc., Boston, Mass. Sept. 1996.

COPHAN. A la veille de l'an 2000, L'égalité des chances: Bilan d'une décennie et perspectives d'avenir. Rapport de la Commission consultative sur la situation des personnes handicapées au Québec. Montréal. 148. 1991.

COPHAN. *Universal compensation fund for the disabled; feasibility study*. Summary report. Montréal. 13. November 1993.

COPHAN. Le crédit d'impôts pour personnes handicapées: coûts additionnels liés aux limitations fonctionnelles pour atteindre l'équité sociale. Document présenté aux ministères du Revenu du Québec et d'Ottawa. Montréal. 8. Décembre 1995.

Coudroglou, Aliki, and Poole, Dennis L. *Disability, Work, and Social Policy*. New York: Springer Publishing Company. 1984.

Council of Canadians with Disabilities. *Identifying the Holes in the Quilt: Social Security System and People with Disabilities.* Winnipeg: CCD. March, 1994.

D'Amour, Jean-Marie. Le développement de l'employabilité des personnes handicapées. *CAMO pour personnes handicapées*. Montréal. 13. Mai 1994.

D'Amour, Jean-Marie. Travail adapté pour les personnes handicapées productives mais non compétitives. *CAMO pour personnes handicapées*. Montréal. 12. Mai 1994.

D'Amour, Jean-Marie. L'harmonisation des programmes et des services d'intégration au travail et de maintien en emploi. Etudes préparatoires en vue de l'Avant-sommet du milieu associatif des personnes handicapées sur le devenir social et économique du Québec. *CAMO pour personnes handicapées*. Montréal. 18. Octobre 1996.

D'Amour, Jean-Marie. Services de main-d'oeuvre spécialisés et de réadaptation socio-professionnelle pour l'intégration à l'emploi des personnes handicapées. *CAMO pour personnes handicapées*. Montréal. 12. Mai 1994.

D'Amour, Jean-Marie. Formation générale et professionnelle des personnes handicapées. *CAMO pour personnes handicapées*. Montréal. 10. Mai 1994.

D'Amour, Jean-Marie. Télétravail, travail autonome et creation d'entreprises: nouvelles avenues ou cul-de-sac pour les personnes handicapées? *CAMO pour personnes handicapées*. Montréal. 11. Mai 1994.

D'Andrea, James A., Corry, David J. and Forester, Heather I. *Illness and Disability in the Workplace: How to Navigate Through the Legal Minefield.* Aurora, Ontario: Canada Law Book. 1996 - updated by supplements.

Davidson, Philip and Adams, Elizabeth. Indicators of Impact of Services on Persons with Developmental Disabilities: Issues Concerning Data-Collection Mandates in P.L. 100-146.

Decker, Paul T. and Thornton, Craig V. The Long-Term Effects of Transitional Employment Services. *Social Security Bulletin*. Summer, 1996.

DeJong, G., Batavia, A.I. and Griss, R. America's Neglected Health Minority: Working-age Persons with Disabilities. *The Milbank Quarterly.* 67 (Supplement 2, Part 2):311-351. 1989.

DeJong, Gerben. Environment Accessibility and Independent Living Outcomes: Directions for Disability Policy and Research. 1981.

Department of Finance Canada. *Disability Tax Credit - Evaluation of Recent Experience*. 1992.

Department of Health, UK. Assessing Older People with Dementia Living in the Community: Practice Issues for Social and Health Services. Social Services Inspectorate. Dec. 1996.

Department of Health, UK. Disability and Isolation: A Joint SSI/Arthritis Care Study of Isolated People with Arthritis. Social Services Inspectorate. Jan. 1996.

Department of Human Services and Health (Australia). *Evaluation of the Disability Reform Package*. Department of Human Services and Health. May 2, 1995.

Department of Human Services and Health (Australia). Review of the Disability Task Force. Department of Human Services and Health, Office of Disability. March, 1995.

Department of Justice. *Myths and Facts About the Americans with Disabilities Act.* Washington, D.C. 1996.

Department of Social Security, Australia. Annual Report 1994-95. 1995.

Department of the Secretary of State of Canada. *Evaluation of the Disabled Persons Participation Program (DPPP): Appendix B: Detailed Findings by Issues.* 1991.

Di Santo, O. *Is Universal Disability Insurance the Answer*? Occupational Health and Safety Canada. V. 8(6),78. Nov./Dec. 1992.

Di Giovanni, Jérôme. Pratique de gestion et utilisation des conventions collectives. *CAMO pour personnes handicapées*. Montréal. 14. Mai 1994.

Di Giovanni, Jérôme. Sommaire des législations provinciales et fédérales. *CAMO pour personnes handicapées*. Montréal. 10.

Drapeau, Maurice. *Un nouveau contrat de travail; Accès à l'égalité, ancienneté et discrimination*. Commission des droits de la personne du Québec. Montréal. 198. Février 1992.

Drug, C.E. *Expanding Employment of People With Significant Disabilities in the Public Sector.* University of Oregon, OR. 1993.

Dunn, Peter. *Barriers Confronting Seniors with Disabilities in Canada*. Ottawa: Statistics Canada. 1991.

Dupuis, Claude J. *The Meaning of "Disability" for Employment Equity Purposes: The Need to Keep a Narrow Definition.* Kingston: Queen's University. 1992.

Duval, Luce et al. *Portrait du champ de l'adaptation scolaire au Québec des années trente à nos jours*. Université de Sherbrooke, 126. 1995.

European Commission. Employment in Europe. Luxembourg, 1996a.

European Commission. Helios II European Guide of Good Practice: Towards Equal Opportunities for Disabled People. Luxembourg, 1996b.

European Day of Disabled Persons 1995. *Disabled Persons' Status in the European Treaties: Invisible Citizens*. Brussels: December 1995.

European Day of Disabled Persons Secretariat. *Towards Equalisation of Opportunities for Disabled People: Into the Mainstream?* Brussels: 1996.

F.T.Q. *Personnes handicapées; Les mythes*. Fédération des travailleurs et travailleuses du Québec. Montréal. 42. 1993.

F.T.Q. *Personnes handicapés; Les lois*. Fédération des travailleurs et travailleuses du Québec. Montréal. 30. 1993.

Falardeau, René. *De l'intégration sociale à la marginalisation et à l'appauvrissement.* Essai soumis à l'Ecole de service cosial de l'Université Laval. Québec. 1991.

Fardeau, Michèle et al. Le budget social du handicap. Les personnes handicapées en France; données sociales 1995. Paris, Editions INSERM-CTNERHI. 251-301. 1995.

Fawcett, Gail. *Living with Disability in Canada: An Economic Portrait*. Human Resources Development Canada, Office for Disability Issues. 1996.

Federal Task Force on Disability Issues. *Equal Citizenship for Canadians with Disabilities: The Will to Act.* Government of Canada. 1996.

Féderation de la Réadaptation en Déficience Physique du Québec et Féderation Québécoise des Centres d'Hébergement et de Soins de Longue Durée. *Réadaptation fonctionnelle intensive pour adultes ayant une déficience motrice: un partenariat efficace.* Montréal. 32. 1994.

Ferron, Donald T. Diagnostic Trends of Disabled Social Security Beneficiaries. *Social Security Bulletin*. Fall, 1995.

Flynn, Robert J., and Boschen, Kathryn A.P. Program Evaluation in Canadian Rehabilitation Services. In Marlett, N.J., Gall, R. and Wight-Felske A. Eds. *Dialogue on Disability: A Canadian Perspective, Vol. 1: The Service System.* The University of Calgary Press. 127-160. 1984.

FOCUS Vermillion. *Individualized Funding of Personal Support Services: Analysis of the Present System and Recommendations for Change.* Alberta Health and Social Services. 1992.

Foley, Susan. *ADA Activities of Professional and Trade Associations*. Paper presented to annual meeting of Society for Disability Studies. 1994.

Fortin, Pierre et Francine Seguin. *Pour un régime équitable axé sur l'emploi*. Rapport soumis à la ministre de la sécurité du revenu, Québec. 137. Mars 1996.

Fox, Daniel M. and Willis, David P. (eds.). Disability Policy: Restoring Socioeconomic Independence. *The Milbank Quarterly*. Vol. 67, Supplement 2, Parts 1 and 2.

Gallup Organization Inc. Baseline Study to Determine Business Attitudes, Awareness and Reaction to the Americans with Disabilities Act. Princeton, N.J. February, 1992.

GAO (General Accounting Office, United States). *Employment and Training: Successful Projects Share Common Strategy*. Testimony. 04/18/96. GAO/HEHS-96-127. 1996a.

GAO. *Medicaid Managed Care: Serving the Disabled Challenges State Programs*. Chapter Report, 07/31/96) GAO/HEHS-96-136. 1996b.

GAO. *Medicaid: Waiver Program for Developmentally Disabled is Promising but Poses Some Risks*. Letter Report, 7/22/96 GAO HEHS-96-120. 1996c.

GAO. *Medicare: Home Health Utilization Expands While Program Controls Deteriorate*. Report to the Chairman, Special Committee on Aging, U.S. Senate. GAO/HEHS-96-16, March 1996d.

GAO. Pass Program. SSA Work Incentive for Disabled Beneficiaries Poorly Managed. Washington, D.C. 1996e.

GAO. People With Disabilities: Federal Programs Could Work Together More Efficiently to Promote Employment. Washington, D.C. 1996f.

GAO. *Social Security: Disability Programs Lag in Promoting Return to Work.* Testimony. 05/23/96. GAO/T-HEHS-96-147. 1996g.

- GAO. SSA Disability: Program Redesign Necessary to Encourage Return to Work. GAO/HEHS-96-62. April 24, 1996h.
- GAO. SSA Disability: Return-to-Work Strategies From Other Systems May Improve Federal Programs. GAO/HEHS-96-133. July, 1996i.
- GAO. *Vocational Rehabilitation: VA Continues to Place Few Disabled Veterans in Jobs*. Report to the Chairman, Subcommittee on Education, Training, Employment, and Housing, Committee on Veterans' Affairs, House of Representatives. Letter Report, September 3, 1996. HEHS-96-155. 1996j.
- GAO. School Facilities: Accessibility for the Disabled Still an Issue. Washington D.C. December, 1995a.
- GAO. Social Security Disability: Management Action and Program Redesign Needed to Address Long-Standing Problems. GAO/T-HEHS-95-233. Aug. 3, 1995b.
- GAO. Social Security: Federal Disability Programs Face Major Issues. GAO/T-HEHS-95-97. Mar. 2, 1995c.
- GAO. Americans with Disabilities Act: Challenges Faced by Transit Agencies in Complying with Act's Requirements. Washington, D.C. 1994a.
- GAO. Americans with Disabilities Act: The Effects of the Law on Access to Goods and Services. Washington, D.C. June, 1994b.
- GAO. Long-Term Care: Diverse, Growing Population Includes Millions of Americans of All Ages. Letter Report, 11/07/94, GAO/HEHS-95-26. 1994c.
- GAO. Long-Term Care: Other Countries Tighten Budgets While Seeking Better Access. Chapter Report, 08/30/94, GAO/HEHS-94-154. 1994d.
- GAO. Multiple Employment Training Programs: Most Federal Agencies Do No Know If Their Programs Are Working Effectively. Letter Report. 03/02/94. GAO/HEHS-94-88. 1994e.
- GAO. Social Security: Disability Rolls Keep Growing While Explanations Remain Elusive. GAO/HEHS-94-34. Feb. 8, 1994f.
- GAO. Americans with Disabilities Act: Initial Accessibility Good But Important Barriers Remain. Washington, D.C. May, 1993a.
- GAO. Vocational Rehabilitation: Evidence for Federal Program's Effectiveness is Mixed. Washington, D.C. 1993b.

GAO. Vocational Rehabilitation: Clearer Guidance Could Help Focus Services on Those With Severe Disabilities. Washington, D.C. 1991.

Gibbs, W.E. Vocational Rehabilitation Outcome Measures: The Probability of Employment and the Duration of Periods of Employment. *American Rehabilitation*. Vol, 17, No. 1. 1991.

Global Strategy Group, Poll for the President's Committee on Employment of Persons with Disabilities. *Report on Disability Programs*. 2. January 4, 1996.

Goldman, J. and Gattozzi, A. Balance of Powers: Social Security and the Mentally Disabled. *The Milbank Quarterly*. Number 66, 531-551. 1988

Goode, David. (Ed.) *Quality of Life for Persons with Disabilities: International Perspectives and Issues*. Cambridge, MA: Brookline Books. 1994.

Goss Gilroy Inc. *Interdepartmental Evaluation of the National Strategy for the Integration of Persons with Disabilities.* HRDC. 1995(a). (Also *Appendices* in separate document). 1995(b).

Goss Gilroy Inc. *Transportation and Disability in Canada - An Overview*. Transport Canada. July, 1995(c).

Gouv. du Québec. L'aide financière aux étudiants: Un équilibre à maintenir. Rapport du Groupe de travail sur le Régime d'aide financière aux étudiants. Québec, 231. Septembre 1995.

Griss, Bob. Access to Health Care, Vol. 1, no. 3 & 4, Strategies for Adapting the Private and Public Health Insurance Systems to the Health Related Needs of Persons with Disabilities or Chronic Illness. World Institute on Disability. 1989.

Gruber, Jonathan. Canada Pension Plan, Disability Insurance Benefits and Labour Supply and Well-Being of Older Workers. MIT and NBER. August, 1996.

Guertin, Isabelle. Résumé d'une étude sur les politiques d'emploi pour les personnes handicapées de dix pays réalisée par Neil Lunt et Patricia Thornton, de l'Université de York. *CAMO pour personnes handicapées*. Montréal. 16. Mai 1994.

Hagner, David and Marrone, Joseph. Empowerment Issues in Services to Individuals with Disabilities. *Journal of Disability Policy Studies*. Vol. 6, no. 2, 17-36. 1995.

Hahn, H. The Political Implications of Disability Definitions and Data. *Journal of Disability Policy Studies*. 4(2), 41-52. 1993.

Haton, C., Emerson, E., Henderson, J. and Cooper, J. The Quality and Costs of Residential Services for Adults with Multiple Disabilities: A Comparative Evaluation. *Research and Developmental Disabilities*. Vol. 16-6. 439-60.

Haveman, Robert. Does the Welfare State Increase Welfare? Reflections on Hidden Negatives and Observed Positives. *The Economist*. V. 133(4), 445-466. 1985.

Haveman, Robert (ed.). *Public Policy Towards Disabled Workers*. Ithica: Cornell University Press, 1984.

Haveman, Robert H., and Wolfe, Barbara L. Decline in Male Labor Force Participation: Comment. *Journal of Political Economy*. 92(3), 532-541. 1984a.

Haveman, Robert H. and Wolfe, Barbara L. Disability Transfers and Early Retirement: A Causal Relationship? *Journal of Public Economics*. V. 24(1), 47-66. 1984b.

Haveman, Robert H., DeJong, G. and Wolfe, Barbara L. Disability Transfers and the Work Decision of Older Men. *The Quarterly Journal of Economics*. 939-949. August 1991.

Hawes, Catherine, Manard, Barbara, Kane, Rosalie and Mollica, Bob. *A National Study of Assisted Living*. Office of the Assistant Secretary for Planning and Evaluation (ASPE) of the USA Dept. Of Health and Human Services. Expected completion date is late 1997.

Health Canada. *National Strategy for the Integration of Persons with Disabilities: Evaluation of Health Canada Initiatives*. Program Evaluation Division, Policy and Consultation Branch, Health Canada. 1995.

Health and Welfare Canada. *Joint Federal-Provincial Study of a Comprehensive Disability Protection Program.* 1983. and *Stage II Report: Program Design Options.* 1985.

Hennessy, C.H. and Hennessy, M. Community-based Long-Term Care for the Elderly: Evaluation Practice Reconsidered. *Medical Care Review*. 47(2):221-259. Summer 1990.

Herr, S.S. Maximizing Autonomy: Reforming Personal Support Law in Sweden and the United States. *Journal of the Association for Persons with Severe Handicaps*. 20(3) 213-223. 1995.

Heumann, Judith E. *Notice of Proposed Funding Priority for Fiscal Years 1996-1997 for the Knowledge Dissemination Program.* National Institute on Disability and Rehabilitation Research, U.S. Department of Education. June 16, 1995.

Heyman, B. and Huckle, S. Normal Life in a Hazardous World: How Adults with Moderate Learning Difficulties and their Carers Cope with Risks and Dangers. *Disability, Handicap & Society*. Vol. 8, no. 2, 143-160. 1993.

HMSO. Community Care (Direct Payments) Act 1996. HMSO. Chapter 30. 1996.

Hollander, Anna. Rights to Special Services for People With Development Disabilities in Sweden. *Scandinavian Journal of Social Welfare*. 2, 63-68. 1993.

House of Commons Standing Committee on Human Rights and the Status of Disabled Persons. *As True as Taxes: Disability and the Income Tax System.* Ottawa. 1993.

House of Commons Standing Committee on Human Rights and the Status of Disabled Persons. *A Consensus for Action: The Economic Integration of Disabled Persons.* Ottawa. 1990.

Hulgin, Kathy, Shoultz, Bonnie, Walker, Pam and Drake, Steve. *Innovative Practices in Supported Living: An Overview of Organizations, Issues and Resource Materials*. Center on Human Policy, Syracuse University. May 1996.

Hum, Derek and Simpson, Wayne. *Sources of Income of Persons with Disabilities in Canada*. Ottawa: Employment Equity Data Program, Housing, Family and Social Statistics Division, Statistics Canada. 1994.

Hum, Derek and Simpson, Wayne. *The Economic Well-being and Labour Market Activity of Persons with Disabilities in Canada*. Ottawa: Employment Equity Data Program, Housing, Family and Social Statistics Division, Statistics Canada. 1993.

Human Resources Development Canada. *Evaluation of the Canada Pension Plan (Disability Component): Final Report*. Evaluation and Data Development, Strategic Policy. September 1996a.

Human Resources Development Canada. *Evaluation of the National Vocational Rehabilitation Project*. Evaluation and Data Development Strategic Policy. 1996b.

Human Resources Development Canada. *Returning to Work: Removing the Barriers*. CPP Program Direction, HRDC. 1996c.

Human Resources Development Canada. *Evaluation of the Employability Improvement Program. Final Report.* Evaluation and Data Development Strategic Policy. December 1995a. Human Resources Development Canada. *Evaluation of the HRDC Initiatives Under the National Strategy for Integration of Persons with Disabilities (NSIPD): Phase II.* Evaluation and Data Development. June, 1995b.

Human Resources Development Canada. *CPP Disability Incidence Study: Main Report*. Income Security Programs Branch. 1995c.

Human Resources Development Canada. *Improving Social Security In Canada. Persons With Disabilities: A Supplementary Paper.* 1994.

Human Resources Development Canada, Canada Pension Plan. *Fifteenth Actuarial Report*. December 31, 1993.

Hutchison, Peggy, Dunn, Peter, Lord, John and Pedlar, Alison. *Impact of Independent Living Resource Centres in Canada*. Centre for Research & Education in Human Services. 1996.

Insight Canada Research. A Review of the Disability Management Pilot Project. Report prepared for HRDC. 1996.

Institute for Community Inclusion. Families as Resources: The Art of Listening, Not Telling. *The Institute Brief.* July 1994.

Insurance Bureau of Canada. Facts of the General Insurance Industry in Canada. Toronto, Ontario: Insurance Bureau of Canada. 1996.

Ison, Terence G. Employment Quotas for Disable People: The Japanese Experience. *Kobe University Law Review* No. 26. 1992.

Ison, Terence G. Compensation Systems for Injury and Disease: The Policy Choices. Toronto and Vancouver: Butterworths Canada. 1994.

Ison, Terence G. *Workers' Compensation in Canada*. 2nd Edition. Toronto and Vancouver: Butterworths Canada. 1989.

James, Steven, Matier, Chris, Sakhini, Humam and Sheikh, Munir. *The Economics of Canada Pension Plan Reforms*. Working Paper: Department of Finance, Fiscal Policy and Economics Branch. November. 1995.

Jamieson, Joan and Hawkins, Linda. A literature review of Outcome and Cost-Benefit Studies on Supported Work Programs for Adults with Mental Handicaps. Research, Evaluation and Statistics Branch, Ministry of Social Services, Province of British Colombia. 1992.

Johnson, David R., Lewis, Darrell R., and Bruininks, Robert H. Evaluating the Effectiveness and Efficiency of Supported Employment Programs. *Policy Research Brief* (Institute on Community Integration, University of Minnesota), Vol. 5, No. 2. 1993.

Johnson, William G. and Baldwin, Marjorie. *Returns to Work by Ontario Workers with Permanent Partial Disabilities*. Ontario Workers' Compensation Board. 1993.

Keeler, John F. Social Insurance, Disability and Personal Injury: A Retrospective View. *University of Toronto Law Journal*. 44, 275. 1994.

Kelly, Michael W. Taking on the Terminator: Issues to Consider When Suing A Disability Insurer. *Advocates' Quarterly*. Vol. 14, 308-318. 1992.

Kemper, Peter. Living Arrangements and the Demand for Home Care: An Illustrative Theoretical Model. Presented at annual meetings of the Allied Social Science Assoc. in Atlanta on Dec 28-30, 1989. Reprint from Agency for Health Care Policy and Research. April 1989.

Kregel, John and Wehman, Paul. Supported Employment Research: Impacting the Work Outcomes of Individuals with Disabilities. *Rehabilitation Research and Training Centre Newsletter*. Virginia Commonwealth University. Richmond. Summer, 1986.

Kulkarni, Madhav R. Quota Systems and the Employment of the Handicapped. Undated.

Kuyvenhoven, Marijke et al. Independently Living Disabled Elderly: First Results from an Innovative Home Care Project in Haarlem, The Netherlands. *Home Health Care Services Quarterly*. 15(2), 99-110. 1995.

Labregere, A. Un quart de siècle d'éducation des amblyopes en France. *Réadaptation*. 346: 24-30. Jany. 1988.

Lakin, K.C., Blake, B.M., Prouty, R.W. et al. *Residential Services for Persons with Developmental Disabilities: Status and Trends Through 1991*. Minneapolis: University of Minnesota, Institute on Community Integration.

LaPlante, M.P. The Demographics of Disability. *The Milbank Quarterly*. 69(½)55-77. 1991.

Lauzon, Léo-Paul. La fiscalité. Etudes préparatoires en vue de l'Avant-sommet du milieu associatif des personnes handicapées sur le devenir social et économique du Québec. *CAMO pour personnes handicapées*. Montréal. 26. Octobre 1996.

Le Groupe DBSF. La sécurité sociale et le rôle et la force économique des personnes handicapées. Etudes préparatoires en vue de l'Avant-sommet du milieu associatif des personnes handicapées sur le devenir social et économique du Québec. *CAMO pour personnes handicapées*. Montréal. 30. Octobre 1996.

Lemieux-Brassard, Lucie. *Comment la codification juridique québécoise constitue-t-elle un obstacle environnemental à l'intégration sociale des personnes handicapées?* Examen de synthèse présenté au programme de Ph.D.en Sciences humaines appliquées de l'Université de Montréal. Décembre 1995.

Leonard, Jonathan S. Labour Supply Incentives and Disincentives for Disabled Persons. In Berkowitz and Hill, eds. *Disability and the Labour Market*. Ithaca: ILR Press (Cornell University). 64-93. 1986.

Lepofsky, David. A Report Card on the Charter's Guarantee of Equality to Persons with Disabilities: After Ten Years - What Progress? What Prospects? *National Journal of Constitutional Law.* (forthcoming, 1996).

Lesemann, Frederic and Martin, Claude. Eds. *Home-based Care, The Elderly, the Family and the Welfare State: An International Comparison*. University of Ottawa Press. 1993.

Lewin-VHI, Inc. *National Study of Assisted Living for the Frail Elderly: Literature Review Update*. Prepared for the Office of the Assistant Secretary for Planning and Evaluation and the Administration on Aging by Lewin-VHI, Inc., subcontractors to Research Triangle Institute. Feb. 1996.

Lindsey, P., Jacobson, P. and Pascal, A. Medicaid Home and Community-Based Waivers for Acquired Immunodeficiency Syndrome Patients. *Health Care Financing Review*. 1990 Annual Supplement. 109-118. Dec. 1990.

Liu, K., Manton, K.G. and Liu, B.M. Home Care Expenses for the Disabled Elderly. *Health Care Financing Review.* 7(2):51-58. Feb. 1986.

Lord, John, McGeown, Mary, Taylor, Andrew and Young, Sherri. *More Than Just Another Human Service: Independent Living Centres in Ontario*. Centre for Research & Education in Human Services. 1992.

Lord, John, Hutchison, Peggy and Farlow, D'Arcy. *Independence and Control: Today's Dream, Tomorrow's Reality: Review of Support Service Needs of Adults with Physical Disabilities in Ontario.* Centre for Research & Education in Human Services. 1988.

Lord, John, Ochocka, Joanna and Roth, Don. *Review of Individualized Funding*. Centre for Research & Education in Human Services. 1993.

Louis Harris and Associates. *The NOD/Harris Survey on Employment of Persons with Disabilities*. New York, New York. July 7, 1995.

Louis Harris and Associates. *Public Attitudes Toward People with Disabilities*. Conducted for National Organization on Disability. New York, New York. June, 1991.

Lunt, Neil and Thornton, Patricia. *Employment Policies for Disabled People: A Review of Legislation and Services in Fifteen Countries*. Social Policy Research Unit, Employment Department. 1993.

Lutfiyya, Zana, Marie. Qualitative Research: Its Value and Role in Policymaking. *Policy Research Brief.* Vol.3, No.1. Fall, 1991.

Mainstream 1992. The Federal/Provincial/Territorial Review of Services Affecting Canadians With Disabilities. *Pathway to Integration*. May, 1993.

Maki, Dennis R. The Economic Implications of Disability Insurance in Canada. *Journal of Labour Economics*. V. 11(1) pt. 2, S148-S169. January 1993.

Marrone, Joe, Hoff, David and Helm, David T. *Person-Centered Planning for the Millennium:* We're Old Enough to Remember When PCP Was Just a Drug. Institute for Community Inclusion, Children's Hospital, Boston, Mass.

Marrone, Joe. If Everybody is Doing it, How Come it Doesn't Get Done? *Psychosocial Rehabilitation Journal*. Vol. 18, no. 1, 131-134. July 1994.

Mashaw, Jerry L. and Reno, Virginia P., eds. *Balancing Security and Opportunity: The Challenge of Disability Income Policy*. Washington D.C.: National Academy of Social Insurance. 1996.

Mashaw, Jerry L., Reno, Virginia P., Burkhauser, Richard V. and Berkowitz, Monroe. (Eds.) *Disability, Work and Cash Benefits*. Kalamazoo, Michigan: W.E.Upjohn Institute for Employment Research. 1996.

McDowell, Banks. *The Crisis in Insurance Regulation*. Westport, Connecticut and London, England: Quorum Books. 1994.

McIlraith, Audrey. *Projet pilote sur l'aide au travail indépendant pour les personnes handicapées; rapport d'évaluation.* Conseil canadien de la réadaptation et du travail. Juin 1996.

McNeil, Jack. *Household Survey Data on Employment and Disability*. Washington, D.C.: U.S. Bureau of Census. 1996.

Mendelsohn, Steve. *Accessibility in the Federal Workplace*. Assistive Technology Funding and Systems Change Project. Washington D.C. 1995.

Metro Toronto DHC. Integration and Independence City of Toronto cited in *It's My Choice:* Strategic Framework for Services with Individuals with Physical Disabilities. June 1995.

Miller, Robert H. Containing Use and Expenditures in Publicly Insured Long-Term Care Programs. *Health Care Financing Review*. Vol. 14, no. 4. Summer 1993.

Ministère de l'Education. *Enquête sur le mode de vie des étudiants du secondaire professionnel et du collégial. E*nquête réalisée par le Bureau de la statistique du Québec à la demande du ministère de l'Education du Québec. Québec. 272. Août 1996.

Ministère de la Santé et des Services Sociaux de Québec. L'allocation directe au Québec: des modes de fonctionnement variés à découvrir des points de vue à faire connaître. Direction générale de la planification et de l'évaluation. Québec. 145. Juin 1996.

Mohr, Lawrence B. *Impact Analysis for Program Evaluation*. London and Thousand Oaks, CA: Sage Publications. 1988.

Murphy, G., Brown, D., Foreman, P., Young A. and Athanasou, J. Paraplegia, Quadriplegia and Employment in Australia. *Australian Journal of Career Development*. 5, 26-31. 1996.

Murphy, G.C. and Athanasou, J.A. Vocational Potential and Spinal Cord Injuries - A Review and Evaluation. *Journal of Applied Rehabilitation Counselling*. 25, 47-52. 1994.

Muszynski, Leon. *Comprehensive Disability Income Security Reform*. Toronto: The G. Allan Roeher Institute. 1992.

Muszynski, Leon. *The Special Needs of Persons with Disabilities*. Department of Secretary of State, Status of Disabled Persons Secretariat. 1992.

National Council on Disability. Making Health Care Reform Work for Americans with Disabilities: Summary Information on Five "Town Meetings" on Health Care Reform: A Report to the President and Congress of the United States. 1994.

National Council on Disability. Voices of Freedom: America Speaks Out on the ADA. Washington, D.C. 1995.

National Council on Disability. *Achieving Independence: The Challenge of the 21st Century*. Washington, D.C. 1996.

National Educational Association of Disabled Students (NEADS). *Study of Financial Assistance Available to Post-Secondary Students with Disabilities: Accommodating Individual Needs for the Future*. October, 1993

National Health Service, UK. Report on Residential Services for People with Learning Disabilities. June 8, 1996.

National Institute on Disability and Rehabilitation Research. Supported Employment for People with Severe Mental Retardation. *Bringing Research into Effective Focus*. Vol.16, No. 2. 1994.

National Transportation Agency of Canada. Estimates. 1996/97.

National Transportation Agency of Canada. Estimates. Estimates 1995/96.

National Transportation Agency of Canada. Estimates. 1994/95.

National Transportation Agency of Canada. Estimates. 1993/94. Part III - Expenditure Plan.

National Transportation Agency of Canada. Revue de la Stratégie Nationale Pour L'Intégration des personnes handicapées. Undated.

Neil Squire Foundation. A Report on Equality, Participation and Inclusion in the Community (EPIC). Vancouver. March, 1994.

Nosek, Margaret A. Effect of Personal Assistance Services on the Long-Term Health of a Rehabilitation Hospital Population. 1991.

Nosek, Margaret A. Personal Assistance Services: A Review of Literature and Analysis of Policy Implications. *Journal of Disability Policy Studies*. Vol. 2, no. 2. 1991.

O'Brien, John. *Getting the Job Done. Learning to Expand the Social Resources Available to People with Severe Disabilities at Work.* Unpublished paper. University of Minnesota Institute on Community Integration. 1996.

Office des Personnes Handicapées du Quebec. *Programme d'évaluation de l'intégration sociale des personnes handicapées. Cadre théorique et démarche d'évaluation préparé en vue du colloque CQCIDIH: Les déterminants environnementaux de la participation sociale.* Drummondville. 38. Septembre 1994.

Office des Personnes Handicapés de Québec. *Le suivi des programmes compensatoires dans le cadre du programme d'évaluation de l'intégration sociale.* Drummondville. 44. Juin 1994.

Office des Personnes Handicapées du Quebec. *Programme d'évaluation de l'intégration sociale. Document d'orientation*. Drummondville. 15. Juin 1994.

Office des Personnes Handicappées du Québec. *Nouvelles formules de travail adapté et soutenu en entreprise régulière; Evaluation de l'expérimentation*. Drummondville. 86. Juillet 1994,.

Oliver, Mike. Changing the Social Relation of Research Production. *Disability Handicap and Society* 7(2), 105-114. 1992.

Oliver, Mike. The Politics of Disablement. London: Macmillan. 1990.

Ontario Public Service. Employment Equity Annual Report, 1992. 1992.

Ontario Task Force on the Vocational Rehabilitation Services of the Workers' Compensation Board. *An Injury to One Is An Injury to All*. Government of Ontario. 1987.

Ordre des Psychologues du Quebec. Des acquis au profit de la réussite éducative. Mémoire présenté à la Commission des Etats généraux sur l'éducation. Mont-Royal, 22. Août 1995.

Ouellette-Kuntz, Heléne, McCreary, Bruce D., Minnes, Paricia, and Stanton, Barbara. Evaluating Quality of Life: The Development of the Quality of Life Interveiw Schedule (QUOLIS). *Journal on Developmental Disabilities*. Vol 3. 17f. 1994.

O'Brien, John. Issues and Challenges in Developing Individualized Supports: Developing Inclusive Self-directed Personal Assistance Services: One Agency's Experiences. Center for Human Policy.

O'Donnell, Allan. *Automobile Insurance in Ontario*. Markham, Ontario: Butterworths Canada Ltd. 1991.

Parsons, Donald O. The Decline in Male Labor Force Participation. *Journal of Political Economy*. V. 88(1), 117-134. February 1980. 1980a.

Paulsson, Karin. L'intégration scolaire des élèves handicapés moteurs. *Réadaptation*. 290,3-9. Mai 1982.

Peacock, Mark G. Recent developments in Education law affecting the linguistic minority and handicapped students in Quebec: Public education that is "separate but equal". *Développements récents en droit de l'éducation*. Editions Yvon Blais, 25-70. 1996.

Percy, Stephen L. *Disability, Civil Rights and Public Policy - The Politics of Implementation*. Tuscaloosa: University of Alabama Press. 1989.

Perrin, Burt. Disability and Labour Market Integration: Clarifying Federal Responsibilities in the Evolving Social Union. Ottawa: Paper prepared for the Federal Task Force on Disability. 1996.

Perrin, Burt. Evaluation and Future Directions for the Job Accommodation Network (JAN) in Canada. Final Report. Employment Policies and Operations, HRDC. 1995.

Perrin, Burt. Exploring the Effectiveness of Canada Employment Centres for People With Disabilities. Final Report. Employment Equity Branch. Employment and Immigration Canada. 1994.

Perrin, Burt. Assessment of the Mainstream 1992 Consultation Process. Final Report. Mainstream 1992 Review. July, 1993.

Perrin, Burt. *The Future of JAN: Evaluation and Identification of Future Options. Final Report.* Employment and Immigration Canada. 1990.

Perrin, Burt. *Unused People, Unused Policies: Career Development Among OPS Employees With Disabilities.* Office for Disability Issues. Ontario. 1993.

Perrin, Burt. Extraordinary Costs of Disability: Conceptual Considerations. Toronto: Office for Disability Issues. 1991.

Perrin, Burt. *Innovative Approaches to Employment For Persons With Disabilities*. Disabled Persons Participation Program, Secretary of State of Canada. 1987.

Perrin, Burt. *Trends and Innovations Regarding Employment for People with Handicaps*. Toronto: Ministry of Community and Social Services. 1986.

Pettigrew, Pierre S. Letter to Task Force Participants. Minister of Human Resources Development. 20 March 1997.

Pfeiffer, Naomi and Coote, Anna. *Is Quality Good for You? A critical review of quality assurance in welfare services.* Institute for Public Policy Research. Social Policy Paper No. 5. 1993.

Pfeiffer, David. Major Problems with the ICIDH. *Proceedings of the Society for Disability Studies*. 1995.

Pfeiffer, David and Finn, Joan. Survey Shows State, Territorial, Local Public Officials Implementing ADA. *Mental and Physical Disability Law Reporter.* #19, 537. 1995.

Pinet, Genevieve, ed. *Is the Law Fair to the Disabled? A European Survey*. Copenhagen: World Health Organization Regional Office for Europe. 1990.

Policy Studies Associates Inc. *Evaluation of the Projects with Industry (PWI) Program - Final Report.* Rehabilitation Services Administration. U.S. Education Department. January 1986.

PRA Prairie Research Associates Inc. *Review of the Diand Component of the National Strategy for the Integration of Persons with Disabilities (NSIPD)*. Department of Indian Affairs and Northern Development (DIAND). 1995.

Price Waterhouse. A Formative Evaluation (Mid-Term Review) of the National Strategy for the Integration of Persons with Disabilities. Department of the Secretary of State of Canada. 1993.

Price Waterhouse. *Evaluation of the Disabled Persons Participation Program: Final Report.*Department of the Secretary of State of Canada. 1991.

Prince, Michael J. Touching Us All: International Context, National Policies, and the Integration of Canadians with Disabilities. In Abele, Frances. Ed. *How Ottawa Spends. The Politics of Competitiveness* 1992-93. Ottawa: Carleton University Press. 1992.

Prichard, J. Robert S. Liability and Compensation in Health Care: A Report to the Conference of Deputy Ministers of Health of the Federal/Provincial/Territorial Review on Liability and Compensation Issues in Health Care. Toronto, Ontario: University of Toronto Press. 1990.

Prost, Alar. *The Dollars and Sense of Disability Management*. Presentatation to the Staff Relations Symposium, Cornwall. October, 1996.

Raskin, C. Canada's Employment Programmes and Policies for Persons with Disabilities, a Study of Cost-effectiveness: Qualitative and Quantitative Considerations. Employment Department, International Labour Organization, Geneva. 1995.

Ratzka, Adolf. *A Brief Survey of Studies on Costs and Benefits of Non-Handicapping Environments*. Presentation at the International Congress on Accessibility, Rio de Janeiro, Brazil. June 1994.

Ravaud, Jean-François. *La scolarisation des personnes handicapées. Les personnes handicapées en France; données sociales 1995.* Paris, Editions INSERM-CTNERHI, 81-126. 1995.

Regroupement des Semo Personnes Handicapées. *Etude de rentabilité socio-économique des services externes de main-d'oeuvre (SEMO) pour personnes handicapees*. Groupe-Conseil DOME. Montréal. 44. Novembre 1995.

Rehab Group, Inc. A Comprehensive Review and Evaluation of Individual Habiltation Plans and Case Management Systems: Design Specifications for Case Management/Service Coordnation and Individualized Planning. Rehabilitation Services Administration. 1980.

Rehabilitation International Commission on Organization and Administration. *Available Disability-Related Databases*. Helsinki: Rehabilitation International. August 1, 1994.

Renwick, Rebecca, Brown, Ivan, and Raphael, Dennis. Quality of Life: A Conceptual Approach to Service Provision. *Journal of Developmental Disabilities*. 3(2) 32-44. 1994.

Revenue Canada. Interpretation Bulletin IT-519R. 1995.

Riddle, Dorothy I. *Analysis of Corporate Data on the Implementation of the Designated Group Policy*. Phase I. Ottawa, HRDC. 1994.

Riddle, Dorothy I. Assessment of the Implementation of the Designated Group Policy. Phase II Report. Ottawa, HRDC. 1994.

Rioux, Marcia H. and Bach, Michael. (Eds.) *Disability Is Not Measles. New Research Paradigms in Disability*. Roeher Institute. 1994.

Rioux, Marcia and Crawford, Cameron. Poverty and Disability. *Canadian Journal of Mental Health*. Volume 9, No. 2, 97-109. 1990.

Roeher Institute. Disability, Community and Society: Exploring the Links. Toronto. 1996.

Roeher Institute. *Disability-Related Supports: Costs and Delivery System for Selected Provinces.* Toronto. 1995a.

Roeher Institute. Research Issues in Evaluating Deinstitutionalization: A Review of the Literature. Toronto. 1995b.

Roeher Institute. People with Disabilities in Transition, The Roeher Institute. Toronto. 1994a.

Roeher Institute. The Canadian Disability Resource Program: Offsetting Costs of Disability and Assuring Access to Disability-Related Supports. Toronto. 1994b.

Roeher Institute. *Direct Dollars: A Study of Individuallized Funding in Canada*. Toronto. 1993.

Roeher Institute. A Study of Personal Supports in Ontario. Toronto. 1992.

Roeher Institute. On Target? Canada's Employment-Related Programs for Persons with Disabilities. Toronto. 1992.

Roeher Institute. *The Power to Choose: An Examination of Service Brokerage and Individualized Funding as Implemented by the Community Living Society.* Toronto. 1991.

Ross, David P. and Shillington, Richard. *An Economic Profile of Persons with Disabilities in Canada*. Ottawa: Secretary of State for Canada. 1990.

Roy, Catherine. Actes du Forum sur l'Emploi et la Formation Générale et Professionnelle. *CAMO pour personnes handicapées*. Montréal. 58. Novembre 1995. Rubin, Jeffrey. *Economic Consequences of Severe Disability*. New Brunswick, N.J.: Rutgers University Bureau of Economic Research. February, 1986.

Rupp, Kalman and Stapleton, David. Determinants of the Growth in SSA's Disability Programs - An Overview. *Social Security Bulletin*. Winter, 1995.

Sainsbury, Roy, Hirst, Michael, and Lawson, Dot. *Evaluation of Disability Living Allowance and Attendance Allowance*. Research Report No. 41. Department of Social Security (U.K.). 1995.

Sales, Arnaud et al. *Le monde étudiants à la fin du XXe siècle. Rapport final sur les conditions de vie des étudiants universitaires dans les années quatre-vingt-dix.* Présenté au ministére de l'Education du Québec par le département de sociologie de l'Université de Montréal. Québec. 372. Mai 1996.

Samorodov, A. *Indicators of Cost-effectiveness of Policy Options for Workers with Disabilities*. Employment Department, Labour Market Policies Branch, International Labour Office, Geneva. 1996.

Sapey, B. Disabling Homes: A Study of the Housing Needs of Disabled People in Cornwall. *Disability & Society.* 10(1). 1995.

Schalock, Robert L. Promoting Quality Through Quality Enhancement Techniques and Outcome Based Evaluation. *Journal of Developmental Disabilities*. 3(2), 1-16. 1994.

Schalock, Robert L. (Ed.) *Quality of Life. Vol. I: Conceptualization and Measurement.* Washington: American Association on Mental Retardation. (also *Vol. II: Applications to Persons with Disabilities.*) 1996.

Scotch, Richard K. Understanding Disability Policy: Varieties of Analysis. *Policy Studies Journal*. 22(1), 170-175. January, 1994.

Sharing the Risk and Ensuring Independence: A Disability Perspective on Access to Health Insurance and Health-Related Services: A Report to the President and the Congress. 1993.

Sheets, D.J., Wray, L.A. and Torres-Gil, F.M. Geriatric Rehabilitation: Linking Aging Health and Disability Policy. *Topics in Geriatric Rehabilitation*. Vol.9, no.2, 1-17. Dec 1993.

Sherman, David. Taxes, Health & Disabilities. Toronto: Carswell. 1995.

Silverman, M., Musa, D., Martin, D., Lave, D.C., Adams, J.R. and Ricci, J. Evaluation of Outpatient Geriatric Assessment: A Randomized Multi-site trial. *Journal l of the American Geriatrics Society*. Vol 43, no. 7. 733-740. July 1995.

Sjögren, Olle. *Individualized Service Programs; A Best Model for Support and Service to Children, Young People and Adults with Disabilities and their Families.* Tome 1 and Tome 2. Sweden. 1996.

Slater, David. Reforming Canada's Retirement Income System. *Canadian Business Economics*. 47-58. Fall, 1995.

Smith, Quentin, Smith, Laura W., King, Kym, Frieden, Lex and Richards, Laurel. *Health Care Reform, Independent Living, and People with Disabilities.* Issues in Independent Living No. 11, Houston: ILRU Program. 1993.

Social Assistance Review Committee. *Transitions*. Toronto: Ontario Ministry of Community and Social Services. 1988.

Social Policy Research Associates/The Evaluation Group Inc. *An Evaluation of the Program for the Employment Disadvantaged*. Employment and Immigration Canada. March 1984.

Societé Logique. Comparaison des réglementations provinciales en matière d'accessibilité architecturale (section 3.7, CNB 1990). Montréal. Février 1994.

SOM Inc. Evaluation des programmes travail: Contrat d intégration au travail et centre de travail adapté. Office des personnes handicapées du Québec. 172. Avril 1996.

Sopinka, John. Lange v. City of Stratford. Supreme Court of Canada.

Special Committee on the Disabled and the Handicapped. *Obstacles*. Ottawa. February 1981.

Spillman, Brenda C. and Kemper, Peter. Long Term Care Arrangements for Elderly Persons with Disabilities: Private and Public Roles, *Home Health Care Services Quarterly*. Vol. 13(1/2) 5-34. 1992.

Steinfeld, E and Shea, S. Enabling Home Environments: Identifying Barriers to Independence. *Technology and Disability*. Vol. 2, no. 4, 69-79. Fall 1993.

SPR Associates. *Evaluation of the National Vocational Rehabilitation Project*. Evaluation and Data Development, HRDC. 1996.

Standing Committee on Human Rights and the Status of Disabled Persons. *The Grand Design: Achieving the 'Open House' Vision*. Report of the Standing Committee on Human rights and the Status of Disabled Persons. December, 1995.

Statistics Canada. A Portrait of Persons With Disabilities. Target Groups Project. 1995.

Statistics Canada. Work Injuries 1994. The Daily. Friday, December 8, 1995.

Statistics Canada. *Selected Characteristics of Persons with Disabilities Residing in Households*. 1991 Health and Activity Limitation Survey. November 1994.

Statistics Canada. *Adults with Disabilities: Their Employment and Education Characteristics*. 1991 Health and Activity Limitation Survey. June 1993.

Stern, Steven. Measuring the Effect of Disability on Labour Force Participation. *The Journal of Human Resources*. V. 24(3), 361-395. 1989.

Stone, Deborah. *The Disabled State*. Philadelphia: Temple University Press. 1984.

Strauss, Karen, Peltz. Telecommunications. In Gostin and Beyer (eds.). *Implementing the ADA*. Brookes. 1993.

Suffolk Coastal and Mid Suffolk Disability Housing Research Group. *Disabling Housing. The Hidden Housing Needs of Disabled People.* 1994.

Sugarman, Stephen. *Doing Away with Personal Injury Law: New Compensation Mechanisms for Victims, Consumers and Business.* New York, Westport, Conn. and London: Quorum Books. 1989.

Swedish Institute. Disability Policies in Sweden. Fact Sheets on Sweden. January, 1995.

Swedish Disability Commission. *Disability welfare justice*. Summary Report of the 1989 Disability Commission. Stockholm. 40. 1991.

Tavistock Institute. Evaluation of the Helios II Programme. Summary Results of European Survey of Information and Exchange Participants. October, 1996.

Tavistock Institute. The Evaluation of Helios II. Interim Report. October, 1995.

Thomason, Terry and Chaykowski, Richard P. (Eds). *Research in Canadian Workers' Compensation*. Kingston, Ontario: Queen's University Industrial Relations Centre IRC Press. 1995.

Thompson, Lawrence H. The Advantages and Disadvantages of Different Social Welfare Strategies. *Social Security Bulletin*. V.57(3), 3-11. Fall, 1994.

Torjman, Sherri Resin. *Income Insecurity: the Disability Income System in Canada*. Toronto: The G. Allan Roeher Institute. 1988.

Tourigny, André et al. L'allocation directe au Québec: Des modes de fonctionnement variés à découvrir, des points à faire connaître. Québec: Ministère de la santé et des services sociaux. Gouvernement du Québec. 1996.

Tremblay, Nicole. Orientation professionnelle. *CAMO pour personnes handicapées*. Montréal. 11. Mai 1994.

United Cerebral Palsy Association. 1996 ADA Snapshot of America. Washington, D.C. 1996.

United Cerebral Palsy Association of America. *An ADA Employment Report Card on America*. Washington, D.C. 1993.

United Nations. Standard Rules on the Equalization of Opportunities for Persons with Disabilities. 1994.

United States Dept. Of Education. *Annual Performance Report, Independent Living Programs, Fiscal Year 1995*. Office of Special Education and Rehabilitation Services Administration pursuant to the Rehabilitation Act.

Velche, Dominique. L'emploi des personnes handicapées. Les personnes handicapées en France; données sociales 1995. Paris, Editions INSERM-CTNERHI. 127-250. 1995.

Verificateur Général du Québec. Rapport à la direction de l'Office des personnes handicapées du Québec concernant certaines activités liées aux mesures d'intégration au travail des personnes handicapées. Québec.

Verificateur General du Québec. Rapport à la direction de l'Office des personnes handicapées du Québec concernant certaines activités liées aux mesures d'intégration au travail des personnes handicapées. Québec. 1993.

Vladeck, Bruce, Miller, C., Miller, Nancy A. and Clauser, Steve B. The Changing Face of Long-Term Care. *Health Care Financing Review*. Vol. 14, no. 4. Summer 1993.

Waddington, Lisa, Hendriks, Aart, McCarthy, Tom and Wall, John. How Can Disabled Persons in the European Union Achieve Equal Rights as Citizens? The Legal and Economic

Implications of a Non-Discrimination Clause in the Treaty on European Union. European Day of Disabled Persons. 1996.

Washington Business Group on Health. Case Management: Improving Employment Opportunities for Persons with Disabilities and Developing a Cost-Benefit Model: Final Report. Washington, D.C. 1994.

Weed, R.O., Lewis, S.H. Workers' Compensation Rehabilitation and Case Management are Cost Effective: True or False? *Journal of Rehabilitation Administration*. Vol. 18, No. 4. November 1994.

Wehman, Paul and Kregel, John. *Supported Work Approach to Competitive Employment of Individuals with Moderate and Severe Handicaps*. Rehabilitation Research & Training Centre of Education. Virginia Commonwealth University. Richmond. 1984.

Wehman, P. and Hill, M. Comprehensive Benefit Cost Analysis of Supported Employment. *Rehabilitation Research and Training Centre Newsletter*. Virginia Commonwealth University. Richmond. 1986.

Weiler, Paul C. Permanent Partial Disability: Alternative Models for Compensation. Toronto, Ontario: Ontario Ministry of Labour. 1986.

Weiler, Paul C. *Protecting the Worker from Disability: Challenges for the Eighties*. Toronto: Ontario Ministry of Labour. 1983.

Weiler, Paul C. Re-shaping Workers' Compensation for Ontario. Toronto, Ontario: Ontario Ministry of Labour. 1980.

Weissert, W.G., Cready, C.M. and Pawelak, J.E. The Past and Future of Home and Community-Based Long-Term Care. *The Milbank Quarterly* 66(2):309-388, Summer 1988.

West, Jane. (ed.). *Implementation of the Americans with Disabilities Act*. New York, New York: Millbank Memorial Fund. 1995.

West, Jane. (ed.). *The Americans with Disabilities Act: From Policy to Practice*. New York, New York: Millbank Memorial Fund. 1991.

West, Jane. Federal Implementation of the Americans with Disabilities Act, 1991-1994. New York, New York: Millbank Memorial Fund. 1994.

WHM Group. *Choice and Opportunity: Formative Evaluation*. Final Report. Document and Literature Review. 1996.

Woodhill, Gary, Renwick, Rebecca, Brown, Ivan and Raphael, Dennis. Being, Belonging, Becoming: An Approach to the Quality of Life of Persons with Developmental Disabilities. In

Goode. *Quality of Life for Persons with Disabilities: International Perspectives and Issues.* 57-74. 1994.

Woodill, Gary. *Independent Living & Participation in Research: A Critical Analysis*. Toronto: Centre for Independent Living in Toronto (CILT), Inc. 1992.

Working Group on the Employment of Persons with Severe Disabilities. *Towards Equality in Ontario's Labour Market: The Employment of Severely Disadvantaged Persons With Disabilities.* Final Report. January, 1994.

Yelin, Edward. Disability and the Displaced Worker. Rutgers University Press. 1992.

Yelin, Edward. Displaced Concern: The Social Context of the Work-Disability Problem. *The Milbank Quarterly*. 67 (supplement 2, part 1), 114-165. 1989.

Yelin, Edward. The Myth of Malingering: Why Individuals Withdraw from Work in the Presence of Illness. *The Milbank Quarterly*. 64(4), 622-649. 1986.

Young, William R. *Disability: Socio-Economic Aspects and Proposals for Reform*. Research Branch, Library of Parliament. Ottawa. January, 1996.

Zeitzer, Ilene R. Social Insurance Provisions for Children with Disabilities in Selected Industrialized Countries. *Social Security Review*. Fall, 1995.

Zola, I.K. Decline in Disability Coverage. *Benefits Canada*. 19(2), 15. Feb. 95.

Zola, I.K. Disability Statistics, What We Count and What It Tells Us: A Personal and Political Analysis. *Journal of Disability Policy Studies*. 4(2), 9-39.

<sup>1</sup>JAN is a United States-based organization which has been contracted to provide information services about accommodation, using a toll-free telephone line within Canada.

<sup>2</sup>We became aware of this volume only very recently and have been unable to review it in detail for the present study. However, it appears to contain very important research relating to the diversity of persons with disabilities and to health services and related in-kind benefits, particularly the eligibility of persons with disabilities who are working to retain necessary health and disability-related services.